AMENDED

AND RESTATED

LEASE AND MANAGEMENT AGREEMENT

BY AND BETWEEN

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT

AND

COLORADO ROCKIES BASEBALL, CLUB, LTD.

DATED MARCH 30, 1995

(WITH AMENDMENTS)

AMENDED

AND RESTATED

LEASE AND MANAGEMENT AGREEMENT

BY AND BETWEEN

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT

AND

COLORADO ROCKIES BASEBALL CLUB, LTD.

RECI'	TALS			•			. 1
	j						
ARTI	CLE :	I LEASED PREMISES	d o		. ,	•	. 2
		· <u> </u>		,			
ARTI	CLE :	II CONSTRUCTION OF STADIUM		•			. 3
÷	2.1	Partnership's Participation				_	. 3
	2.2	District Responsibilities and Disclaimers					. 4
	2.3	Basic Stadium Features					. 7
	2.4	Partnership Property					. 7
	2.5	Leased Equipment					10
.=	ii	· · · · · · · · · · · · · · · · · · ·					
ARTIC	TLE 1	III TERM				_	10
	3.1	Term	•	-	•	•	10
	3.2	Deferral of Commencement Date		•	•	•	11
	3.3	Options to Extend				•	11
	11 11 11	· · · · · · · · · · · · · · · · · · ·	•	•	•	•	-
ARTIC	ile i	V OPERATING COSTS AND PAYMENTS TO THE DIST	RIC	T.			13
	4.1	Operating Costs				-	
4	4.2	Payments to the District	•	• •	•	•	14
	4.3	Budget Review					
	4.4	Access to Partnership's Books and Records	•	• •	•	•	20
			•	• •	•	•	20
ARTIC	ile v	USE OF STADIUM					21
	5.1	Partnership's Use	•	• •	۰	٠	22
	5.2	District's Rights					2-
	5.3	Non-Baseball Events	•	• •	•	•	28
	5.4	Management	•	• •	•	•	28
	l li		•	• •	•	•	20
ARTIC	LE V	I INSURANCE AND SUBROGATION					28
	6.1	Type of Insurance	•	• •	•	•	
	6.2	Insurance Specifications	•	• •	•	•	28 30
	6.3	District Approval	•	• •	•	٠	30
4	6.4	District Approval District Copies	•	• •	•	•	31
1	6.5	Phird-Darty Ingurance	*	• •	•	•	31
	6.6	Third-Party Insurance		• •	•	•	32
	6.7	District Approval of Underwriter	•	• •	•	•	32
	6.8	Notice of Cancellation or Change	•	• •	•	•	33
ì	6.9	Valver of Subrogation	•	• •	•	•	
= 1	6.10	Waiver of Subrogation	•		•	•	34
ì	6 11	Bonds District's Governmental Immunity	•	• •	•	•	34
	6 12	Application of Insurance Proceeds	•		•	•	34
•	46	ANNTICATION OF INSULANCE PROCEEDS	•	• •	•	•	35
ייידיים	רער פור	T TWD PART PT (2 PT () 2 PT					
ბ∞# <i>ძ</i> ბა <i>ძა</i> აბა <i>დ</i> ~		II INDEMNIFICATION	•	۰ ۰	٠	5	35
-	7 2	Partnership Indemnification	•	• •	•		35

7.3 <u>District Cooperation</u>	3 €
ARTICLE VIII MAINTENANCE, REPAIRS, AND OPERATIONS	3 -
8.1 <u>Definitions</u>	3 -
8.2 Partnership Obligations	. (
8.3 Playing Field Maintenance and Repairs	2.2
8.4 Partnership's Failure to Maintain or Repair	: -: ! 6
8.5 Utilities	: (
8.6 Improvements and Alterations by the Partnership 4	
8.7 Partnership's Personal Property	: <i>1</i> : <i>C</i>
8.8 Partnership's Management Qualifications	: 7
8.9 Inspection by the District	-
8.10 Repairs Procedure	1 2
8.11 Repairs Fund	0
8.11 Repairs Fund 68.12 Use and Return of Repairs Fund 6	· A
8.13 Emergency Repairs	- 2
8.14 Competitive Process	-
8.14 Competitive Process	
8.15 <u>Limitation of Responsibility</u>	
8.16 Affirmative Covenants 6	6
ARTICLE IX UNTENANTABILITY	_
9.2 Continuing Obligations	
9.3 Rebuilding Obligations	8
ARTICLE & RIGHT OF ENTRY	_
ARTICLE & RIGHT OF ENTRY	9
ARTICLE XI DEFAULT AND REMEDIES	_
11.1 Default by the Partnership	Ü
11.2 District's Remedies	U
	1
11.3 Default by District	4
11.4 Partnership's Remedies	
11.5 General Provisions	6
ARTICLE XII SURRENDER OF LEASED PREMISES	
ARTICLE XII SURRENDER OF LEASED PREMISES	7
12.1 <u>General</u>	7
12.2 Partnership's Property	7
12.3 Abandoning Leased Premises	8
ADMICIU VIII DDIACAMPAN AD MESSA	
ARTICLE XIII RELOCATION OF TEAM	9
ADMIT I VIII CAIR OR BRANGWING	
ARTICLE XIV SALE OF FRANCHISE	9
ARTICLE XV ASSIGNMENT	0
ARTICLE XVI EMINENT DOMAIN	
ARTICIE XVI EMINENT DOMATN	7

1,		
16.1		g
16.2	Performance of Work	9
16.3	Temporary Taking	8
16.4	Broadway Viaduct	8.
16.5	Broadway Viaduct Condemnation of Parking Facilities	Q .
	TYPESHALLON OF FAIRTING FACTURES	8
ARTICLE XV	VII SALE OF STADIUM LAND AND PURCHASE OPTION	8
ARTICLE XV	VIII STADIUM SECURITY AND CROWD CONTROL	8
18.1	Event Security	01
18.2	Twenty-Four (24) -Hour Security	Q!
ļi.		
ARTICLE XI	EX BONDS	8
19.1	Bond Constraints	QI
4.2.2	pereasance	Ωί
19.3	Repair, Improvement and Alteration Restrictions	0 (
). H		86
ARTICLE XX	CONCESSIONS	۰
20.1	Partnership Rights	0.
20.2	Construction of Concession Facilities	0.5
20.3	Concessionaire Agreement	87
20.4	Additional Stands	87
20.5	Student Morrors Day/Doctors	88
20.5	Student Movers Bar/Restaurant Operations	88
20.6	Environmental Responsibilities	88
ARTICLE XX	I ADVERTISING AND SIGNS	0.0
21.1	Stadium Signs.	85
21 3	Other Advertising	89
21 4	Approval Dights	90
60 de • 72	Approval Rights	90
ARTICLE XX	II BROADCAST FERS	0.0
l _e		90
ARTICLE XX	III COVENANT OF QUIET ENJOYMENT	91
}		
ARTICLE XX:	IV ADDITIONAL OBLIGATIONS OF PARTNERSHIP	91
24.1	Promotion Attendance of Children From Low Income Families Cooks Field Wideston	91
24.2	Attendance of Children From Low Income Families	01
Z**)	CUULA PIRIO VIGAA ABA UBAFAAYANN BAAG	92
24.4		
24.5	Public Parking	
24.6	Parking Structure	92
24 7	Parking Structure	93
4	Commitment to Stadium	94
ARTICLE XX	V REPRESENTATIONS BY PARTNERSHIP	
25.1	Valid Evictore	95
25.2	Valid Existence Power: No Limitation on Ability to Perform	95

1' 2						
25.3	Valid Execution		,			9
25.4	<u>Defaults</u>					9
25.5	Good Standing of Corporation					9
25.6	Power of Corporation	_	_	_	_	9
25.7	Valid Execution by Corporation	_	_	_	_	Ġ,
25.8	Team Ownership				_	9'
25.9	Team Ownership Compliance With Laws			•		9.
25.10	Maintenance of Good Standing in League .					9'
31						
ARTICLE XX	VI REPRESENTATIONS BY DISTRICT		۰	۰	۰	91
26.1	Valid Existence					91
26.2	<u>Power: No Limitation on Ability to Perform</u>					9,
26.3	Valid Execution					0.6
40.4	<u>peraults</u>					Q
26.5	Compliance With Laws		•		_	99
ij						
ARTICLE XX	VII MISCELLANEOUS	۰	٠			99
27.1	Naming Rights					99
27.2	Naming Rights Architectural Drawings and Logo			•		99
27.3	Entire Agreement					100
27.4	Default Interest, Attorneys' Fees					100
27.5	Notices Force Majeure Reasons for Disapproval				_	101
27.6	Force Majeure				_	102
27.7	Reasons for Disapproval				_	102
٧, ١, ٥	Successor Bound	_	_	_	_	102
27.9	Governing Law	_	_			102
27.10	<u>Captions and Headings</u>	_	_	_	_	103
27.11	Plural/Singular	_	_			ากร
27.12	EXECUTION OF COUNTERDARTS		_	_	_	103
27.13	Further Assurances					102
27.14	Anti-Discrimination Clause	_				707
27.15	Material Adverse Effect	_	_	_		103
27.16	Amendment: Walver				-	104
27.17	Severability	_				104
27.18	No Joint Venture					105
27.19	Words of Limitation					105
27.20	No Third Party Beneficiary				_	105
	•	-	-	-	-	
index of de	FINED TERMS					7 0 7

EXEIBIT A
Stadium Land

EXHIBIT B Map

EXEIBIT C
Development Documents

EXHIBIT D
District's Construction Obligations

Partnership Property

EXHIBİT F Repair Areas

EXHIBIT G
Transportation Management Program

٧

AKENDED AND RESTATED LEASE AND MANAGEMENT AGREEMENT

THIS AMENDED AND RESTATED LEASE AND MANAGEMENT AGREEMENT (the "Agreement") dated MARCH 30, 1995, is by and between the Denver Metropolitan Major League Baseball Stadium District, a body corporate and politic and a political subdivision of the State of Colorado (the "District"), and Colorado Rockies Baseball Club, Ltd., a Colorado Limited Partnership, formerly known as Colorado Baseball Partnership 1993, Ltd., a Colorado limited partnership (the "Partnership").

RECITALS

- A. The General Assembly of the State of Colorado (the "Legislature") has declared that a Major League Baseball franchise in the State of Colorado would stimulate economic development throughout the State resulting in increased tourism, the creation and maintenance of new jobs, and the attraction and retention of sports and entertainment events. Accordingly, pursuant to the Denver Metropolitan Major League Baseball Stadium District Act at \$32-14-101, et seq., of the Colorado Revised Statutes, as it may be amended from time to time, (the "Act"), the Legislature has created the District for the purposes of acquiring a site, financing, constructing, and leasing a major league baseball stadium.
- B. The Partnership is the holder of a franchise (the "Franchise") located within the boundaries of the District issued by The National League of Professional Baseball Clubs ("National League") and is the owner of the "Colorado Rockies" professional baseball team (the "Team").
- C. The District has issued the Denver Metropolitan Major League Baseball Stadium District Sales Tax Refinance and Improvement Revenue Bonds (Baseball Stadium Project), Series 1991A, and the Denver Metropolitan Major League Baseball Stadium District

Sales Tax Refunding Revenue Bonds (Baseball Stadium Project), Series 1994 (collectively, the "Bonds"), to be used to pay the costs of land acquisition, construction, and related costs in connection with the Leased Premises (as defined in Article I hereof).

- D. The District desires to lease to the Partnership and the Partnership desires to lease from the District the Leased Premises.
- E. The Partnership has determined that certain property not provided by the District as part of the Leased Premises is necessary for the Partnership's use of the Leased Premises and the Partnership is responsible for providing and shall provide and own this property.
- F. On March 14, 1991 the District and the Partnership entered into a Memorandum of Lease Terms (the "MOL") outlining the basic terms and conditions under which the Partnership will use the Leased Premises, at which time the District and the Partnership anticipated this amendment to memorialize the details of the concepts outlined in the MOL.
- G. This Agreement amends the basic terms and conditions set forth in the MOL and establishes the final terms of the agreement regarding the Partnership's use and management of the Leased Premises and supersedes all prior written and oral agreements between the parties as to all matters.

ARTICLE I

LEASED PREMISES

In consideration of the mutual agreements contained in this Agreement, and incorporating herein the recitals set forth above, the District hereby leases to the Partnership, and the Partnership hereby leases from the District, the real property described in Exhibit A attached hereto and made a part hereof (the "Stadium Land"), together with all the improvements to be constructed 90

thereon, including but not limited to the stadium constructed thereon, and all improvements, additions, alterations, fixtures, equipment, and installations owned or leased by the District and constructed, provided or added to the stadium at any time (the "Stadium"), and all other improvements, additions, alterations, fixtures, equipment and installations owned or leased by the District and constructed, provided or added to the Stadium Land at any time (collectively, the "Leased Premises"). A map of the Leased Premises delineating the Stadium and the Stadium Land is attached hereto as Exhibit B.

ARTICLE II

CONSTRUCTION OF STADIUM

2.1 Partnership's Participation. The District shall construct the Stadium and all other portions of the Leased Premises in accordance with the GMP documents list, specifications and qualifications attached hereto as Exhibit C and incorporated herein, and any supplements, amendments or revisions agreed to by the District and the contractor (the "Development Documents"). The Partnership and the District shall each cooperate and actively work with the other in the construction of the Stadium. The Partnership shall have the right to approve the following matters, which approval shall not be unreasonably withheld, and which approval shall be deemed to have been given by the Partnership if no written objection is delivered to the District within three (3) business days after receipt by the persons designated by the Partnership to receive such matters: (a) new Development Documents dated after the date of this Agreement, (b) new Material Alterations to then existing Development Documents either dated on or prior to the date of this Agreement or otherwise previously approved by the Partnership, or (c) any contracts to be entered into by the District subsequent to the execution of this Agreement which may

materially affect the Partnership's maintenance and operations rights and obligations pursuant to this Agreement. The Partnership shall provide the District, at the time of the Partnership's execution of this Agreement, a separate writing stating the persons to whom notice of such matters shall be given. Alterations shall mean any change having a contract value in excess of \$150,000.00. Subject to the foregoing, the District shall be entitled to add to or delete from, the plans and specifications for the Leased Premises and the Development Notwithstanding the above, the District shall be Documents. entitled to delete from such plans and specifications and Development Documents any facilities, if necessary, due to the District's financial constraints, provided that the persons designated to receive notice in this Section 2.1 are notified of If the Partnership does not pledge in writing to such deletions. commit the financial resources to the District necessary to alleviate the need for such deletions within three (3) business days after receipt by such persons of the notice provided for in the preceding sentence, the consent of the Partnership to such deletion(s) shall be deemed to have been given. If such written commitment is received within three (3) business days after the receipt of such notice, the deletions shall not be made. expressly agreed that the District possesses final decision authority regarding the design, planning, financing construction of the Leased Premises. The Partnership shall purchase and own certain property to be included in the construction to be completed on the Stadium Land, as set forth and defined in Section 2.4.

2.2 <u>District Responsibilities and Disclaimers</u>. Subject to the provisions of this Section 2.2, the District is ultimately responsible for the design, development and construction of the

Leased Premises in accordance with the Development Documents. This responsibility includes construction of the Stadium and all other aspects of the Leased Premises within the financial constraints of the District as determined by, and in the sole discretion of, the District's Board of Directors, as established by the Act (the "Board"). Nothing in this Agreement shall be construed as requiring the District to operate in excess of these constraints in the construction of the Stadium or development of any other aspect of the Leased Premises.

The Leased Premises shall meet all requirements of the National League and the Office of the Commissioner of Baseball. At the time of the execution of this Agreement by the District, the parties agree that the National League and the Office of the Commissioner of Baseball have stated that the Leased Premises does meet all such requirements.

Subject to the provisions of Section 2.1, the District may supplement, amend or revise the Development Documents as the District desires or deems necessary, provided the Partnership is advised of any such supplements, amendments or revisions in a timely manner. The District may also, if deemed necessary by the District due to its financial constraints, defer the construction of specific Stadium features or any other aspects of the Leased Premises described in the Development Documents, subject to the provisions of Section 2.1, so long as it does not unreasonably interfere with the Partnership's operations.

Notwithstanding anything contained herein to the contrary, after the 1995 Major League Baseball Season the District shall fund additional improvements and modifications to the Leased Premises, as requested by the Partnership, in an amount not to exceed \$1,500,000.00, subject to the availability of District funds, as soon as funds become available to the District for that purpose.

All or part of the amount of such additional improvements and modifications to be funded by the District may be used as a portion of the Partnership's cash contribution toward a parking structure pursuant to Section 24.6(b). If such additional improvements generate additional revenues not previously received by the Partnership and the disposition of such revenues is not provided for pursuant to the provisions of this Agreement, the District and the Partnership shall mutually agree upon a modification to this Agreement regarding payments to the District, before the additional improvements are undertaken.

NOTWITHSTANDING ANYTHING IN THIS SECTION 2.2 OR ELSEWHERE IN THIS AGREEMENT TO THE CONTRARY, THE DISTRICT SHALL DELIVER THE LEASED PREMISES TO THE PARTNERSHIP ON THE COMMENCEMENT DATE "AS THE COMMENCEMENT DATE, THE LEASED PREMISES CONSTITUTE AN OPERATIONAL BASEBALL STADIUM, BUT NO WARRANTY OR REPRESENTATION IS MADE THEREAFTER OR WITH RESPECT TO ANY PARTICULAR COMPONENT OF THE LEASED PREMISES ON THE COMMENCEMENT DATE. DISTRICT HEREBY DISCLAIMS ANY OTHER WARRANTY OR REPRESENTATION, EXPRESS OR IMPLIED, OR ARISING BY OPERATION OF LAW, INCLUDING BUT LIMITED TO ANY WARRANTY OF CONDITION. HABITABILITY, MERCHANTABILITY, TENANTABILITY OR FITNESS FOR PARTICULAR PURPOSE OR The District will make available to the Partnership all warranties of all manufacturers, contractors, subcontractors, lessors (including Equipment Lessors), architects and engineers provided for the benefit of the District. The Partnership shall pursue at its expense all claims made under such warranties. the extent that the work performed as a result of the enforcement of any such warranty or work performed despite the Partnership's inability to enforce any such warranty could be considered a Repair as defined in Section 8.1(c), the Partnership may use funds in the Repairs Fund to pay for its reasonable costs incurred in pursuing

- claim(s) made under such warranty, including its reasonable attorney's fees, subject to the provisions of Section 8.10.
- Basic Stadium Features. The Stadium will be a firstclass, state-of-the-art (with respect to design, construction and operating efficiency), natural turf, open-air facility, which, while traditional and intimate in basic design, will be constructed with modern amenities for fan comfort. The provisions of the Development Documents control the specific features and appurtenances which the Stadium will contain. The District will construct, at its expense, the portions of the Stadium and the Leased Premises set forth in Exhibit D attached hereto. Except as expressly provided in this Agreement, the Partnership agrees that in no event shall any sums be expended for the acquisition or construction of the Leased Premises including personal property to be located on the Leased Premises by the Partnership or any third party at any time prior to or during the Term, which would have a Material Adverse Effect on the District.

2.4 Partnership Property.

- (a) The Partnership shall provide and bear all costs of acquisition, ownership, operation, maintenance, repair and replacement of all property on the Stadium Land owned by the Partnership, including, but not limited to, the property listed on Exhibit E attached hereto (the "Partnership Property").
- (b) The District may purchase at its option and the Partnership shall sell to the District, at its then fair market value, the Partnership Property, or any part thereof, at any time during the Term or Grace Term, or upon the expiration or termination of this Agreement. The fair market value of the Partnership Property shall be determined by the mutual agreement of the District and the Partnership. If the District and the Partnership cannot agree on a fair market value, they shall

mutually select an appraiser to determine the fair market value of the Partnership Property. If the District and the Partnership cannot agree upon an appraiser to determine the fair market value, the District and the Partnership shall each select an appraiser, and those two appraisers shall determine, together, the fair market value of the Partnership Property. If the two appraisers cannot reach a mutual determination of the fair market value, they shall mutually select a third appraiser who shall determine the fair market value of the Partnership Property, which determination shall be binding upon the District and the Partnership.

(c) the expiration or termination this Agreement, the Partnership may voluntarily remove the Partnership Property from the Stadium Land. The District may require the Partnership to remove from the Stadium Land the Partnership Property, or any part thereof, which the District has not purchased and the Partnership has not voluntarily removed upon the expiration or termination of this Agreement. If the Partnership Property, or any portion thereof designated by the District, is not removed from the Stadium Land within a reasonable time after the expiration or termination of this Agreement and upon the District's written request to so remove, not to exceed ninety (90) days, Partnership Property, or designated portion thereof, shall be deemed abandoned, and shall, at the District's election, become part of the real property constituting the Leased Premises, with title thereto held by the District or its successors in interest. Notwithstanding, if abandonment of any item of Partnership Property will cause a Material Adverse Effect, the Partnership shall remove such item at its expense if the District elects not to purchase it regardless of the effect on the value, utility or damage to the Stadium.

- (d) The Partnership shall not grant or permit any liens, claims, security interests or encumbrances against any Partnership Property without the District's prior approval, except for a purchase money security interest which satisfies the following conditions:
- (i) the loan documents relative to the purchase money security interest shall provide that:
- (A) the lender shall provide notice to the District of any default of the Partnership thereunder and shall deliver to the District copies of the then current loan documents at the time of such notice;
- (B) the District shall have an opportunity to cure such default within no less than thirty (30) days from the District's receipt of notice of the Partnership's default;
- (C) the District's right to notice of default and to cure cannot be amended without the prior written approval of the District;
- (D) the original principal amount of the loan shall not exceed the invoice cost of the item or items of Partnership Property securing the loan;
- (E) all of the proceeds of the loan shall be used to purchase the item or items of Partnership Property securing the loan;
- (F) the original principal amount of the loan cannot be increased nor payments or principal or interest be waived or rescheduled without the prior written approval of the District; and
- (G) the original term of the loan shall not exceed the lesser of eight (8) years or the useful life of the item or items of Partnership Property securing the loan, and cannot be extended without the prior written approval of the District.

- (ii) the lender shall agree in writing with the District prior to the perfection of the purchase money security interest that the loan agreement contains all of the provisions referred to above, that the lender agrees with the District to abide by all of such provisions, and that to the best of its knowledge items i(D), (E) and (G) above are then accurate. The lender shall also provide to the District a writing setting forth the original principal amount, interest rate, amortization, and term of the purchase money loan and a description of the items of Partnership Property serving as collateral, which shall be marked "Confidential" and treated as confidential financial information by the District. under such a purchase money loan will constitute a default under this Agreement. If the District advances money to cure a default under such a loan, the Partnership will be in default under this Agreement and will be liable to the District for the amounts advanced plus interest at the rate of four percent (4%) above Colorado National Bank's (or its successor bank(s)) prime lending rate.
- 2.5 <u>Leased Equipment</u>. The District will be leasing from third parties (the "Equipment Lessors") certain escalators, elevators, seats and bleachers (the "Leased Equipment") which will be installed in the Stadium and are part of the Leased Premises. As part of the Leased Premises, the Leased Equipment is subleased to the Partnership pursuant to the terms of this Agreement as of the Commencement Date.

ARTICLE III

TERM

3.1 Term. This Agreement shall become effective as of the date first above written upon execution of this Agreement by the District and the Partnership. The Partnership's occupancy of the Leased Premises shall commence upon substantial completion of

construction of the Stadium, permitting the Partnership to use the Stadium for its intended use (the "Commencement Date"). Subject to the provisions of Section 3.3, the Partnership's occupancy of the Leased Premises shall end twenty-two (22) years after the Commencement Date. Upon determination of the Commencement Date, the parties shall execute an appropriate supplement to this Agreement setting forth the last day this twenty-two (22) year period of occupancy.

- 3.2 <u>Deferral of Commencement Date</u>. The District hereby agrees that it will cause the Commencement Date to occur on or before April 6, 1995. If the District does not cause the Commencement Date to occur on or before April 6, 1995, or April 6 of any subsequent year during the Term, the Partnership shall be entitled to defer the Commencement Date until ninety (90) days following the completion of the Major League Baseball Season for 1995 dr such subsequent Major League Baseball Season (or such earlier date as agreed to by the District and the Partnership), by giving written notice to the District of the Partnership's election to so defer no later than April 6, 1995 or April 6 of any subsequent year during the Term. Each such "Major League Baseball Season shall commence on the day of the Team's first home game (including pre-season play, if any; other than those pre-season home games played outside the boundries of the District) in any calendar year and shall end on the day of the Team's last Major League Baseball Game (including post-season play, if any) in such calendar year. During any period of deferral of the Commencement Date pursuant to this Section 3.2, the Partnership shall not be obligated to make any payments to the District pursuant to Section 4.2(b)
 - 3.3 Options to Extend.
 - (a) Provided the Partnership is not then in default

under the provisions of this Agreement, the Partnership has the option to extend this Agreement for an additional five (5) year term (the "First Extension Term"). The First Extension Term shall commence the day after the last day of the twenty-two (22) year occupancy period as determined pursuant to Section 3.1., subject to the provisions of Sections 3.3(b) and (d) below. Provided the Partnership is not then in default under the provisions of this Agreement, the Partnership has the option to extend this Agreement for a second additional five (5) year term (the "Second Extension Term"). The Second Extension Term shall commence the day after the expiration of the First Extension Term. Provided the Partnership is not then in default under the provisions of this Agreement, the Partnership has the option to extend this Agreement for a third additional five (5) year term (the "Third Extension Term"). Third Extension Term shall commence the day after the expiration of the Second Extension Term.

The initial twenty-two(22) year occupancy, the First Extension Term, the Second Extension Term and the Third Extension Term shall, collectively, constitute the "Term" of this Agreement.

- (b) To exercise its options to extend, the Partnership must provide the District with written notice of its intent to renew no later than one hundred eighty (180) days prior to the last day of the twenty-two (22) year occupancy period, for the First Extension Term, and no later than one hundred eighty (180) days prior to the expiration of the First Extension Term, for the Second Extension Term, and no later than one hundred eighty (180) days prior to the expiration of the Second Extension Term, for the Third Extension Term.
- (c) If, prior to the expiration of the Third Extension Term, the District and the Partnership have not executed a new lease agreement or amended this Agreement to provide for an

additional extension term, and if the Partnership is not then in default under the provisions of this Agreement, the Partnership shall be entitled to occupy the Leased Premises for an additional one year period under the terms and conditions of this Agreement then in effect (the "Grace Term"), as long as written notice is provided to the District by the Partnership of its intent to so occupy no later than one hundred eighty (180) days prior to the expiration of the Third Extension Term and subject to the provisions of this Section 3.3(c) and Section 3.3(d).

(d) Time is of the essence with regard to the notices required pursuant to Sections 3.3(b) and (c).

ARTICLE IV

OPERATING COSTS AND PAYMENTS TO THE DISTRICT

- 4.1 Operating Costs. The Partnership shall be responsible for all costs associated with the Repair, Maintenance, operation, use and ownership of the Leased Premises and Partnership Property, including but not limited to the following costs:
- (a) wages and salaries of all Partnership employees engaged in operation, Repair, Maintenance or security of the Leased Premises and Partnership Property, including taxes, insurance and benefits related thereto;
- (b) all supplies and materials used by the Partnership in operation and Maintenance of the Leased Premises and Partnership Property;
- (c) costs of all utilities for the Leased Premises and Partnership Property, including but not limited to the cost of water, power, heating, lighting, air conditioning, ventilating, sewer, waste water, and trash disposal;
- (d) cost of all Maintenance, Stadium Management and service agreements for the Leased Premises and Partnership Property and the equipment therein, including but not limited to alarm

service, janitorial service, window cleaning, security service, elevator maintenance, police, game-day operations, landscaping, snow removal, parking lot maintenance(subject to the provisions of Section 4.2(c) hereof), and playing field maintenance;

- (e) cost of all insurance as provided in Article VI, relating to the Leased Premises and Partnership Property, including but not limited to the cost of casualty and liability insurance applicable to the Leased Premises and Partnership Property; and
- (f) all taxes and assessments and special assessments imposed upon the Partnership Property and the Partnership's assets on the Stadium Land by any governmental bodies or authorities, and all charges specifically imposed in lieu of such taxes.

The foregoing is presented by way of example only and is not intended to be an all-inclusive list.

- 4.2 Payments to the District. In addition to the payment of all costs as set forth in Section 4.1 and Section 8.11(a), the Partnership will pay to the District the following payments (which payments shall not be set-off against any sums due to the Partnership from the District hereunder) for the Term and Grace Term:
- (a) The District will incur, in its administration after completion of the construction of the Leased Premises, day-to-day operating costs. These operating costs are to be reimbursed from tax revenues designated for such reimbursement pursuant to § 32-14-115 of the Act. Notwithstanding, if revenues to the District from all sources (including tax revenues legally available for such use and revenues under this Agreement) are not sufficient to reimburse the District for its day-to-day operating costs, the Partnership shall pay the District, in cash or by providing services as requested by the District, the difference between the actual operating costs and such revenues; provided, however, that

the amount payable from the Partnership plus the value of the services provided by the Partnership to the District shall not exceed \$150,000 per year, as adjusted annually on January 1st of each year during the Term and Grace Term for annual percentage increases in the Denver-Boulder Consumer Price Index. All payments received by the District and costs of services provided by the Partnership to the District pursuant to Sections 4.2(b), (c), (d), (e) and (f), 5.2(e) and (f) and 8.9(b) of this Agreement, in any calendar year shall constitute a credit against any obligation of the Partnership for such reimbursement or additional payment under this Section 4.2(a) to the District in that calendar year.

The annual percentage increases in the Denver-Boulder Consumer Price Index shall be the percentage difference between the average index (computed by tallying all index publications during the course of any given year and dividing the sum by the number of publications made during the year) for the year prior to each annual payment and the average index for the calendar year two years prior to such annual payment (i.e., if the average index for 1994 equals 100 and the average index for 1995 equals 103, the amount the Partnership will be obligated to pay the District in 1996 (in cash or by providing services) for any shortfall in revenues available to pay the District's administration costs shall increase by 3% (3/100) to \$154,500.00 (\$150,000.00 + \$4,500.00 (.03 x \$150,000.00) = \$154,500.00)).

If the Denver-Boulder Consumer Price Index shall become unavailable to the public because publication is discontinued, or otherwise, then a comparable index which reflects changes in the cost of living or purchasing power of the consumer dollar published by an agency of the United States Government, any other governmental agency, a major bank, or other financial institution, a university, or a recognized financial publication shall be

substituted therefor, as agreed to by the District and the Partnership.

The District shall provide an accounting by January 31 of each year of its actual operating expenditures for the previous year. If actual operating expenses are greater than the amount paid or provided to the District pursuant to sections 4.2(b), (c), (d), (e) and (f), 5.2(e) and (f) and 8.9(b) of this Agreement for any year during the Term or Grace Term, the Partnership shall promptly pay any additional sum to the District necessary to insure that all of the District's actual operating costs are reimbursed to the extent provided for in this Section 4.2(a) (subject to the annual limitation, as adjusted). Any such reimbursement or additional payment shall be made no later than February 15 of each year.

During the Term and Grace Term, the Partnership will pay the District twenty-five cents (\$.25) per Major League Baseball Game attendee ("Attendee") in excess of 2,250,000 up to 2,500,000 per year; fifty cents (\$.50) per Attendee in excess of 2,500,000 up to 3,000,000 per year; and one dollar (\$1.00) per Attendee in excess of 3,000,000 per year. A "Major League Baseball Game" shall be any baseball game during any Major League Baseball Season in which the Team acts as the host Team for its opponent (i.e., the Team takes the field in the first half of each inning and bats in the last half of each inning of such baseball game); any All Star game which the Team has requested, through application otherwise, be played within the boundaries of the District; any baseball game which the Partnership or the Team gives its permission to be played within the boundaries of the District; or any other baseball game played by the Team within the boundaries of the District. The Partnership shall report attendance to the District upon a turnstile count and make payments to the District

under this Section 4.2(b) based upon the attendance so reported. The Partnership will guarantee a minimum payment of \$100,000.00 per year to the District pursuant to this provision (the "Guaranteed Minimum Payment"). The foregoing amounts per Attendee shall be adjusted annually (and rounded to the nearest 1¢) in an amount equal to the percentage increase in the Partnership's Average Ticket Price. The "Average Ticket Price" for each Major League Baseball Season during the Term and Grace term shall mean the weighted average price of all tickets determined by multiplying each ticket price by the number of all seats at that price, adding the results of those calculations for all ticket prices, dividing that sum by the total number of seats in the Stadium. average ticket price shall be determined exclusive of suite rental (including admission thereto), Stadium Club membership and other amenity charges as reasonably determined by the Partnership. Payment shall be made no later than December 1 of each year during the Term and Grace Term, commencing December 1 of the same year as the year of the Commencement Date.

(c) Subject to the provisions of Section 24.6, all parking net profits (net profits shall equal gross receipts less operating costs; operating costs are all direct incremental costs paid to or retained by third parties or Affiliates of the Partnership, provided costs attributable to any Affiliates of the Partnership are competitive with the costs third parties would charge for providing the same services, as a direct result of parking operations) shall be shared on a twenty percent (20%) (to the District), eighty percent (80%) (to the Partnership) basis for all Major League Baseball Games played during the Term and Grace Term and Non-Baseball Events for which the Partnership is entitled to use the Leased Premises pursuant to Section 5.1. Subject to the provisions of Section 24.6, all other parking net profits shall be

shared on an eighty percent (80%) (to the District), twenty percent (20%) (to the Partnership) basis. Payment shall be made to the District, along with a written report itemizing gross receipts and operating costs, for each calendar month on the 15th day of the succeeding calendar month, with the first payment to be received on the 15th day after the calendar month in which the Team's first Major League Baseball Game is played in the Stadium.

The Partnership shall ensure that the operator of the parking on the Stadium Land, including, but not limited to, the parking structure, if any, provides an annual accounting of all revenues from the previous year, to be provided on or before January 31 of each year during the Term and Grace Term beginning the year after the year of the Commencement Date. If necessary, the amounts paid to the District pursuant to this Section and Section 24.6 shall be adjusted annually pursuant to the results of the accounting.

The Student Movers building at 2145 Blake Street has (d) been designed to become an integral part of the Stadium and is included in the term "Stadium". All or part of the first floor and basement of that building will be incorporated into the Stadium design as a bar and restaurant for use by fans during Major League Baseball Games and potentially year-round. As part of the construction of the Stadium, the District will provide, at its expense, a shell space, as more particularly described in Exhibit E attached hereto, on the first floor and basement of the Student Movers building to be used for a bar and restaurant. The District will receive from the Partnership three percent (3%) of monthly gross sales of the bar and restaurant if the Partnership contracts with the concessionaire under the concession agreement described in Section 20.3 for operations. If the Partnership contracts with any operator other than such concessionaire, the District shall receive

from the Partnership five percent (5%) of the monthly gross sales of the bar and restaurant. Payment shall be made by the Partnership to the District, along with a written report of gross sales, for each calendar month on the 15th day of the succeeding calendar month with the first payment to be received after the bar and restaurant's first full calendar month of operation.

The Partnership shall ensure that the concessionaire operating the bar and restaurant provides an annual accounting of all revenues from the previous year, to be provided on or before January 31 of each year during the Term and Grace Term beginning the year after the year of the Commencement Date. If necessary, the amounts paid to the District pursuant to this subsection shall be adjusted annually pursuant to the results of the accounting.

- The Partnership will pay the District an amount equal to forty percent (40%) of the net profit (defined as gross receipts from all sources less operating costs - operating costs are all incremental out-of-pocket costs paid to third parties or Affiliates of the Partnership, provided costs attributable to any Affiliates of the Partnership are competitive with the costs third parties would charge for providing the same services, as a direct result of the event) from any and all of the Non-Baseball Events allowed to be sponsored by the Partnership and from any and all other events held by the Partnership on the Leased Premises (except for Major League Baseball Games or any event held immediately preceding or following a Major League Baseball Game) from which the Partnership realizes revenue. These payments shall be submitted along with a written report itemizing gross receipts and operating costs within thirty (30) days after each such Non-Baseball Event held by the Partnership.
- (f) The Partnership will pay the District 50% of the net proceeds realized by the Partnership directly as a result of all

Off-Season Non-Baseball Events held by the District pursuant to Section 5.2(d), including, without limitation, net proceeds from novelty and concession sales. Net proceeds shall equal gross revenues less direct incremental costs paid to or retained by third parties or Affiliates of the Partnership, provided costs attributable to any Affiliates of the Partnership are competitive with the costs third parties would charge for providing the same services, as a direct result of Off-Season Non-Baseball Events held by the District. Payments shall be submitted along with a written report itemizing gross receipts and direct incremental costs within thirty (30) days after the conclusion of each Non-Baseball Event held by the District.

- Budget Review. The District agrees to conduct its business in a prudent, cost efficient manner and to use its best efforts to operate within the limits set forth in §32-14-115(a) of the Act in the performance of its administrative duties under the Act and, philosophically, agrees to prepare its operating budgets as if it were spending its own funds. Beginning with the District's budget for the first fiscal year of the Term, the Partnership shall have the right to review the District's annual budget with the District, prior to adoption. The District shall cooperate with the Partnership in such review. Nothing in this Section 4.3 shall be construed, however, as giving the Partnership the authority to dictate or require any addition, deletion, change, modification, or supplement to any District annual budget or any Nothing in this Section 4.3 shall be provision(s) therein. construed as altering or limiting the Partnership's obligations pursuant to Section 4.2(b).
- 4 4 Access to Partnership's Books and Records. Each payment pursuant to Section 4.2 shall be accompanied by a statement certified as true and correct by the Partnership's chief financial

officer, showing with reasonable specificity all computations relating thereto as may be reasonably required by the District. The District shall have the right as reasonably required, through the use of an independent certified public accounting firm selected by the District, at any time during normal business hours and at the offices of the Partnership, to review all records of the Partnership which relate to any such payments. In the event such review results in a determination that the computations are erroneous and that such error resulted in an underpayment to the District of five percent (5%) or more of what the District was actually entitled to, the error shall be promptly corrected by the Partnership, the Partnership shall promptly remit payment to the District of all amounts underpaid, plus interest thereon at the rate of four percent (4%) per annum above the prime rate charged from time to time by Colorado National Bank or its successor bank(s), and the Partnership shall pay any out-of-pocket costs incurred by the District in reviewing the Partnership's books and records and assuring such errors are corrected. In addition, the Partnership will require that the operators of the Student Movers bar and restaurant and the parking facilities provide the Partnership, and the Partnership will provide the District, each month, with copies of all accountings and reports regarding revenue provided to the Partnership.

- 4.5 Affiliates. Affiliates shall mean persons or entities controlled by, under common control with or controlling the Partnership.
- 4.6 New Taxes. The District shall not impose or assess any user, ticket, seat, amusement or service tax or any other tax of comparable nature upon the Partnership or on events held at the Leased Premises.

ARTICLE V

USE OF STADIUM

- 5.1 <u>Partnership's Use</u>. Commencing with the Commencement Date and continuing for the balance of the Term and Grace Term, the Partnership shall, subject to all limits and conditions imposed upon the Partnership in this Agreement, be entitled to use the Leased Premises on a year-round basis for the following purposes:
- (a) Playing of Major League Baseball Games and activities related to the playing of Major League Baseball Games and the maintenance of a Major League Baseball team, including, but not limited to: training, practices, maintenance and preparation of the Leased Premises to suit such purposes, advertising and marketing of games and ticket sales;
- (b) the operation of club/restaurant facilities to be known as the "Stadium Club", a separate "Club Seating Lounge", and the restaurant and bar facilities to be constructed in the Student Movers Building;
 - (c) the operation of the Partnership's general offices;
- (d) the sale of food and alcoholic and non-alcoholic beverages, souvenirs and other items normally considered "concessions";
- (e) the sale of space on or for advertising signs and billboards as provided in the Development Documents, subject to the provisions of Article XXI;
- (f) any and all other activities which, from time to time during the Term and Grace Term, are associated with, or are conducted in connection with, or are related to, the conduct of the business of a Major League Baseball team; and
- (g) up to five (5) Non-Baseball Events annually. The Partnership may, from time to time, request from the District on a specific event basis an increase in the number of Non-Baseball Events which the Partnership may sponsor annually, and the District

review such request and determine in its sole discretion wither to grant the request.

The Partnership's possession and use of the Leased

- The Partnership shall not occupy or use the Leased Fremises (or permit the use or occupancy of the Leased Premises) for any purpose or in any manner which is unlawful or which violates this Agreement or any other agreement binding upon the District, a copy of which has been provided to the Partnership or law under which the District or its property is bound.
- Non-Baseball Event held by the Partnership, which approval will not be unreasonably withheld. The Partnership shall provide written notice to the District, no later than sixty (60) days prior to the occurrence of any Non-Baseball Event, of its intent to hold such an event. The District shall approve or disapprove such Non-Baseball Event within thirty (30) days after receipt of such notice by the Executive Director of the District. The District shall not be deemed to have unreasonably withheld its consent to a Non-Baseball Event if the District shall, in good faith, determine that such use:
 - (i) is likely to cause material damage to the Leased Premises;
 - (ii) is incompatible with the character of the neighborhood in which the Leased Premises is located;
 - (iii) is likely to tarnish the image or reputation
 of either the Leased Premises or the
 District; or
 - (iv) is in violation of any applicable law, ordinance, rule or regulation, of any

governmental agency with jurisdiction over the Stadium or its use.

This District shall develop a policy and procedure regarding its internal approval process which shall not be inconsistent with the provisions of this Agreement. The District shall advise the Partnership, in writing, of the provisions of such policy and procedure, and any amendments thereto.

- Baseball Game to be played in any facility other than the Leased Premises without first obtaining the written approval of the District, which may be withheld in its sole and absolute discretion. This subsection shall not require the Team to play any Major League Baseball Game in the Stadium if any governmental agency with jurisdiction over the Stadium and its use prohibits such use on the basis of health or safety risks attendant to such use.
- (4) The Partnership shall be responsible at its own expense for advertising and promoting events it conducts on the Leased Premises.

Except for the foregoing limitations and subject to the provisions of Section 5.2 and all other provisions of this Agreement, the Partnership shall have the right during the Term and Grace Term to determine how the Leased Premises shall be used and who may be present therein, and to select, competitively, and employ, concessionaires, licensees, or subtenants with respect to any Stadium operations, including, but not limited to parking lots, concession areas, Stadium Club, Suite catering or advertising space in the Stadium. Subject to the provisions of this Agreement (including but not limited to Section 4.2 and Article XXII), the Partnership shall have the right to proceeds from the conduct of any of the activities permitted to be conducted by the Partnership,

including but not limited to radio and television fees and revenues, signage and advertising sales, concession and vending machine rents or payments, sales of souvenirs, sales of food and beverage (including alcoholic beverages), and club membership fees related to any or all of the foregoing upon terms and conditions deemed acceptable by the Partnership, provided that no contract entered into by or on behalf of the Partnership shall impair any right of the District hereunder, impose any contractual liability on the District or have a Material Adverse Effect on the District.

- 5.2 <u>District's Rights</u>. During the Term and Grace Term, the District will have the following rights regarding the use of the Stadium:
- (a) The District reserves for itself, and the Partnership will provide to the District, without payment of rent or any other consideration therefor, the use of suite number 5.20.05 as described in the Development Documents ("Suite") and all normal privileges which lessees of Suites receive upon payment of rent.
- (b) The District reserves for itself, and the Partnership will provide to the District admission tickets and passes (including but not limited to parking passes) for all Suite seats for all events in the Stadium at no cost to the same extent as such tickets and passes are provided to all suite holders as attendant privileges of owning rights to a suite in the Stadium.
- (c) The District reserves for itself, and the Partnership will provide to the District, at no cost, six season tickets at a location between first base and third base along the infield and within the first twenty-five (25) rows from the field, and all normal privileges which season ticket holders receive, including three (3) parking passes, at no cost.

Subject to the unavailability of the Leased Premises due to Off-Season, Non-Baseball Events previously scheduled by the Partnership and approved by the District, the District shall have priority for the use of the Stadium for Non-Baseball Events held during any Off-Season, during the Term and Grace Term, which Off-Season shall commence upon the expiration of any Major League Baseball Season and end fifteen (15) days prior to commencement of the following Major League Baseball Season ("Off-Season"). During any such District use for Non-Baseball Events, the District will be required to ensure the integrity of the playing field in the Stadium to the reasonable satisfaction of the Partnership. District shall notify the Partnership at least forty-five (45) days in advance of any Off-Season, Non-Baseball Event, which notice will allow the Partnership to reasonably reach a good faith estimate of managerial and operational costs that may be incurred as a result of such Non-Baseball Event, and coordinate with the Partnership in the scheduling of all such events so as not to interfere with the Partnership's rights and duties under this Agreement. fifteen (15) days after receipt of such notice the Partnership shall advise the District if it reasonably believes that the proposed Non-Baseball Event will harm the integrity of the playing field or if the proposed Non-Baseball Event cannot be held due to a conflict with a Partnership Non-Baseball Event previously approved by the District. The failure by the Partnership to so notify the District shall be deemed an affirmative statement by the Partnership that it does not reasonably believe that the proposed Non-Baseball Event will harm the integrity of the playing field and that no such scheduling conflict exists. If the Partnership does not so notify the District, within thirty (30) days after receipt the notice from the District provided for in this section 5.2(d) the Partnership shall provide the District with a good faith

estimate of managerial and operational costs which will be incurred by the Partnership as a result of any Off-Season, Non-Baseball Event, which shall not include prorated salaries or incremental payments to employees of the Partnership or its Affiliates not specifically incurred as a result of such event. Within thirty (30) days after receipt of an itemized statement of such actual costs from the Partnership, the District shall reimburse the Partnership for all reasonable operational and managerial costs incurred as a result of any Off-Season, Non-Baseball Event.

- The District reserves for itself and shall be entitled, at no expense, to office space which will be incorporated into the structure of the Stadium on the 4th floor of the Student Movers building. Any change in the location of the District's office space after the Commencement Date shall only be with the approval of the Partnership, which approval shall unreasonably withheld. Attendant to this office space, the District will also have, at no expense, the use of parking spaces (as agreed to between the District and the Partnership or, if no such agreement can be reached, as reasonably designated by the District) for its office staff. The Partnership is obligated to maintain and pay all costs of maintenance and utilities for the District's office space pursuant to the provisions of Article VIII. The Stadium security system and parking plan developed by the Partnership shall not cause unreasonable interference with the ingress and egress of any of the District's employees, agents, or representatives to and from the Leased Premises, necessary for the District to perform all of its operations.
- (f) The District, at no expense, will have use and control of the community rooms to be constructed adjacent to the District's offices on the 4th floor of the Student Movers building or as otherwise agreed by the parties. The District shall manage

the use of the community rooms and, subject to the Stadium security system developed by the Partnership, shall be entitled to allow outside community groups to use the community rooms as the District deems appropriate. Nothing herein shall be construed to preclude use of the community rooms by the Partnership or its designees. The Partnership is obligated to maintain and pay all costs of maintenance and utilities for the community rooms pursuant to the provisions of Article VIII.

- 5.3 <u>Non-Baseball Events</u>. A Non-Baseball Event shall be any use of the Leased Premises that includes the use of any portion of the seating bowl, the playing field, or both, and that is not a Major League Baseball Game and is not held immediately preceding or following a Major League Baseball Game, including, by way of example and without limitation, concerts, fireworks displays, shows, fairs, markets, and exhibitions unless such events immediately precede or follow a Major League Baseball Game.
- 5.4 <u>Management</u>. The Partnership shall be exclusively responsible for the management and operation of the Leased Premises pursuant to the terms and conditions of Article VIII.

ARTICLE VI

INSURANCE AND SUBROGATION

6.1 Type of Insurance.

- (a) The District, through its general contractor, shall, prior to the Commencement Date, during the entire construction period, cause to be maintained in full force and effect, at the general contractor's expense, all insurance coverages which the District has, in its sole discretion, deemed appropriate to protect the District until the Commencement Date.
- (b) Upon the occurrence of the Commencement Date or at such earlier time as requested by the District so that there is no lapse in insurance coverage between that referenced in section

6.1(a) and that to be provided by the Partnership pursuant to this Agreement, the Partnership shall have purchased and shall maintain such insurance as will protect it and the District, the District's Board, employees, Executive Director, agents, attorneys and representatives, and Equipment Lessors from all claims arising as a result of the ownership, use, management, and operation of the Leased Premises and Partnership Property, all as may be typical in coverage and amount to similar facilities including but not limited to the following:

- Claims under worker's compensation, disability benefit and other similar employee benefits acts and employer's liability;
- Claims for damages because of bodily injury, sickness or disease, or death of employees;
- 3. Claims for damages because of bodily injury, sickness or disease, or death of persons other than employees;
- 4. Claims for damages because of damage to property owned by third parties;
- 5. Claims for damages insured by personal injury liability coverage which are sustained by third parties, including libel, slander, false arrest, detention or imprisonment, malicious prosecution, and wrongful eviction from, wrongful entry into, or invasion of the right of private occupancy of a premises that a person occupies by or on behalf of its owner, landlord or lessor;
- 6. Claims for damages (including, in the Partnership's reasonable discretion, loss of income and rents due to business interruption) because of injury to or

- destruction of tangible property, including loss of use resulting therefrom;
- 7. Claims for loss of or damage to boiler and machinery (including but not limited to heating, ventilation, air conditioning, scoreboard(s), electrical system equipment and all other equipment and machinery).

6.2 <u>Insurance Specifications</u>.

- The Partnership agrees to Liability Insurance. rchase, secure and deliver to the District and to keep in force all times during the Term and Grace Term, a liability insurance policy, including public liability, personal injury and property damage, in a form and with a company acceptable to and reasonably approved by the District covering all Leased Premises and partnership Property operations, including comprehensive Auto lability and Property Damage Insurance and Commercial General Liability and Property Damage Insurance in amounts to be determined by the Partnership in consultation with its advisers. Such amounts shall be consistent with the amounts of liability and property damage coverages customarily maintained by other Major League Baseball clubs, but in no event will the limits of public liability \$1,000,000.00 per occurrence than insurance less \$10,000,000.00 aggregate. The aggregate limit of insurance must apply on a per occurrence basis. Commercial Liability Coverage shall include blanket contractual liability and shall include the its Board, employees, Executive Director, District. attorneys and representatives and the Equipment Lessors additional insureds, with primary coverages as respects the District and the Equipment Lessors.
 - (b) <u>Property Insurance</u>. The Partnership shall purchase and maintain in the name of the District, special form insurance

upon the entire Leased Premises and, in the name of the Partnership with the District named as loss payee, upon the Partnership Property to the full insurable value thereof, with reasonable deductibles. The Equipment Lessors shall be named as joint loss payees with the District as to the Leased Equipment. coverage shall include the Stadium (including Partnership Property), all materials and supplies of any nature included in the Leased Premises (including personal property, fixtures equipment) and all materials and supplies of any nature whatsoever to be used on the Stadium Land (including personal property, fixtures and equipment) whether any or all of the foregoing are located on the Leased Premises, in transit or while temporarily stored elsewhere. The coverage shall be replacement value for "risk of physical loss" with exceptions as reasonably approved by the District.

- (c) <u>Umbrella Policy</u>. The Partnership shall purchase and maintain in its name, the name of the District, its Board, employees, Executive Director, agents, attorneys, other representatives, and the name of the Equipment Lessors an "umbrella" excess liability insurance policy in an amount per occurrence to be determined by the Partnership in consultation with its insurers. Such amounts shall be consistent with the amounts of umbrella liability insurance coverage customarily maintained by other Major League Baseball clubs.
- 6.3 <u>District Approval</u>. The form, terms, amounts, and provisions of the insurance policies (including but not limited to all exclusions in such policies) required by this Article VI shall be subject to the reasonable approval of the District.
- 6.4 <u>District Copies</u>. A certified copy of the above policy or policies, or certificates evidencing the existence thereof, all in such form as the District may reasonably require, shall be

delivered to the District upon the Commencement Date or such earlier date as the insurance coverage is required hereunder.

- Third-Party Insurance. The Partnership shall require the concessionaires, any of the concessionaires' subcontractors, any contractor of the Partnership performing services related to the use, operation, and management of the Leased Premises Partnership Property, and any other user of the Leased Premises and Partnership Property for a Non-Baseball Event, to maintain worker's compensation insurance additional and insurance substantially the same as that coverage set forth in sections 6.1 and 6.2, except that, in the Partnership's reasonable discretion, the foregoing entities need not provide boiler and machinery coverage. The District and its Board, employees, Executive Director, agents, attorneys and representatives shall be named as additional insureds in all such policies. The provisions of sections 6.3, 6.4, 6.7, 6.8 and 6.9 apply to all such policies.
 - 6.6 Repairs and Improvements Insurance.
- Contractors and Subcontractors. The Partnership shall require all contractors and subcontractors hired in the performance of any Repairs or Improvements to maintain insurance under the terms and conditions set forth in this Article VI, except that the property insurance to be maintained by such contractors and subcontractors shall be all risk builder's coverage. All such policies shall include coverage for loss due to (i) the Leased Premises or Partnership Property being made unavailable because of delay or damage, and (ii) cost overruns. All such policies shall name the District as loss payee and the District and its Board, Executive Director, agents, attorneys, representatives as additional insureds. The provisions of Sections 6.3, 6.4, 6.7, 6.8 and 6.9 shall apply to such policies.

- Architect's Errors and Omissions. As to any Repairs or Improvements to be performed hereunder, the Partnership shall require any architect and any other person or entity doing any type of design work to purchase and maintain liability insurance for damages for claims by reason of any negligent act, error or omission committed or alleged to have been committed by such Architect, consultant, engineer, or other person or entity performing design work relating to any Repair or Improvement in an appropriate amount as determined by the Partnership and subject to the District's reasonable approval. Professional errors and omissions insurance coverage shall be on an occurrence basis if available but may be structured to provide coverage on a "claims made" basis; provided, however, that the Partnership requires the architect to maintain and keep in effect said professional errors and omissions insurance coverage until substantial completion of any Repair or Improvement, and three (3) years extended discovery. All such policies shall name the District as loss payee and the District and its Board, employees, Executive Director, agents, attorneys, and representatives as additional insureds.
- 5.7 <u>District Approval of Underwriter</u>. The Partnership, the concessionaires or any contractor or subcontractor of the Partnership or the concessionaires shall obtain the prior approval of the District regarding any company or other entity proposed to underwrite any insurance coverage to be provided pursuant to this Article VI, which approval shall not be unreasonably withheld.
- 6.8 Notice of Cancellation or Change. All policies shall provide that they may not be canceled or materially changed or altered without first giving at least thirty (30) days written notice thereof to the District and, as to policies under which they are named as additional insureds or joint loss payees with the District, the Equipment Lessors. The insurance policies required

hereunder shall be reviewed by the District and the Partnership on an annual basis to determine the adequacy of the coverage amounts.

- 6.9 Waiver of Subrogation. The District and the Partnership agree that all insurance policies against loss or damage to property and business interruption or rent loss, and all liability insurance policies required hereunder, shall be endorsed to provide that any release from liability of, or waiver of claim for, recovery from the other party entered into in writing by the insured thereunder prior to any loss or damage shall not affect the validity of said policy or the right of the insured to recover thereunder. Such insurance policies shall further provide that the insurer waives all rights of subrogation which such insurer might have against the other party. Without limiting any release or waiver of liability or recovery contained in any other section of this Agreement, but rather in confirmation and furtherance thereof, each of the parties hereto waives all claims for recovery from the other party for any loss or damage to any of its property or damages as a result of business interruption, rent loss or liability of the types covered in this Article VI, insured under valid and collectible insurance policies to the extent of any recovery collectible under such insurance policies.
- 6.10 Bonds. The Partnership shall ensure that the bond requirements set forth in Sections 38-26-105, 38-26-106, and 38-26-107 of the Colorado Revised Statutes are complied with in relation to any Repair or Improvement.
- 6.11 <u>District's Governmental Immunity</u>. The parties recognize that the District is a Colorado governmental entity within the scope and purview of § 24-10-101, <u>et seq.</u>, of the Colorado Revised Statutes and has not taken steps, by resolution or otherwise, to waive its governmental immunity or to increase or modify its statutory limits of liability of \$150,000/\$600,000.

6.12 <u>Application of Insurance Proceeds</u>. The proceeds of any claims made under the insurance policies required by Section 6.6 shall be the property of the District and shall be used to correct the damage upon which the claim was based, unless otherwise agreed to by the District and the Partnership.

ARTICLE VII

INDEMNIFICATION

7.1 Partnership Indemnification. Subject to the limitations hereinafter set forth, the Partnership hereby agrees, upon its occupancy and use of the Leased Premises, to protect, defend, indemnify and hold harmless the District, its officers, Board members, employees, agents, attorneys, representatives, and the Equipment Lessors from and against all loss, cost, expense and obligation including but not limited to reasonable attorneys' and consultation fees in connection with or arising out of any proceedings, judicial or otherwise, claims, demands, judgments, damage or injury to person or property occurring within or about the Leased Premises arising out of or in connection with the Partnership's operation, use, Maintenance, ownership, Improvement, or Repair of the Leased Premises or Partnership Property or any events occurring thereon, or performance of the Partnership's other obligations under this Agreement. Notwithstanding anything to the contrary contained in this Agreement, in no event shall the Partnership be obligated to protect, defend, indemnify or hold harmless the District, its officers, Board members, employees, agents, attorneys or representatives from or against any loss, cost, expense, obligation, claim, demand, judgment, damage or injury to the extent the same arises out of, or is connected with:

- (a) the negligence or misconduct of the District, its Board members, Executive Director, employees, agents, attorneys, sponsors, promoters, representatives or independent contractors;
- (b) the failure of the District to perform or observe any covenant or condition to be performed or observed by the District under this Agreement; or
- (c) defects in (i) the design of the Leased Premises; or (ii) the workmanship or materials employed in the construction of the Leased Premises.

Notwithstanding anything to the contrary contained in this Agreement, in no event shall the Partnership be obligated to protect, defend, indemnify or hold harmless the Equipment Lessors from or against any loss, cost, expense, obligation, claim, demand, judgment, damage or injury to the extent the same arises out of, or is connected with the negligence or misconduct of the Equipment Lessors.

District Indemnification. The District shall make available to the Partnership any indemnities under which the District is entitled to indemnification from any and architects, engineers, consultants, contractors or subcontractors employed by the District or the Partnership in the construction of the Stadium, the Leased Premises and the Partnership Property. The Partnership shall, at its expense, pursue all claims for indemnity directly against such architects, engineers, consultants. contractors or subcontractors.

7.3 <u>District Cooperation</u>

(a) The District shall cooperate with the Partnership and its counsel in any action being actively contested or defended by the Partnership pursuant to its obligations under Section 7.1, shall join in making any appropriate counterclaim or cross-claim in connection with the actions, and shall provide such access to the

books and records of the District as shall be necessary in connection with such defense or contest, all at the sole cost and expense of the Partnership. The Partnership shall provide its defense or contest through attorneys, accountants, and others selected by the Partnership with the reasonable concurrence of the Notwithstanding that the Partnership is actively conducting such defense or contest, any such action may be settled, compromised, or paid by the District without the consent of the Partnership, provided however, that if such action is taken without the Partnership's consent, its indemnification obligations in respect of such claim shall thereby be nullified. Any such action may be settled, compromised, or paid by the Partnership without the District's consent, so long as such settlement or compromise does not cause the District to incur any present or future cost, expense, obligation or liability of any kind or nature with respect to such action.

The District shall also cooperate with the Partnership in its pursuit of any indemnity under which the District is entitled to indemnification from any architect, engineer, consultant, contractor or sub-contractor.

(b) In the event any action involves matters partly within or partly outside the scope of the indemnification by the Partnership hereunder, then the attorneys' fees, costs, and expenses of contesting or defending such action shall be allocated between the District and the Partnership as they shall agree.

ARTICLE VIII

MAINTENANCE, REPAIRS, AND OPERATIONS

8.1 <u>Definitions</u>. As used in this Article VIII and elsewhere in this Agreement, the terms listed below shall have the following meanings:

(a) Stadium Management. "Stadium Management" means the planning, supervision and conduct of the day-to-day management of the Leased Premises and Partnership Property and all activities connected with the operation of the Leased Premises and Partnership Property on a year-round basis, including but not limited to the provision of (or arrangements for third parties to provide) all personnel, supplies, equipment and services necessary for Repairs, Maintenance, Improvements and the security required by Article XVIII on the Leased Premises and Partnership Property, which are the responsibility of the Partnership as provided in this Article VIII.

دور

- means any equipment, fixture, furnishing, facility, surface, structure or any other component of a Repair Area which has become dysfunctional due to ordinary wear and tear. (For purposes of this Section 8.1(b), any equipment, fixture, furnishing, facility, surface, structure or any other component shall be deemed dysfunctional if such equipment, fixture, furnishing, facility, surface, structure or any other component has deteriorated to a degree that cannot be remedied through Maintenance.)
- (c) Repairs. "Repair" or "Repairs" means any work (including all labor, supplies, materials and equipment) reasonably necessary to repair, restore, or replace any equipment, facility, structure or any other component of a Repair Area, if such work is necessitated by: (i) any material defects in design, construction or installation which, if not repaired, would create a threat to health or safety or a material interference with the Team's ability to play any Major League Baseball Game in the Stadium; (ii) Deterioration (including replacement necessitated by the repeated breakdown of a component despite successive efforts to repair or restore it short of such replacement); (iii) requirements initiated

and imposed prospectively by The National League or the Commissioner of Baseball, which requirements are made of all National League open-air stadiums; (iv) modifications required by city, state or federal laws, ordinances, rules, or regulations, including without limitation, accommodations required to be made to the Leased Premises or Partnership Property under the Americans with Disabilities Act (unless such modifications are for components negligently excluded by the design consultants from the original construction); and (v) requirements of any insurance carrier, which requirements are necessary to enable the Partnership to obtain insurance coverage at commercially reasonable rates.

If any Repair is necessitated due to the gross negligence or intentional or willful misconduct of the Partnership, the Partnership shall not be entitled to reimbursement or advances from the Repairs Fund for the cost of such Repair.

- (d) <u>Repair Areas</u>. "Repair Area" means the Leased Premises and Partnership Property and the contiguous sidewalks set forth in Exhibit F attached hereto.
- (e) Maintenance. "Maintain" and "Maintenance" means all work (including all labor, supplies, materials and equipment) reasonably necessary for the cleaning and routine upkeep of any property, structures, surfaces, facilities, fixtures (including but not limited to media plug-ins and cable and all wiring attendant thereto), equipment or furnishings, or any other component of a Repair Area in order to preserve such items in their existing condition, ordinary wear and tear excepted. By way of illustration, and without limiting the generality of the foregoing, Maintenance shall include: (i) preventive or periodic maintenance procedures for equipment, fixtures or systems; (ii) periodic testing of buildings systems, such as mechanical, card-key security, fire alarm, lighting, and sound systems; (iii) ongoing

trash removal; (iv) regular maintenance procedures for HVAC, plumbing, mechanical, electrical and structural systems, such as periodic cleaning, lubrication, and changing of air filters and lights; (v) touch up painting; (vi) cleaning prior to, during and following all games, baseball-related events and Non-Baseball Events and (vii) any other work of a routine, regular and generally predictable nature that is reasonably necessary in order to keep the Leased Premises in good order and condition. As used in this Article VIII, the term "Maintenance" shall not include any work defined in Section 8.1(c) as a "Repair".

- (f) <u>Damaged Property</u>. "Damaged Property" means any equipment, facility, structure, furniture, fixture or any other component of a Repair Area which is damaged or destroyed, including any damage or destruction resulting from the acts or omissions of third parties (including licensees or invitees of the Partnership or the Team).
- (g) <u>Improvements</u>. "Improvements" or "Improvement" means any alterations, modifications, additions, or other construction, development, or improvement of any part of the Leased Premises or Partnership Property which the parties jointly determine are desirable for a first-class Major League Baseball facility.

8.2 Partnership Obligations

(a) General Obligations. The Partnership is hereby designated operator/manager of the Leased Premises and shall be responsible for all Stadium Management unless such responsibility is terminated by the District pursuant to Section 8.8(e). Subject to the provisions of Section 8.10, the Partnership shall at its own expense, Repair and Maintain and operate all Repair Areas. The Partnership shall obtain or provide, at its expense, all labor, services, materials, supplies and equipment needed to perform all Stadium Management, Maintenance, Improvements, and security. The

Partnership shall not reduce, diminish or eliminate any equipment or services that it is required to provide under this Agreement without the prior written consent of the District. The District shall provide the Partnership with a copy of all plans, specifications and other documents in the control of the District necessary to assist the Partnership in fulfilling its obligations pursuant to this Section 8.2. The Partnership shall enter into and complete negotiations with the City and County of Denver, regarding maintenance of pedestrian lighting, trees, trash receptacles and sidewalks contiguous to the Leased Premises. The Partnership agrees that the costs of such maintenance shall not be borne by the District.

- (b) <u>Standards</u>. The Partnership shall Maintain, operate and Repair the entire Leased Premises, Partnership Property and Repair Area in a safe and first-class manner.
- (c) Cleaning and Trash Removal. The Partnership shall provide ongoing sweeping and trash removal services, and shall provide Maintenance (including cleaning) of the Leased Premises, Partnership Property and Repair Areas after all Major League Baseball and Non-Baseball Events, and at all other times, as necessary, during the Term and Grace Term, so as to Maintain the Leased Premises, Partnership Property and Repair Areas in conformity with the standard set forth in Section 8.2(b).
- (d) <u>Emergency Maintenance</u>. The Partnership shall provide for all necessary emergency maintenance of mechanical, electrical and plumbing facilities of the Leased Premises and Partnership Property which directly affect the public's safe access to or use of the Leased Premises and Partnership Property, including but not limited to the Stadium's elevators and escalators, for the duration of each Major League Baseball Game and

all Major League Baseball and Non-Baseball Events using the playing field or stands.

- (e) Conduct of Repairs and Maintenance. The Partnership shall cause all Maintenance and Repairs required to be performed by the Partnership under this Article VIII to be performed promptly and diligently, and in a good and workmanlike manner. The Partnership shall plan, schedule and conduct such Maintenance and Repairs so as to prevent or at least minimize (i) inconvenience to any patrons, (ii) any reduction in seating capacity at the Stadium, (iii) any reduction in the number of spaces available in the parking facilities, and (iv) interference with the District's or Partnership's use and enjoyment of the Leased Premises.
- (f) <u>Partnership's Property</u>. The Partnership shall be responsible, at its sole expense, for repair, operation and maintenance of the Partnership Property, except for repair of such property necessitated by damage caused by the District.
- Game and Event Operations. On the dates of Major League Baseball Games and all baseball and Non-Baseball Events which are open to the public and use the playing field or stands, beginning with the time of day by which an admissions ticket to such games or events is required in order for the public to enter the Stadium or the Leased Premises, the Partnership shall be responsible for: (i) providing and supervising all personnel, including ushers; (ii) providing crowd control and management within the Leased Premises; (iii) providing and supervising firstaid personnel to operate the first-aid facilities of the Leased Premises; (iv) working cooperatively with the City and County of Denver to provide event security and emergency medical assistance such games and events; (v) implementation transportation management program substantially similar to that developed by the District's consultants in cooperation with the

Partmership, neighborhood groups representing neighborhoods affected by the use of the Leased Premises and the Partnership Property, and the City and County of Denver, a copy of which is attached hereto as Exhibit G (the "Transportation Management (The Partnership shall enter into and complete Program") negotiations with the City and County of Denver, regarding the necessary personnel to provide additional traffic control and comply with all other requirements of the City and County of Denver during Major League Baseball Games, baseball related events and Non-Baseball Events. The Partnership agrees that the costs of all personnel and all other costs associated with the provision of additional traffic control and compliance with other requirements of the City and County of Denver shall not be borne by the District.); and (vi) providing all other services and materials necessary to fulfill its obligations under this Agreement.

This list is provided by way of example only and is not a limitation upon the Partnership's Stadium Management responsibilities or any other obligations of the Partnership under this Agreement.

- (h) Notice to District. The Partnership shall promptly notify the District (orally first, then by written notice if written notice alone is not practicable) whenever the Partnership knows of or discovers any material defects in, damage to, or destruction of any part of the Leased Premises, Partnership Property or Repair Areas, or any dangers or hazards on the Leased Premises, Partnership Property or Repair Areas.
- (i) Obligations Regarding Damaged Property. The Partnership shall provide insurance, pursuant to the provisions of Article VI, to cover the repair, restoration or replacement of any Damaged Property. The Partnership shall be responsible for the payment of all deductibles attendant to such insurance. To the

extent that such insurance is insufficient to cover the costs of any repair, restoration or replacement of any Damaged Property, the Partnership shall be solely responsible for such costs (including the costs of all labor, supplies, materials and equipment). No moneys in the Repairs Fund shall be used for the repair, restoration or replacement of any Damaged Property if the damage is the result of the gross negligence or intentional or willful misconduct of the Partnership.

(j) <u>Public Art</u>. Notwithstanding anything contained in this Article VIII to the contrary, the District shall create a fund in the amount of \$25,000.00, derived from public donations for art work, which shall be used by the Partnership solely for the purposes of maintaining and refurbishing the public art work paid for by the District and located on the Stadium Land. The Partnership shall not be responsible for any maintenance costs related to such art work in excess of this amount.

8.3 Playing Field Maintenance and Repairs.

- (a) <u>Playing Field</u>. As used in this Article VIII, "Playing Field" shall mean the infield grass, the outfield grass, the infield skinned area, the ground lying to the foul side of the foul lines of the playing field, the warning track, the sod farms, the dugouts, and the bullpens.
- (b) <u>Major League Baseball Season and Off-Season</u>
 <u>Obligations</u>.
- (i) <u>During Major League Baseball Season</u>. During any Major League Baseball Season, the Partnership shall be responsible, at its expense, for the day-to-day landscaping and Maintenance of the Playing Field and Stadium Land, and for the costs of all supplies in connection with such day-to-day landscaping and maintenance (such as the cost of fertilizer and grass seed). Nothing in this Section 8.3(b)(i) shall be

interpreted to mean that any Repairs of the Playing Field, Stadium Land or any part thereof or any repair, restoration or replacement of any part of the Playing Field which becomes Damaged Property, shall be included in the Partnership's day-to-day landscaping and Maintenance responsibilities.

Off-Season Non-Baseball Events. (ii) The Partnership shall be responsible for performing and completing any day-to-day landscaping and maintenance of the Playing Field and Stadium Land required in connection with Off-Season Non-Baseball Events, including any necessary preparation and conditioning of the Playing Field before or during such Events, and the landscaping and maintenance (including sodding and seeding) as may be required in the Partnership's reasonable judgment after such Events in order to restore the Playing Field to first-class major league condition. Such restoration shall be completed by the beginning of the next Major League Baseball Season after such Non-Baseball Events occurring during the Off-Season. For Off-Season Non-Baseball Events held by the District, the District shall promptly reimburse the Partnership (or cause a promoter or other third party to reimburse the Partnership) for the Partnership's personnel and material costs in providing such landscaping and maintenance and for reasonable costs associated with the repair or restoration of any portion of the Playing Field which becomes Damaged Property as a result of an Off-Season Non-Baseball Event held by the District. The Partnership shall supply the District and any promoters or third parties designated by the District with itemized invoices for such personnel and material costs, including supporting invoices for costs incurred in acquiring materials. If the District believes that the Partnership is failing to perform its obligations under this Section 8.3(b)(ii), the District shall promptly notify

the Partnership, and the Partnership shall then be obligated to cause its groundskeepers to perform such duties.

- Partnership's Failure to Maintain or Repair. Partnership fails to fulfill its obligations under this Article VIII, and if the District reasonably determines that the condition in question, if not promptly remedied, will pose a threat to the health or safety of persons or property, the District shall promptly notify the Partnership in writing of such failure, explaining the reasons why the condition in question meets the standard set forth in this Section 8.4, and give the Partnership a reasonable opportunity to perform the Maintenance or Repair in If the Partnership fails to perform the work in a question. reasonably prudent time and manner despite such written notice from the District, the District may (in addition to any other rights which it may exercise hereunder), at its election, cause such Maintenance or Repair to be performed at commercially reasonable rates and advance the costs for doing so. The District shall thereupon promptly notify the Partnership, in writing, of the nature of the work performed by the District (or its agents or contractors) and the cost and expense incurred by the District in performing such Maintenance or Repair. Within thirty (30) days after receiving the District's written notice the Partnership shall: (i) reimburse the District for all such costs and expenses, or (ii) inform the District, in writing that the Partnership disputes the District's entitlement to charge such costs or expenses, in which case the District may consider the Partnership in default and proceed pursuant to the provisions of Article XI.
 - 8.5 <u>Utilities</u>. The Partnership shall bear the cost of all utilities consumed or used in or on any Repair Area or in connection therewith (such as, by way of example and without limitation, gas, steam, electricity, water, sewer, telephone, cable, trash

collection, etc.), except such costs associated with any Off-Season Non-Baseball Event held by the District, which shall be paid by the District or other user of the Leased Premises or Partnership Property pursuant to Section 5.2(d). The parties shall endeavor to avoid waste in the use of such utilities, in the interest of conserving resources. If there is any interruption in utility services affecting the Leased Premises or Partnership Property not caused by the District or its employees, Board members, agents or representatives, the District shall not be responsible for such interruption, and shall endeavor to actively assist the Partnership in arranging, with the appropriate providers of such utilities, for the prompt and complete restoration of such service. In addition, if and to the extent that the District has any cause of action for damages or any other remedy against the providers of utilities for losses caused by the interruption in utility services on the Leased Premises or Partnership Property, the District shall assign its right to pursue such cause of action or other remedies to the Partnership. Notwithstanding the foregoing, the Partnership shall provide, Maintain, Repair and regularly test the Stadium's emergency electrical back-up systems.

8.6 Improvements and Alterations by the Partnership.

(a) Improvements by Partnership. During the Term and Grace Term, the Partnership, with the District's prior written approval, may make any improvements to the Leased Premises which do not (i) adversely affect the Stadium aesthetics, structure or its systems as determined by the District; (ii) materially increase the cost of Repairs to the Leased Premises or Partnership Property; (iii) violate any laws, ordinances, or regulations; or (iv) have a Material Adverse Effect on the District. In constructing any such improvements, the Partnership shall comply with and is subject to the provisions of this Agreement, including Sections 8.10, 8.15 and

- 8.16. Notwithstanding anything to the contrary contained in this Agreement, the Partnership shall not be entitled to any advances or reimbursements from the Repairs Fund for any such improvements constructed on the Leased Premises, unless expressly agreed to, in writing, by the District. If any such improvement is determined to be Partnership Property, as agreed to by the District and the Partnership, such improvement shall be subject to the purchase, request for removal and abandonment provisions set forth in Section 2.4. Any such improvement constructed on the Leased Premises not so determined by the District to be Partnership Property shall become a part of the real property constituting the Leased Premises and title thereto shall be held by the District or its successors in interest.
- Alterations to the Partnership's Offices. (b) Partnership may make any interior alterations, additions or improvements to the Partnership's offices, at the Partnership's and expense, provided such alterations, additions or improvements do not (i) adversely affect the Stadium aesthetics, structure, or its systems, as reasonably determined by the District; (ii) materially increase the cost of Repairs to the Leased Premises; (iii) violate any laws. ordinances. regulations; or (iv) have a Material Adverse Effect on the The Partnership shall provide the District a written District. notice at least thirty (30) days prior to commencing any such alterations, additions or improvements, together with reasonably detailed plans and specifications for the proposed work and estimated costs. The District's receipt of such plans and specifications shall not constitute any assumption by the District of any responsibility for their accuracy or sufficiency, which shall be the sole responsibility of the Partnership. alterations, additions or improvements shall be performed in a good

and workmanlike manner, shall not cause damage to any part of the Stadium, and shall be completed in accordance with all applicable legal requirements (including the procurement of any required governmental permits). Before commencing any such alterations, additions or improvements, the Partnership shall obtain and furnish to the District, proof of compliance with all appropriate insurance and bonding requirements as set forth in Article VI, as agreed to by the District and the Partnership. The form of such bond and insurance policy and the surety or insurance carrier must be reasonably acceptable to the District. In connection with any such alterations, additions or improvements, the Partnership shall also agree in writing to indemnify the District (or make whatever alternative arrangements are reasonably required by the District) with respect to any liens or claims or notices for liens that may filed against the Stadium, the Leased Premises, Partnership's leasehold interest in the Leased Premises arising out of such alterations, additions or improvements. The affirmative covenants set forth in Section 8.16 shall also apply to any such alterations, additions or improvements of the Partnership's offices performed in accordance with this section.

shall not make any alterations, additions or improvements, whether interior or exterior, structural or non-structural, to any area of the Leased Premises other than the Partnership's offices without the District's prior written consent, which shall not be unreasonably withheld. Notwithstanding anything to the contrary contained in this Agreement, the Partnership shall not be entitled to any advances or reimbursements from the Repairs Fund for any alterations, additions or improvements constructed on the Leased Premises, unless expressly agreed to, in writing, by the District.

- Alterations Made Without the District's Consent. If the Partnership makes any alterations, additions or improvements,, to any part of the Stadium, except the interior of Partnership's offices, without the District's prior approval, or any alterations, additions or improvements to the interior of the Partnership's offices without prior notice to the District, in violation of this Section 8.6, the District shall determine whether such alterations, additions or improvements shall remain part of the Stadium or be removed. If the District determines that such alterations, additions or improvements shall be removed, Partnership, at its cost and expense, shall cause their prompt removal, and shall restore the Stadium to its condition prior to any such alterations, additions or improvements. In addition, if the Partnership cuts through or pierces or alters in any manner the roof or exterior walls of the Stadium structure without first obtaining the District's written consent, the Partnership shall be obligated, at the District's request, to restore the roof and exterior walls to their prior condition at the Partnership's sole cost and expense. Further, the Partnership may be liable to the District for consequential damages caused as a result of such breach by the Partnership.
- 8.7 Partnership's Personal Property. Prior to the installation of any and all safes, telephone switchboards, computers, or other heavy equipment which may be placed or installed at the Stadium by the Partnership following the Commencement Date, the Partnership shall obtain an engineering report, satisfactory to the District, regarding the affect of such placement or installation upon the Stadium structure and shall comply with all provisions contained in such report, including, but not limited to, all provisions regarding the proper placement or installation of such equipment. The Partnership shall perform any

reinforcing required for such installation at the Partnership's cost and expense. The Partnership shall install and place, at its own expense, all items of personal property, fixtures or leasehold improvements which the Partnership installs, places, or maintains in the Stadium pursuant to this Agreement, and shall install and place such property, fixtures or improvements in settings which are sufficient, in the District's reasonable judgment, to absorb and prevent vibration in or damage to the Stadium. If such placements or installations cause such vibration or damage, the Partnership shall, at its expense, take such steps as the District may reasonably direct to remedy any such condition.

8.8 Partnership's Management Oualifications.

- Standards. The District has the obligation (pursuant to § 32-14-125 of the Act) to ensure that the Partnership has experience, expertise and specialization in the management of and operation of sports, entertainment or convention facilities sufficient to perform its Stadium Management The Partnership has demonstrated to the District that the Partnership, through its own staffing or contracts with third parties, has experience, expertise and specialization in, without limitation:
- (1) Maintenance of all HVAC, electrical, plumbing, and communications support systems;
 - (2) Custodial and groundskeeping operations;
 - (3) Development and implementation of security procedures for the physical protection of personnel and property on the Leased Premises:
 - (4) Management of all administrative functions associated with the Leased Premises, including, but not limited to, operation of any computer based maintenance management system, administration of the purchasing system, coordination and reporting

of payroll and employee benefits, and proper supervision of all clerical staff;

- (5) Management and planning of all Maintenance and Repair operations;
- (6) Management of all audio, video, telecommunications and scoreboard facilities;
 - (7) Management of all concession operations; and
 - (8) Management of all parking operations.

Based upon this demonstration, the District has approved, by resolution of the Board, the Partnership as manager and operator of the Leased Premises.

- (b) Operations Manual. No later than March 31, 1995, the Partnership shall present to the District a preliminary draft of a Stadium Operations Guidelines and Policy Manual (the "Operations Manual") reflecting the Partnership's proposed plan for implementing the proper operation of the Stadium, Partnership Property and Leased Premises. The final Operations Manual shall be completed no later than July 1, 1995, and shall include a delineated policy regarding the use of all lighting on the Leased Premises which policy address the concerns of the neighborhoods affected by the Stadium to have all lighting, except that related to security purposes, extinguished as soon as practicable following any event held upon the Leased Premises.
- (c) <u>Maintenance Manual</u>. No later than March 31, 1995, the Partnership shall present to the District a preliminary draft of a Maintenance and Procedures Manual (the "Maintenance Manual") which details proposed policies and practices the Partnership intends to use to Maintain the mechanical and electrical equipment within the Stadium, Partnership Property and Leased Fremises. The final Maintenance Manual shall be completed no later than July 1, 1995, and shall detail the routine and normal maintenance

procedures to be used on each and every piece of equipment and the procedures for monitoring compliance.

- (d) On-going Oualifications. The Partnership must at all times during the Term and Grace Term remain qualified to manage the Leased Premises as a first class facility. So that the District can adequately discharge its duties pursuant to Section 32-14-125 of the Act, the Partnership shall notify the District of significant changes in the management of the Partnership's maintenance and operation of the Leased Premises and Partnership Property or any part of such maintenance and operation, as set forth in Section 8.8(a).
- Termination of Stadium Management Responsibilities. If during the Term or Grace Term, the District determines that the Partnership has materially defaulted in its obligation to perform or be qualified to perform all Stadium Management, the District shall notify the Partnership of such default in accordance with the provisions of Section 27.5. The Partnership shall have the right to contest, only via the arbitration process described herein, whether a default has occurred, is material, or has been cured. order to contest whether a default has occurred or is material, the Partnership must provide written notice to the District of its claim that the alleged default did not occur or is not material within thirty (30) days after receipt of the notice of such default from the District. In order to claim the default has been cured, the Partnership must send written notice making such claim within thirty (30) days after the alleged cure or the notice provided by the District, whichever occurs later.

All material defaults shall be cured by the Partnership within thirty (30) days after notice from the District or within one (1) year after notice from the District if the default (i) is not

capable of cure within thirty (30) days and (ii) the Partnership is working diligently toward cure.

In the event of a dispute between the District and the Partnership as to whether a default has occurred, is material, or has been cured, the District and the Partnership agree to submit those matters to binding arbitration. The arbitration shall be limited to those issues and any procedural or evidentiary issues that may arise in attempting to resolve those issues.

The arbitration shall occur in Denver, Colorado pursuant to \$\$13-22-201 et seq, of the Colorado Revised Statutes, except where the provisions of that article conflict with this Agreement. Colorado substantive law shall control.

Any arbitration pursuant to this Agreement shall be conducted before the Judicial Arbiter Group ("JAG"). If for some reason JAG cannot or will not conduct the arbitration and the District and the Partnership are unable to agree on another private company providing services comparable to JAG, then the American Arbitration Association ("AAA") shall conduct the arbitration.

Upon a disagreement between the Partnership and the District regarding default, the materiality of a default, or whether a default has been cured, either party may commence the arbitration by written demand to the other party with a copy to JAG. JAG shall provide to the parties, within fifteen (15) days after the demand, a list containing an odd number of the names of at least five (5) potential arbiters. The parties, beginning with the Partnership, shall each strike one name in turn until only one (1) name remains. If it is one party's turn and that party does not strike a name within three (3) business days, then that turn is forfeited, and the other party shall strike the next name. When only one (1) name remains, the remaining person shall be the single arbiter.

Once the arbiter is selected, an arbitration hearing shall be commenced in no more than one hundred twenty (120) days. The Colorado Rules of Evidence, as then in effect, shall control. Discovery shall be permitted as allowed by the Colorado Rules of Civil Procedure, as then in effect, except as those Rules conflict with this Agreement. Scheduling of such discovery may be determined by the arbiter, and any discovery disputes shall be finally determined by the arbiter.

In any arbitration pursuant to this Agreement, the arbiter shall decide (either after reviewing only documents, or with a hearing, at the discretion of the arbiter) any pre-hearing motions which the parties may file.

In any arbitration conducted pursuant to this Agreement, the arbiter shall not discuss the arbitration or any part thereof with any of the parties <u>ex parte</u>.

In any arbitration conducted pursuant to this Agreement, the arbiter shall express its decision in a writing to be delivered to each of the parties, which writing shall explain the reasons for the decision. The arbiter shall render a written decision on, and deliver the decision to the parties within fifteen (15) days after the close of the presentation of evidence by all the parties to the arbitration.

Each of the parties to any arbitration hereunder shall bear its own costs and fees and an equal share of the fees and expenses of the arbiter and administrative fees of the arbitration.

Any party to an arbitration conducted hereunder may seek reconsideration of all or any part of the original decision of the arbiter by filing a written request therefor with the arbiter and serving the same on all the other parties to the arbitration within ten (10) days after the date of delivery of the arbiter's original decision. Any other party may file a written response to such

request within ten (10) days after receipt of the request for reconsideration. No further replies will be permitted. The arbiter shall consider any such request either based solely on the documents filed or after a hearing, as he or she may determine in his or her sole discretion. The arbiter shall issue a written decision on any such request within fifteen (15) days after the submission of any response hereunder or within twenty (20) days after the submission of the request for reconsideration if no response is submitted.

There shall be no right to appeal from the final decision of the arbiter in any arbitration conducted hereunder, except on those grounds set out in §§13-22-201 et seq. of the Colorado Revised Statutes. Judgment upon any decision rendered by the arbiter may be entered in any court having jurisdiction and enforced in the manner judgments of said court are normally enforced.

If the arbiter shall find that there is an uncured material default and the Partnership thereafter fails to cure such default within thirty (30) days after the arbiter's decision, then the District shall have the right to terminate the management responsibilities of the Partnership under this Agreement and retain separate management for the Leased Premises; provided, however, that the District shall not be allowed to so terminate the management responsibilities of the Partnership as to a material default that cannot reasonably be cured within thirty (30) days, so long as the Partnership has promptly commenced such cure, diligently proceeds in a reasonable manner to complete the same thereafter, and effectuates such cure within one (1) year after the arbiter's decision. If the arbiter does not find an uncured material default, then there shall be no right to terminate the management responsibilities of the Partnership based upon the claimed default; however, such determination by the arbiter shall

not affect the District's other rights or remedies under this Agreement.

If the date upon which any such termination would be effective falls within any Major League Baseball Season during the Term or Grace Term, the effective date of such termination shall be the date of the Team's final home game of such Major League Baseball Season. Any such termination of management responsibilities shall have no effect upon the Partnership's tenancy, rent and other rights and obligations hereunder, including, but not limited to, the Fartnership's obligation to pay all costs of Maintenance, Repairs, Improvements, Alterations, and operations of the Leased Premises and Partnership Property (including the District's costs to hire and maintain a substitute manager), throughout the Term and Grace Term, which shall all remain in full force and effect.

8.9 <u>Inspection by the District</u>

District's Right to Inspect. acknowledge that the District will have an ongoing obligation to maintain the integrity of the design of the Stadium and ensure that the Leased Premises and Partnership Property are properly Maintained and Repaired throughout the Term and Grace Term. will be necessary for the District to periodically inspect the Leased Premises and Partnership Property to be certain that they are being maintained in a first-class manner. In upholding this obligation, the District shall have the right to inspect the Leased Premises and Partnership Property, at least annually, at times agreed to by the parties. If any such time cannot be agreed to by the parties, the District shall be entitled to provide the Partnership with written notice of such inspection date. No later than 15 days after receipt of such notice, the Partnership shall be required to allow the District to proceed with its inspection of

the Leased Premises and Partnership Property, and cooperate with and assist the District during the course of the inspection.

Scope of Inspection. The District shall have the to inspect the entire Leased Premises and Partnership > Property and shall comply with all security provisions regarding access to the Leased Premises and Partnership Property in conducting such inspections. The Partnership shall, as part of such security provisions, provide for District access to conduct inspections and shall cooperate fully with the District in its conducting such inspections. In the course of its inspections, the District may hire mechanical, structural or electrical engineers, or other experts to assist in making a determination that the Leased Premises and Partnership Property have been Repaired and maintained as a first-class facility. The reasonable costs of any inspection, including the cost of any engineers or experts retained by the District, shall be borne by the Partnership. shall submit an invoice to the Partnership of all costs incurred in any inspection within thirty (30) days after that inspection. Partnership shall be required to reimburse the District for all such expenses within thirty (30) days of receipt of that invoice. In the alternative, the Partnership may retain in its name, any engineers or experts designated by the District to assist in the inspection and remit payments directly to such engineers and experts as agreed between the Partnership and those hired.

8.10 Repairs Procedure.

(a) <u>Repairs Ownership</u>. Any Repair, resulting, upon completion of the Repair, in the addition to the Leased Premises of any component which shall be deemed to have, in the District's sole discretion, an economic life longer than the years, or fractions thereof, then remaining until the expiration of this Agreement and paid for by the Partnership or from the Repairs Fund shall be

Parthership Property and shall be subject to the purchase, request for removal, and abandonment provisions set forth in Section 2.4. Any Repair resulting, upon completion of the Repair, in the addition to the Leased Premises of any component which shall be deemed to have, in the District's sole discretion, an economic life shorter than the years, or fractions thereof, then remaining until the Expiration Date, shall, upon the District's election, become a part of the real property constituting the Leased Premises and title thereto shall be held by the District or its successors in interest. No Repairs shall be performed which would have a Material Adverse Effect on the District.

- (b) Conditions of Repair Work. On or before November 15, 1995 and on the 15th day of November of each succeeding year during the Term and Grace Term, the Partnership shall provide the District with a list of anticipated Repairs, with estimated budgets and time tables for completing such Repairs. Except as provided in Section 8.13, the Partnership will not initiate any listed Repairs or any other Repairs the cost of which the Partnership in good faith estimates will exceed \$100,000 without obtaining written approval from the District, which approval shall not be unreasonably withheld or delayed. Except as provided in Section 8.13, as to all Repair work, the Partnership shall ensure the fulfillment of the following conditions:
- (1) <u>District Liability</u>. Assurance to the District's reasonable satisfaction that the performance of any Repair work will not have a Material Adverse Effect on the District, result in the necessity for the District to remit any payment, or result in any liability to the District.
- (2) <u>Approval of Contractors. Subcontractors, and Materialmen</u>. The Partnership shall provide to the District, annually on or before November 1 of each year during the Term and

Grace Term, a list of contractors to be employed in connection with Repairs to the Leased Premises and Partnership Property for District approval, which approval shall not be unreasonably withheld. Failure of the District to disapprove of such list, in writing, within thirty (30) days after receipt by the District shall be deemed an approval of the list of contractors. The Partnership shall use reasonable good faith efforts to not employ any contractors who discriminate in violation of applicable federal, state and local laws.

22.0

ينخ

- Budgets. The District shall be provided or have the opportunity to review and inspect: (i) a complete set of plans and specifications setting forth all construction requirements for the Repair; (ii) copies of all permits and requisite approvals of any governmental bodies necessary for the construction and use of the Repair; (iii) an executed copy, if any, of the architecture contract and an executed copy, if any, of the construction contract; and (iv) detailed budget and cash flow projections of the Repair costs and a schedule of the estimated amount and time of disbursement of each advance, if any.
- (4) <u>Insurance</u>. The Partnership shall have delivered to the District insurance policies or evidence thereof providing for the applicable coverages and policy terms set forth in Article VI.
- (c) Advances or Reimbursements from the Repairs Fund. Subject to the provisions of Section 8.10 (a), 8.10 (b), and 8.11, the Partnership shall be entitled to use funds from the Repairs Fund for the payment of Repair work on the Leased Premises and Partnership Property. The Partnership may, in its discretion, elect to perform any Repair and seek reimbursement from the Repairs Fund after completion or seek advances from the Repairs Fund to pay

for the Repair construction as the work progresses. Partnership recognizes that the purpose of the Repairs Fund is to ensure the availability of adequate funds to perform Repair Work throughout the Term and Grace Term. The Partnership shall exercise prudence in its use of the Repair Fund. The District may withhold its approval of advances or reimbursements from the Repairs Fund if the Fartnership does not comply with the provisions of Section Nothing in this Section 8.10(c) or elsewhere in this 8.10(b). Agreement shall be interpreted as a limitation upon Partnership's obligation to perform all Repair work, at its sole expense, throughout the Term and Grace Term. The Partnership shall not submit a request for advance from the Repairs Fund or part thereof prior to completion of any Repair or portion thereof for which advance or reimbursement is requested. By submission of the request, the Partnership warrants that to the best of knowledge, the Leased Premises or Partnership Property, is free and clear of liens, claims, security interests or encumbrances (except such liens, claims, security interests or encumbrances previously approved by the District or permitted pursuant to the provisions of Section 2.4) in favor of the Partnership, subcontractors, material suppliers or other entities making a claim by reason of having provided labor, materials or equipment relating to the Repair and that the requirements of any governmental authority with jurisdiction over the Leased Premises, Partnership Property or both have been met. The District may refuse to consent to payments from the Repairs Fund to such extent as may be necessary to protect the District from loss because of:

- (i) Defective performance in effectuating the Repair or any part thereof;
- (ii) Third-party claims made or filed or reasonable evidence indicating the probability of such filing; or

(iii) Failure of the Partnership to make payments properly to subcontractors for labor, materials or equipment. If the District in its sole discretion determines that any work or materials do not conform to any approved plans and specifications or sound building practices, or otherwise depart from any of the requirements of this Agreement, the District may require the work to be stopped and withhold disbursements until the matter is corrected. In such event, the Partnership will promptly correct the work to the District's reasonable satisfaction.

Upon completion of any Repair, the Partnership shall provide or make available to the District a complete set of record documents regarding such Repair.

8.11 Repairs Fund.

The Partnership will pay the District, on January (a) 15, 1996, \$550,000.00. Thereafter, the Partnership shall make annual payments to the District on January 15 of each succeeding year throughout the Term and Grace Term equal to the previous year's payment plus an increase thereto equal to the lesser of 4% or the percentage difference between the average F.W. Dodge Construction Costs Index for Denver (computed by tallying all index publications during the course of any given year and dividing that sum by the number of publications made during the year) for the calendar year prior to each annual payment and the average index for the year two calendar years prior to such annual payment (i.e., if the average index for 1995 equals 100 and the average index for 1996 equals 103, the January 15, 1997 payment would increase by 3% from the January 15, 1996 payment; 3% being the lesser of 4% or the percentage difference (3/100) between the 1996 and 1995 average The amounts provided for in this Section 8.11(a) will indexes). fund the "Repairs Fund". The Repairs Fund shall be the property of the Partnership, but shall be held by the District, subject to the

terms of this Agreement. All interest earned on the Repairs Fund shall remain in the Repairs Fund and become a part thereof. The Repairs Fund shall be transferred and assigned to the Partnership upon the expiration of this Agreement, so long as no Partnership Default then exists and subject to the provisions of Section 8.12.

If the F.W. Dodge Construction Costs Index shall become unavailable to the public because publication is discontinued, or otherwise, then a comparable index published by the United States government, any other governmental agency, or a recognized construction publication shall be used, as agreed to by the District and the Partnership.

The amounts required to be paid under this Section 8.11(a) correspond to the anticipated costs of repairs to the Leased Premises and Partnership Property during the Term, and are based upon an examination of repair histories for similar facilities. The Repairs Fund will be maintained as a separate escrow account to be used solely for Repairs to the Leased Premises and Partnership Property and will not inure to the benefit of the District.

Notwithstanding, if a Partnership Default occurs during the Term or Grace Term and is not cured upon the expiration or termination of this Agreement. the Repairs Fund shall not be transferred and assigned to the Partnership until the Partnership default is cured or otherwise resolved, and may be used by the District, in its discretion, to pay Partnership obligations which have accrued and are owing to the District as of such date.

(b) The Repairs Fund may be drawn only upon the signature of the designated signatory or signatories of the District and the Partnership. Disbursement of funds from the Repairs Fund shall be made only with the District's approval, which shall not be unreasonably withheld. Notwithstanding anything in

: .

this Agreement to the contrary, the District shall have no liability for any Repair, replacement or restoration, which shall be paid for with funds in the Repairs Fund. The District shall have no obligation to fund any Repair, replacement or restoration to the Leased Premises or Partnership Property. The Partnership remains solely liable for the funding of all Repairs from all sources available to the Partnership.

- obligation to fund all Repairs that relate to the Partnership's use of the Leased Premises and Partnership Property and full payment of all amounts accrued and owing to the District upon the termination or expiration of this Agreement, the Partnership hereby grants to the District a security interest in the Repairs Fund necessary to fund all such Repairs and satisfy all such accrued payment obligations. The security interest shall be a first lien. The Partnership shall not grant or permit any other security interest in or lien on the Repairs Fund or any sums contained therein or interest earned thereon.
- (d) The Partnership shall reimburse to the Repairs Fund any amounts taken therefrom for any Repair or repair of Damaged Property to the extent that the Partnership receives funds from any source (including, but not limited to, insurance proceeds and recovery from third parties) to reimburse it for costs and expenses incurred in the performance of such Repair of repair of Damaged Property.
- 8.12 Use and Return of Repairs Fund. At the end of the Term or Grace Term, or upon termination of this Agreement, all sums in the Repairs Fund shall continue to be held in a separate escrow account, exclusively for the purpose of funding Repairs that relate to the Partnership's use of the Leased Premises and Partnership Property (except for Partnership Property removed by the

Partnership) hereunder, until such Repairs have been completed, and any remaining funds shall be returned to the Partnership. Nothing contained in this Section 8.12 shall be interpreted to limit the Partnership's obligation to bear the costs of all Repairs that relate to the Partnership's use of the Leased Premises and Partnership Property.

- 8.13 Emergency Repairs. If the Partnership reasonably determines that the health or safety of persons or property will be jeopardized absent the immediate commencement of a Repair, the Partnership may commence such Repair without the prior written approval of the District. The Partnership shall provide the District, monthly during the Term and Grace Term, a report of all such Repairs made by the Partnership.
- 8.14 Competitive Process. The Partnership shall ensure that contracts for the performance of any Repair work are fair and competitive. Prior to being entitled to any advancements or reimbursements from the Repairs Fund, the Partnership must provide evidence, satisfactory to the District, of such competitiveness. The Partnership shall provide the District with an annual report regarding the participation of minority and women owned businesses in the maintenance and repair of the Leased Premises. This report shall be provided to the District on November 1 of the first year after the Commencement Date and annually thereafter, on November 1 of each succeeding year during the Term and Grace Term.
- 8.15 <u>Limitation of Responsibility</u>. The making of any advance by the District shall not constitute or be interpreted as either (a) an approval or acceptance by the District of the work done through the date of the advance, or (b) a representation or indemnity by the District to any party against any deficiency or defect in the work or against any breach of any contract. Inspections and approvals of the plans and specifications, the

workmanship and materials used in any Repair, and the exercise of any other right of inspection, approval, or inquiry granted to the District in this Agreement are acknowledged to be solely for the protection of the District's interest, and under no circumstances shall they be construed to impose any responsibility or liability of any nature whatsoever on the District to any party. No disbursement or approval by the District shall constitute a representation by the District as to the nature of any Repair, its construction or its intended use for the Partnership or for any other person, nor shall it constitute an indemnity by the District to the Partnership or to any other person against any deficiency or defects in any Repair or against any breach of contract.

- 8.16 Affirmative Covenants. The Partnership covenants and agrees with the District that while this Agreement is in effect, the Partnership will:
- (a) Project Claims and Litigation. Subject to the Partnership's actual knowledge, promptly inform the District of (1) all material adverse changes in the financial condition of any general contractor; (2) any litigation and claims, actual or threatened, affecting any Repair, Improvement or any other repair, replacement or restoration affecting the Leased Premises, Partnership Property, or the general contractor therefor, which could materially affect the successful completion of any Repair, Improvement or any other repair, replacement or restoration affecting the Leased Premises or Partnership Property, or the ability of the general contractor therefor to complete the Repair, Improvement or other repair, replacement or restoration, as agreed.
- (b) <u>Payment of Claims and Removal of Liens</u>. (1) Cause all claims against the Partnership, the Leased Premises or the District for labor done and material and services furnished in connection with any Repair, Improvement or any other repair,

replacement or restoration affecting the Leased Premises or Partnership Property, including but not limited to any claims made against the District pursuant to § 31-26-107 of the Colorado Revised Statutes, to be fully paid and discharged in a timely manner, (2) diligently file or procure the filing of a valid notice of completion of any Repair, Improvement or any other repair, replacement or restoration affecting the Leased Premises or such comparable document as may be permitted under applicable lien laws, (3) diligently file or procure the filing of a notice of cessation, or such comparable document as may be permitted under applicable lien laws, upon the happening of cessation of labor on any Repair, Improvement or any other repair, replacement or restoration affecting the Leased Premises or Partnership Property for a continuous period of thirty (30) days or more, and (4) take all reasonable steps necessary to remove all claims of liens against the Leased Premises or Partnership Property or any rights or interests appurtenant to the Leased Premises or Partnership Property. Upon the District's request, the Partnership shall make proper demands claims or upon or against materialmen, subcontractors, or other persons who have furnished or claim to have furnished labor, services, or materials in connection with any Repair, Improvement or any other repair, replacement or restoration affecting the Leased Premises or Partnership Property. The Partnership shall, within ten (10) days after the filing of any claim of lien that is disputed or contested by the Partnership, or upon the filing of any claim pursuant to § 38-26-107 of the Colorado Revised Statutes, provide the District with a surety bond issued by a surety acceptable to the District sufficient to release the claim of lien or to pay any amount claimed beyond any bonds obtained by the Partnership pursuant to §§ 38-26-105 and 38-26-106 of the Colorado Revised Statutes as required by Section 6.10 of

this Agreement, deposit with the District an amount satisfactory to the District for the possibility that the contest will be unsuccessful, or provide other security reasonably acceptable to the District. If the Partnership fails to remove any lien or pay any claim on the Leased Premises or Partnership Property or provide a bond or deposit pursuant to this provision, the District may pay such lien or claim, or may contest the validity of the lien or claim, and the Partnership shall pay all costs and expenses of such contest, including the District's reasonable attorneys' fees.

ARTICLE IX

UNTENANTABILITY

- 9.1 Alternate Site. If the Stadium is rendered untenantable in whole or in any material part (as defined in Section 16.1), then for the period of such untenantability, the Partnership shall be entitled to make arrangements for an alternate site for its Major League Baseball Games during any such Major League Baseball Season throughout the Term and Grace Term. Notwithstanding any provisions herein to the contrary, any such period of untenantability shall not give the Partnership any right to terminate this Agreement, unless the Stadium is rendered untenantable in whole or in any material part within the last five (5) years of the Term or Grace Term and a new lease agreement or long term extension (at least five (5) years of this Agreement has not been agreed to between the parties to this Agreement or their successors in interest.
- 9.2 Continuing Obligations. Any period of untenantability shall not relieve the Partnership of any of its necessary Stadium Management obligations hereunder; however, the Partnership shall not be obligated to make any payments to the District pursuant to Section 4.2(b) and (c) for any such period of untenantability.
- 9.3 <u>Rebuilding Obligations</u>. If it becomes necessary for the District to rebuild the Stadium, the District will only be required

to complete such rebuilding to the extent it receives insurance proceeds sufficient to effectuate the same. If such insurance proceeds are not sufficient to cover the entire cost of rebuilding, any rebuilding shall be as agreed to between the District and the Partnership.

The District shall receive and hold, in its name, all insurance proceeds payable due to the destruction of the Stadium from forces outside of the District's control. If the Stadium is not rebuilt after destruction, the District shall be entitled to disburse all insurance proceeds received as required by the Act.

The District shall have no obligation to rebuild any portion of the Leased Premises during the last five years of the Term or Grace Term unless a new lease agreement or long-term (at least ten (10) years) extension of this Agreement has been agreed to between the parties to this Agreement or their successors in interest.

ARTICLE X

RIGHT OF ENTRY

Notwithstanding the provisions of Section 8.9 the District and its agents and representatives, with notice to the Partnership, during the Term and Grace Term, shall have rights of access and entry into and upon the Leased Premises and Partnership Property, provided any such entry does not unreasonably interfere with the Partnership's operations. The District shall comply with all security provisions regarding access to the Leased Premises and Partnership Property. The Partnership shall, as part of such security provisions, provide for District access and shall cooperate fully with the District in its entry into and upon the Leased Premises and Partnership Property.

ARTICLE XI

DEFAULT AND REMEDIES

- 11.1 <u>Default by the Partnership</u>. The occurrence of any one or more of the following events constitutes a default by the Partnership under this Agreement ("Partnership Default"):
- (a) Failure by the Partnership at any time to pay any sums payable by the Partnership to the District hereunder within ten (10) days after notice from the District that any such payment is past due if not paid when due;
- (b) Failure by the Partnership to obtain any advance approval of or give advance notice to the District required by this Agreement prior to any material action or omission by the Partnership;
- (c) Failure by the Partnership to observe or perform any other covenant, agreement, condition or provision of this Agreement, if such failure shall continue for more than thirty (30) days after notice of such failure is given to the Partnership by the District; provided, however, that the Partnership shall not be in default with respect to matters that cannot be reasonably cured within thirty (30) days, so long as the Partnership has promptly commenced such cure, diligently proceeds in a reasonable manner to complete the same thereafter, and effectuates such cure within one (1) year after commencing such cure;
- (d) the Partnership admits in writing its inability to pay its debts as they mature, or makes an assignment for the benefit of creditors, or applies for or consents to the appointment of a trustee or receiver for the Partnership or for a major part of its property;
- (e) a trustee or receiver is appointed for the Partnership or for a major part of its property and is not discharged within ninety (90) days after such appointment;

- bankruptcy, reorganization, receivership, arrangement, insolvency or liquidation proceedings, or other proceedings for relief under any federal or state bankruptcy law, or similar law for the relief of debtors, are instituted by or against the Partnership, and, if instituted against Partnership, are allowed against it or are consented to by it or are not dismissed within ninety (90) days after such institution; or
- (g) any action of the Partnership results in a Material Adverse Effect upon the District or could so result with the passage of time.
- 11.2 <u>District's Remedies</u>. If a Partnership Default occurs, in addition to any other rights or remedies the District may have at law or in equity, the District shall have the following rights:
- (a) The District may enforce the provisions of this Agreement and may enforce and protect the rights of the District hereunder by a suit or suits in equity or at law for the specific performance of any covenant or agreement contained herein, or for the enforcement of any other appropriate legal or equitable remedy, including recovery of monetary damages (including consequential damages) and all moneys due or accrued or accruing and not yet paid from the Partnership under any of the provisions of this Agreement, or any other relief or remedies to the extent permitted by law, by filing a cause of action or actions for such damages, equitable relief, of other appropriate remedies or relief from the Partnership in any court of competent jurisdiction.
- (b) After the time when the District has given notice and any applicable cure period provided has expired, if any sums payable by the Partnership shall remain due and payable, or after the time for performance by the Partnership of any other term, covenant, provision or condition of this Agreement, or before the

expiration of that time in the event of a bona fide emergency (in which case the District shall only be required to give notice as is reasonable and practical under the circumstances), the District may, at the District's election (but without obligation), make any payment required of the Partnership under this Agreement, or perform or comply with any covenant or condition imposed on the Partnership under this Agreement, as the District deems advisable. The amount so paid plus the cost of such performance or compliance, plus interest on such sums at the rate of four percent (4%) above Colorado National Bank's (or its successor bank(s)) prime lending rate, shall be deemed to be additional rent payable by the Partnership immediately upon demand. No such payment, performance or compliance by the District shall constitute a waiver of default or of any remedy for default. The District may, with notice to the Partnership and at any time or from time to time, charge, set off, and otherwise apply all or any part of any costs incurred by the District in connection with its performance of the Partnership's obligations against any of the District's obligations now or in the future.

Stadium Management functions and after a determination that such default is a material default pursuant to the provisions of Section 8.8, the District may terminate the Partnership's management authority and responsibilities under this Agreement. Termination of the Partnership's management authorities and responsibilities pursuant to this section shall not relieve the Partnership of any of its other obligations under this Agreement, including but not limited to its obligation to occupy the Leased Premises for the Term, to continue to make the payments as set forth in Section 4.2, to bear the costs of all Repairs, Improvements, Maintenance, and operations of the Leased Premises and Partnership Property,

including, but not limited to the cost to employ a substitute operator/manager and to fund the Repairs Fund pursuant to Section 8.11(a). In addition, the District shall be entitled to recover, in addition to all other damages to which it may be entitled pursuant to this Article XI, damages equal to the District's costs and expenses incurred in obtaining, and retaining until the expiration or termination of this Agreement, replacement management services for the Leased Premises and Partnership Property.

- (d) The District may retake possession of the Leased Premises without terminating this Agreement.
- (e) No termination of any or all of the Partnership's rights under this Agreement shall deprive the District of any of its remedies or actions against the Partnership for past or future rent or other sums due from the Partnership hereunder, nor shall the bringing of any action for rent or other sums or other Partnership Default be construed as a waiver of the right to obtain possession of the Leased Premises.
- (f) The District may exercise any of its remedies hereunder with or without exercise of its right to terminate hereunder. No action on the part of the District shall be deemed to be a termination of this Agreement unless expressly stated by the District to be intended to be a termination.
- (g) If any conduct of the Partnership in violation of this Agreement results in a default by the District of any of its obligations under the Bonds, the District may, in addition to any other remedies it may have hereunder, recover from the Partnership any consequential damages suffered by the District as a result of such default, including but not limited to all sums which the District shall be held liable to pay to any city, state or federal taxing authority, or any holders of the Bonds. Notwithstanding the foregoing, payments expressly required to be made by the

Partnership under this Agreement or expressly agreed to in writing by the District shall not create any Partnership liability or right of recovery by the District hereunder.

- (h) The District shall not be required to post any bond in connection with any action or proceeding filed by the District to enforce Article XIII of this Agreement.
- 11.3 <u>Default by District</u>. The occurrence of any one or more of the following events constitutes a default by the District under this Agreement ("District Default"):
- (a) Failure by the District at any time to pay any sums payable by the District to or on behalf of the Partnership hereunder within ten (10) days after notice from the Partnership that any such payment is past due if not paid when due (no advance or reimbursement to be paid to the Partnership pursuant to Article VIII shall be considered due until such time as all conditions precedent to the Partnership's (or any third party's) entitlement to such advance or reimbursement have been satisfied);
- (b) Failure of the District to observe or perform any other covenant, agreement, condition or provision of this Agreement, if such failure shall continue for more than thirty (30) days after notice of such failure is given to the District by the Partnership; provided, however, that the District shall not be in default with respect to matters that cannot be reasonably cured within thirty (30) days, so long as the District has promptly commenced such cure, diligently proceeds in a reasonable manner to complete the same thereafter, and effectuates such cure within one (1) year after commencing such cure.
- (c) A trustee or receiver is appointed for the District or for a major part of its property and is not discharged within ninety (90) days after such appointment or;

(d) Bankruptcy, reorganization, receivership, arrangement, insolvency or liquidation proceedings, or other proceedings for relief under any federal or state bankruptcy law, or similar law for the relief of debtors, are instituted by or against the District, and, if instituted against the District, are allowed against it or are consented to by it or are not dismissed within ninety (90) days after such institution.

11.4 Partnership's Remedies.

- (a) If a District Default occurs, the Partnership's sole remedy hereunder shall be a suit or suits at law for recovery of monetary damages (including consequential damages) and all moneys due or to become due from the District under any of the provisions of this Agreement. The Partnership may enforce and protect such rights by filing a cause of action or actions for such damages in any court of competent jurisdiction within the boundaries of the District. The Partnership shall have no right to terminate this Agreement and is entitled to no equitable remedies as a result of any District Default; provided, however, nothing herein shall preclude the Partnership from seeking injunctive relief against the District.
- (b) In the event the District breaches the provisions of Section 3.2 by failing to have the Leased Premises available for use by the Partnership on or before April 6, 1995, and the Partnership is required to play and actually plays its Major League Baseball Games at another facility, the Partnership shall be entitled, as its exclusive remedy, to liquidated damages of \$50,000.60 per day from April 6, 1995 until the Leased Premises are made available for use by the Partnership. The Partnership shall not, by deferral of the Commencement Date by the Partnership pursuant to Section 3.2 or otherwise, be entitled to liquidated

damages subsequent to the date the Leased Premises are made available for use by the Partnership.

11.5 General Provisions.

- (a) No right or remedy herein conferred upon, or reserved to the District or the Partnership is intended to be exclusive of any other right or remedy, but each shall be cumulative and in addition to every other right or remedy given herein or now or hereafter existing at law, or in equity or by statute.
- (b) waiver by either party of No any breach of obligations, agreements, or covenants herein shall be a waiver of any subsequent breach of any obligation, agreement or covenant, nor shall any forbearance by either party to seek a remedy for any breach by the other party be a waiver by such party of any rights or remedies with respect to such or any subsequent breach, nor shall any express waiver by either party be deemed to apply to any other existing or subsequent right to remedy any default by the other party, nor shall any waiver by either party of any default or breach by the other party in the performance of any of the covenants or obligations of such other party under this Agreement be deemed to have been made by the party against which the waiver is sought to be charged unless contained in a writing executed by such party.
- (c) In the event that either party fails to pay any payment required hereunder when due, then, without limiting any other rights of the non-defaulting party, the defaulting party shall be liable for interest thereon at the rate of four percent (4%) above Colorado National Bank's (or its successor bank(s)) prime lending rate from the date that such installment was due until the date paid in full.

ARTICLE XII

SURRENDER OF LEASED PREMISES

- General. Upon the expiration or termination of this Agreement (by lapse of time or otherwise) the Partnership shall peaceably deliver up and surrender the Leased Premises to the District in good order, Maintenance, and Repair. The Partnership shall surrender to the District all keys for the Leased Premises at the place then fixed for the receipt of notice by the District, and shall notify the District in writing of all combinations of locks, safes and vaults, if any. The Partnership's obligations to observe and perform the covenants set forth in this Article XII shall survive the expiration or earlier termination of this Agreement (by lapse of time or otherwise); provided however, that if the Partnership has a continuing obligation, upon expiration or termination of this Agreement, to perform or complete any Repair(s), alterations or Improvement(s) and the Repairs Fund contains, in the District's sole discretion, adequate funds to complete such Repair(s), alterations or Improvement(s), Partnership may surrender the Leased Premises, waive claim to all or that portion of the Repairs Fund required to cover the cost of such Repairs, alterations or Improvements pursuant to Section 8.11 or elsewhere in this Agreement, and be relieved of all liability for the performance or completion of such Repair or Improvement.
- 12.2 Partnership's Property. At the expiration or termination of this Agreement (by lapse of time or otherwise), all structures and fixtures incorporated into the Leased Premises shall be deemed a part of the Leased Premises and the same shall not be removed, except that upon the expiration or termination of this Agreement, Partnership Property, or any part thereof, shall be incorporated into the Leased Premises only if such Partnership Property is purchased by the District at the then fair market

value, as determined pursuant to Section 2.4, for such Partnership Property. - The District may require the Partnership to remove the Partnership Property, or any part thereof, which the District has not purchased and the Partnership has not voluntarily removed upon the expiration or termination of this Agreement, from the Stadium Land. Subject to the provisions of Section 2.4, if the Partnership Property, or any portion thereof designated by the District, is not removed from the Stadium Land within a reasonable time after the District's written request to so remove, not to exceed ninety (90) days, the Partnership Property, or designated portion thereof, shall be deemed abandoned, and shall, at the District's election, become part of the real property constituting the Leased Premises, with title thereto held by the District or its successors in interest. Thereafter the District may, at its option, (a) cause that property to be removed at the Partnership's expense, (b) sell all or any part of such property at public or private sale, without notice to the Partnership; or (c) declare that title to such property shall be deemed to have passed to the District.

12.3 Abandoning Leased Premises. The Partnership shall not vacate or abandon the Leased Premises at any time during the Term or Grace Term. If the Partnership does so vacate or abandon the Leased Premises or is dispossessed by process of law, any Partnership Property which may be left on the Leased Premises following such abandonment or dispossession shall be deemed to have been abandoned by the Partnership, and in that event such property shall be disposed of by the District in accordance with the provisions of Section 12.2. The District agrees that the Partnership shall not be deemed to have abandoned or vacated the Leased Premises during any Off-Season, so long as the Partnership's Stadium Management obligations are being performed by the Partnership or a replacement manager for the Leased Premises if the

Partnership has been terminated as manager pursuant to the provisions of Section 8.8.

ARTICLE XIII

RELOCATION OF TEAM

The Partnership acknowledges that the District will be irreparably harmed by the relocation of the Team to a location other than the Leased Premises during the Term of this Agreement. Accordingly, the Partnership agrees as follows:

- (a) The District does not have an adequate remedy at law for breach of this Article XIII.
- (b) Except during any period of untenantability pursuant to Article IX, temporary taking pursuant to Section 16.3 or deferral of the Commencement Date pursuant to Section 3.2, the Partnership shall not apply to the National League for approval to allow the Team to play any Major League Baseball Game anywhere other than in the Stadium during the Term.
- (c) The Partnership recognizes that the Stadium is being constructed, the sales tax was imposed, and the bonds were issued solely to bring the Team and the franchise to the District, and agrees that in the event of a violation of this Article XIII, the District shall, without posting any bond, be entitled to seek and obtain an injunction from the District Court of the City and County of Denver, Colorado, or any other court of competent jurisdiction, to enjoin any violation of this Article XIII. Additionally, the District shall have the right to enforce any breach of Sections 5.1(3) or 24.6 through specific performance as provided in Section 11.2(a), or otherwise as provided in this Agreement.

ARTICLE XIV

SALE OF FRANCHISE

Upon sale of the Franchise or 80% of the beneficial interests in the Partnership during the Term, or Grace Term, the Partnership

shall pay to the District an amount equal to 2% of the net profit realized by the Partnership or the selling partners or beneficial owners of the Partnership, as the case may be, not to exceed \$2 million. Net profit shall be the gross proceeds of the sale less capital contributions (or capital contributions of partners selling their beneficial interests) to the Partnership, plus a five percent (5%) imputed annual return on such capital contribution(s), and less Partnership debt if such debt is not assumed or paid by the purchasing entity. The sale of the Franchise or 80% beneficial interests in the Partnership to any person or entity (or any affiliate thereof) who is a partner or a stockholder of a partner on the date of this Agreement and who has been a partner or a stockholder of a partner in the Partnership for at least three (3) years prior to the subject sale will not trigger this profit sharing provision. Individual sales of the Partnership interests in the Partnership and sales of stock of the general partner of the Partnership will not trigger this profit-sharing provision, provided such sales do not result in the sale of 80% or more of the beneficial interests in the Partnership to a person or entity or related persons or entities that have not been a partner or partners in the Partnership for at least three (3) years prior to the imitial sale, subsequent to the Commencement Date, of any beneficial interest to such person(s) or entity(ies). Nothing in this Article XIV shall be interpreted as limiting the Partnership's obligations or the District's rights set forth in Article XIII.

ARTICLE XV ASSIGNMENT

The Partnership shall not assign this Agreement, except with the prior written consent of the District which consent may be withheld at the sole discretion of the District. Notwithstanding the foregoing, it is agreed that the District will consent in writing to an assignment of this Agreement to any person, firm, corporation or entity which acquires the Franchise now held by the Partnership in accordance with applicable Major League and National League rules and regulations and subject to the prior compliance with all the provisions of this Agreement, including the provisions of Article XIII; provided such assignee assumes unconditionally in writing in advance all of the Partnership's past, present and future obligations hereunder and agrees to be bound hereby. Upon the effective date of any such assignment, provided the District has received from the Assignee the written assumption of obligations referred to in the previous sentence in form and substance satisfactory to the District, the Partnership shall have no further liability hereunder.

ARTICLE XVI EMINENT DOMAIN

Termination for Condemnation. In the event that any material part of the Leased Premises shall at any time during the Term be taken by exercise of the power of eminent domain ("Condemnation"), whether by formal condemnation proceedings or by purchase under threat of exercise of the power of eminent domain proceedings, this Agreement shall terminate on the date on which possession is required to be delivered to the condemning authority. As used herein, a "material part" shall mean any loss of Stadium seating in excess of a number of seats having a face ticket price equal to 10% or more of the aggregate face ticket price of all seats in the Stadium, loss of any material portion of the concourse area, or the failure of the District, within twelve (12) months after any Condemnation of parking facilities on the Leased Premises, to have on the Leased Premises or adjacent land at least seventy-five percent (75%) of the parking spaces that were available to the Partnership immediately prior to the Condemnation,

including spaces acquired with condemnation proceeds as provided in Section 16.5.

Notwithstanding, the Partnership may elect in its sole discretion to treat any loss as not a "material part" of the Leased Premises, and this Agreement will not terminate upon such Condemnation. The Partnership must make any such election within fifteen (15) days after an award of immediate possession of the condemned portion of the Leased Premises to the condemning authority and must provide written notice of such election to the District within that same fifteen (15) days. If this Agreement terminates pursuant to the provisions of this Section 16.1, all rights, obligations and liabilities of the parties hereto shall end as of the effective date of such termination, without prejudice to any rights which have accrued prior to such termination.

- 16.2 Performance of Work. If there shall be a Condemnation and this Agreement shall not terminate as a result thereof in accordance with the provisions of Section 16.1, the District shall endeavor to perform any and all work necessary to restore the Stadium to a complete architectural unit suitable for the Partnership's use in as expeditious a manner as possible, with the costs of such restoration to be borne as agreed to by the District and the Partnership, provided, however that the District shall have no obligation to perform such restoration unless and until an agreement regarding costs is reached. All proceeds of any condemnation award or settlement shall be used to pay for costs of restoration, including, but not limited to, the purchase of any necessary additional real property.
- 16.3 Temporary Taking. In the event of any temporary taking of the Leased Premises or any portion thereof for public use, this Agreement shall not terminate by reason thereof, and the rights and obligations of the parties shall continue in full force and effect

as provided herein except that during any period of a temporary taking of the Stadium or a material part thereof (as defined in Section 16.1), the Partnership shall be entitled to make arrangements for an alternative site for its Major League Baseball Game playing dates, and the Partnership shall be relieved of its Stadium management responsibilities during the period of such temporary taking to the extent such management responsibilities are inconsistent with the terms of the taking, and shall not be liable for payment of rent as set forth in Sections 4.2(b) - (f) during the time of such temporary taking.

The Partnership shall be entitled to make a separate claim against the condemning body for an award of any damages sustained by it as a result of such temporary taking.

- acknowledges the District's agreement with the City and County of Denver regarding future construction of Broadway across the parking facilities, which construction by the City and County will reduce the number of parking spaces on the Stadium Land. The District shall use reasonable efforts to replace such lost parking but shall have no obligation to pay any costs of replacement of such lost parking facilities in excess of the proceeds received from the City and County of Denver pursuant to the current agreement. Replacement parking facilities may be constructed as agreed by the District and the Partnership.
- 16.5 Condemnation of Parking Facilities. If more than five percent (5%) of the parking spaces located on the Leased Premises at the time of any Condemnation are taken by such Condemnation, the District shall use the proceeds received as a result of the Condemnation to construct on the remaining Leased Premises, or acquire on adjacent land, substantially similar parking facilities to the extent possible with such proceeds.

ARTICLE XVII

SALE OF STADIUM LAND AND PURCHASE OPTION

Section 34-14-129 of the Act provides that the District cannot sell the Stadium Land to any buyer other than a "qualified" buyer. The District shall accept input from the Partnership as to what criteria will define a "qualified" buyer during the Term or Grace Term. The Board, however, shall have the ultimate authority to determine, in its sole discretion, what criteria will define a "qualified" buyer.

If at any time during the Term or Grace Term the District proposes to sell the Leased Premises to a third-party offeree, the District will first give the Partnership a written offer to sell the Leased Premises to the Partnership on the same terms and conditions on which the District proposes to sell the Leased Premises to the third-party offeree. The Partnership must provide the District with a written acceptance or rejection of such offer within thirty (30) days after receipt. Failure to provide such acceptance or rejection within that time shall be deemed a rejection of the District's offer and the District shall thereupon be entitled to proceed with its sale to the third-party offeree upon the terms and conditions of the original offer. If at any time during the Term or Grace Term the Partnership shall desire to purchase the Leased Premises, the District will negotiate in good faith to sell the Leased Premises to the Partnership, subject to all rules, ordinances, regulations, laws or other agreements by which the District is or may be bound. If the Partnership accepts any contract for sale from the District, notwithstanding anything in this Agreement to the contrary, such Contract shall be specifically enforceable by each party against the other.

ARTICLE XVIII

STADIUM SECURITY AND CROWD CONTROL

72

- expense, at all Major League Baseball Games, baseball related events and Non-Baseball Events such security, traffic control, police, and crowd control personnel as shall be required for such purposes, except that such security, traffic control, police and crowd control for the District's Off-Season Non-Baseball Events shall be at the District's or other user's expense. The Partnership will permit the District to or will itself utilize the Partnership's security and crowd control administrative personnel, and the District shall be required to utilize such personnel for supervision of security at the District's Non-Baseball Events.
- 18.2 Twenty-Four (24)-Hour Security. The Partnership, at its expense, will provide such security guards and night watchmen as may reasonably be necessary in order to provide twenty-four (24) hour per day, year-round, protection and security of the Leased Premises and Partnership Property.

ARTICLE XIX

BONDS

- 19.1 Bond Constraints. Other than payments required to be made under this Agreement or expressly agreed to in writing by the District, the Partnership shall not make or agree to make any payments to, or for the benefit of, the District that would be considered used, directly or indirectly, to pay principal of, or interest on, the Bonds for purposes of the private security or payment test of section 141(b)(2) of the Internal Revenue Code. Such payments may include, but are not limited to:
- (1) Payments for the lease or purchase of property from the District or any related party of the District;

- (2) Payments for services provided by the District or any related party of the District, other than payments of generally applicable taxes, and
- (3) Payments to third parties that directly or indirectly benefit the District or any related party of the District.

For purposes of this Agreement, any payment that is prohibited by this section shall be deemed to have a Material Adverse Effect on the District, unless the District's prior written consent to such payment has been given.

- 19.2 <u>Defeasance</u>. The District intends to designate certain revenue received by the District, as the District determines in its sole discretion, to be deposited in a reserve account (the "Defeasance Account"), which account will be used to pay the District's debt service pursuant to the Bonds. The District will continue to deposit funds in the Defeasance Account until the account contains funds sufficient to satisfy all of the District's remaining debt service pursuant to the Bonds until fully defeased as provided in the documents relating to the Bonds (the "Defeasance Date").
- Repair, Improvement and Alteration Restrictions. Prior to the Defeasance Date, no Repair, improvement or alteration shall be incorporated into the Leased Premises by the Partnership or any of its employees, agents, representatives, or independent contractors, if such Repair, improvement or alteration would, upon incorporation into the Leased Premises, have a Material Adverse Effect on the District. Any such incorporation of a Repair, improvement or alteration into the Leased Premises by the Partnership without prior notification to and written consent by the District as provided in this Agreement or contrary to any direction from the District to not incorporate such Repair,

improvement or alteration shall subject the Partnership, in addition to any other remedies to which the District may be entitled hereunder, to damages as set forth in Section 11.2(g). The District may, in its discretion, require the Partnership to remove any such Repair, improvement, or alteration from the Leased Premises.

ARTICLE XX CONCESSIONS

- 20.1 Partnership Rights. Except as expressly provided in this Agreement, including Sections 4.2(d), (e) and (f), the Partnership shall have all rights to the concessions operated on the Leased Premises, including the right and duty to select concessionaires and to negotiate and enter into concession agreements for the Leased Premises for a term no longer than the Term.
- shall construct the portion of the concession areas as described in Exhibit D and the Partnership shall construct and furnish the portion of the concession areas as described in Exhibit E. The Partnership's obligations under this section shall be accomplished by one or more separate contracts among the Partnership, its contractors and its vendors, and the District will coordinate the performance of the Partnership's obligation under this section with the performance of its own obligations.
- 20.3 <u>Concessionaire Agreement</u>. The Partnership will obtain, or has obtained through a competitive process, a concession agreement that will provide: (a) the concessionaire will service all events at the Stadium, including Non-Baseball Events sponsored by the District, on the same terms (including concession commissions) as baseball related events; (b) the concessionaire will clean and Maintain all concession areas; (c) the concession-

aire will name the District and the Partnership as insureds, as appropriate, in all insurance policies required to be held by the concessionaire pursuant to the concession agreement and this Agreement; (d) the concessionaire and the Partnership will indemnify the District in connection with the operation of the concession areas and in connection with all concession activities; and (e) the Partnership will determine the prices and quality of the products sold by the concessionaire, and the Partnership will determine the products to be sold.

- 20.4 Additional Stands. The Partnership agrees to not construct or place concession stands, portable or other, in the Stadium that will unreasonably restrict the view of the Playing Field from the main concourse of the Stadium.
- 20.5 Student Movers Bar/Restaurant Operations. If an operator other than the concessionaire under the concession agreement in Section 20.3 is contracted with by the Partnership for operation of the Student Movers bar and restaurant and such operator requires support facilities additional to those encompassed in the Development Documents, the operator so contracted with shall be responsible for the cost of constructing and equipping such support facilities, subject to the provisions of this Agreement, including, but not limited to, obtaining District approval, where required, and the prohibitions against the incorporation of any additional facilities into the Leased Premises which would have a Material Adverse Effect on the District.
- 20.6 Environmental Responsibilities. The Partnership shall encourage its concessionaires to demonstrate an environmental awareness by managing their operations to include recycled and recyclable paper and plastic products and no toxic cleaners in their operations and otherwise practice on-going environmental awareness.

ARTICLE XXI

F.

ADVERTISING AND SIGNS

- Stadium Signs. Subject to the provisions of Sections 25.9 and 26.5 herein, the District shall provide the supports and power outlets for all advertising signs, an outdoor message board, and signs of a purely informational nature (such as exit signs and directional signs) in the designated locations specified in the Development Documents. Subject to the provisions of sections 21.4, 25.9 and 27.2 herein, the Partnership shall have the exclusive right to: (a) sell and at its sole expense create, install and maintain such advertising panels and maintain signs and other advertising on the Stadium scoreboard and to receive and retain all revenues therefrom; (b) at its expense, create and erect additional signs in the Stadium for such advertising which do not materially or adversely affect the Stadium structure or design integrity; (c) determine the size, form and content of such advertising; (d) ensure compliance with all applicable laws and regulations and obtain all necessary governmental approvals for such signs; and (e) at it's expense, create and erect additional informational and directional signs of a non-advertising nature; all only to the extent that such advertising and signs are not detrimental to the maintenance of the Stadium as a first-class facility, and only if such construction does not have a Material Adverse Effect on the District.
- 21.2 <u>District Advertising</u>. In its temporary advertising for Non-Baseball Events held by the District, if any, the District shall not violate any valid and binding exclusivity agreements relating to advertising in the Stadium entered into by the Partnership as to which the District has been given notice by the Partnership. The Partnership shall provide the District with a list of all such agreements annually, on October 1 of every year

during the Term and Grace Term, or more frequently, as the Partnership determines necessary.

- 21.3 Other Advertising. Subject to the provisions of Section 27.2, the Partnership may conduct, or permit to be conducted, in the Stadium, any and all other forms of advertising, including any advertising to be worn or carried by the Partnership's or any concessionaires' personnel, promotional events sponsored by advertisers, logos or other forms of advertising to be affixed to or included with cups, hats, t-shirts, and other concession items or giveaways, real time signage on television broadcasts, and any and all other forms of such advertising or promotion.
- 21.4 Approval Rights. The District shall have a right to approve the size and location of all advertising and signs located on the exterior of the Stadium structure or elsewhere on the Stadium Land outside of the Stadium, which approval shall not be unreasonably withheld. The Partnership shall submit a request for approval of any such proposed advertising or signs at least ninety (90) days prior to initiating the implementation of any such advertising or signs. Failure of the District to promptly object to any proposed advertising and/or signs within sixty (60) days after receipt of a request for approval therefore may be conclusively relied on by the Partnership as constituting the District's approval thereof. Nothing in this Section 21.4 or elsewhere in this Article XXI shall be construed as permitting the Partnership to erect additional signs on the exterior of the Stadium or billboards or other signs on the Stadium land, which is expressly prohibited by applicable laws, ordinances, rules, or regulations.

ARTICLE XXII

BROADCAST FEES

All rights and other fees and arrangements relating to the production and distribution of the Team's Major League Baseball

Games for commercial television, non-commercial television (by over-the-air, cable or otherwise), including direct sales of advertising by the Partnership, radio broadcast or any other media fees and revenues, and any income attributable to such broadcasts (whether in or out of the local market) ("Broadcast Fees") shall be retained and exclusively controlled by the Partnership. Broadcast Fees for Non-Baseball Events held by the Partnership shall be distributed according to Section 4.2(e). Broadcast Fees for Off-Season Non-Baseball Events held by the District shall be paid to, retained and exclusively controlled by the District.

ARTICLE XXIII

COVENANT OF QUIET ENJOYMENT

The District covenants that if, and so long as, the Partnership materially performs each and every covenant, agreement, term, provision and condition of this Agreement on the part and on behalf of the Partnership to be kept and performed, the Partnership shall quietly enjoy its rights under this Agreement without hinderance or molestation by the District or by any other person lawfully claiming the same by, through, or under the District, subject to the covenants, agreements, terms, provisions and conditions of this Agreement.

ARTICLE XXIV

ADDITIONAL OBLIGATIONS OF PARTNERSHIP

- 24.1 <u>Promotion</u>. The Partnership shall reasonably advertise and promote all Baseball and Non-Baseball Events it conducts at the Stadium.
- 24.2 Attendance of Children From Low Income Families. The Partnership has established and will maintain the Colorado Rockies Baseball Club Foundation, a Colorado non-profit corporation (the "Foundation"). One of the many purposes of the Foundation is to assure that children from low income families are able to attend

Major League Baseball Games. The Partnership shall be responsible for the operation of the Foundation and realization of its purposes. The District may participate in the activities of the Foundation.

- 24.3 Coors Field Video and Photograph Book. With the District's assistance and approval, the Partnership will produce, market and distribute a video tape and photograph book documenting the history and construction of Coors Field. The District shall receive fifty percent (50%) of the net profits from the sale of the video and book, payable annually on or before December 15th of each year during the Term and Grace Term.
- 24.4 <u>Baseball Museum</u>. The District and the Partnership, through the Foundation, will cooperate in the finishing of the Baseball Museum. The operations of the Baseball Museum shall be the responsibility of the Partnership. The District shall receive fifty percent (50%) of the net profits realized from museum operations, payable annually for the previous calendar year on or before January 30th of each year during the Term and Grace Term, which shall be contributed to the Foundation.
- 24.5 <u>Public Parking</u>. The Partnership shall allow a portion of the parking facilities located on the Stadium Land to be used for paid public parking on days on which no Baseball or Non-Baseball Events are being held on the Leased Premises and Partnership Property. The parking facilities to be used for such public parking shall be as agreed to by the District and the Partnership or, if no agreement can be reached, as reasonably designated by the District. In no event will the Stadium Land be used for public parking if it would result in a monetary loss to the Partnership or the District.

24.6 Parking Structure.

- (a) A feasibility study (to be paid for by the Partnership) will be conducted by a parking consultant acceptable to both the District and the Partnership to study whether or not, after the first season of play at Coors Field, there will be a need for additional parking and whether or not a new parking facility, as described below, would be economically feasible.
- (b) If the feasibility study is positive, all necessary governmental approvals secured, are and the participation will not have a Material Adverse Effect on the District, then the District and the Partnership shall each contribute 50% of cash necessary to design and construct a 900 to 1,000 - space parking structure on the block bounded by 19th, 20th, Wazee and Wynkoop Streets, subject to the availability of District The development may be commenced even if the District does not have funds available, if a means for the Partnership to finance the District's cash contribution can be mutually agreed to by the District and the Partnership.
- (c) Neither the District nor the Partnership shall encumber its interests in or revenues derived from the parking structure, or the parking structure itself without the prior written consent of the other. In calculating the total amount of necessary cash so as to determine each party's contribution requirement described above in subparagraph (b), the amount of any secured debt as described in the previous sentence shall not be included.
- (d) Notwithstanding the provisions of Section 4.2(c) or any other provisions of this Agreement to the contrary except subparagraph (e) below, the Partnership shall pay the District 50% of the net profits, as defined in Section 4.2(c), from the parking structure.

- The Partnership may request that the District contribute less than 50% of the cash necessary, as described above in subparagraph (b). In such event, the District's share of net profits shall equal its percentage of cash contributed, but in no event will it be less than the amount the District would have received had the parking structure not been built, which amount shall be computed for each particular event or use (i.e., Major League Baseball Games, Non-Baseball Events and all other parking uses) as the difference between: (1) the product of (i) 291 parking spaces, times (ii) the highest parking rate charged for the particular event or use, times (iii) the applicable percentage payable to the District as described in Section 4.2(c) for the particular event or use (i.e., 20% or 80%); less (2) the operating costs of the parking structure (before debt service depreciation) allocable to the 291 spaces for the particular event or use, as defined in Section 4.2(c) (i.e., 291/1,000 of the operating costs if the parking structure capacity is 1,000 spaces).
- (f) Parking operations in the parking structure shall be subject to the same audit provisions set forth in Section 4.2(c). The participation by the District and the Partnership in the construction of the parking structure and the payments to the District based upon the operation of the parking structure shall be structured to avoid the occurrence of a Material Adverse Effect upon the District.
- 24.7 <u>Commitment to Stadium</u>. Except as expressly provided in Section 3.2 and Articles IX and XVI, the Team shall play all Major League Baseball Games during the Term and Grace Term in the Stadium.

ARTICLE XXV

REPRESENTATIONS BY PARTNERSHIP

The Partnership represents and warrants as follows, as of the date hereof, and as to Sections 25.9 and 25.10 only at all times from and after the date hereof until the expiration or termination of this Agreement:

- 25.1 <u>Valid Existence</u>. The Partnership is a limited partnership duly organized and validly existing under the laws of the State of Colorado. The Partnership has full partnership power to own its property and conduct its business as presently conducted.
- . 25.2 Power: No Limitation on Ability to Perform. The Partnership has full partnership power and authority to execute and deliver this Agreement and to carry out and perform all of the terms and provisions of this Agreement, and all transactions contemplated hereby, to the extent required to be carried out or performed by the Partnership. Neither the Partnership's partnership agreement or certificate of limited partnership or the articles of incorporation or by-laws of Colorado Baseball 1993, Inc. (the "Corporation"), the Partnership's general partner, nor any rule, policy, constitution, by-law or agreement of the National League or the Office of the Commissioner of Baseball, nor any other agreement, law or other rule in any way prohibits, limits or otherwise affects the right or power of the Partnership or the Corporation to enter into and perform all of the terms and provisions of this Agreement and all transactions contemplated hereby. Neither the Partnership nor any of its partners (general or limited, including, without limitation, the Corporation), or stockholders of any corporate partner, officers, directors, or any of their personal or legal representatives are party to or bound by any contract, agreement, indenture, trust agreement,

obligation or other instrument which could prohibit, limit or otherwise affect the same. No consent, authorization or approval of (except the approval(s) of the National League and the Office of the Commissioner of Baseball already obtained by the Partnership), or other action by, and no notice to or filing with, any governmental authority, regulatory body or any other person is required for the due execution, delivery and performance by the Partnership and the Corporation of this Agreement or any of the transactions contemplated hereby.

- 25.3 <u>Valid Execution</u>. The execution and delivery of this Agreement by the Partnership has been duly and validly authorized by all necessary action. This Agreement will be a legal, valid and binding obligation of the Partnership, enforceable against the Partnership in accordance with its terms. The Partnership shall provide to the District a written resolution of the Partnership authorizing the execution of this Agreement.
- 25.4 <u>Defaults</u>. The execution, delivery and performance of this Agreement (a) do not and will not violate or result in a violation of, contravene or conflict with, or constitute a default under (i) any agreement, document or instrument to which the Partnership or the Corporation is a party or by which the Partnership's or the Corporation's assets may be bound or affected, (ii) any law, statute, ordinance or regulation applicable to the Partnership or the Corporation, or (iii) the articles of incorporation or by-laws of the Corporation, or the partnership agreement or certificate of limited partnership of the Partnership, and (b) do not and will not result in the creation or imposition of any lien or other encumbrance upon the assets of the Partnership or the Corporation.
- 25.5 <u>Good Standing of Corporation</u>. The Corporation is the Partnership's sole general partner, and is a corporation duly

organized, validly existing and in good standing under the laws of the State of Colorado. The Partnership shall provide to the District a certificate of good standing for the Corporation issued by the Colorado Secretary of State.

- 25.6 <u>Power of Corporation</u>. The Corporation has full corporate power and authority to execute and deliver this Agreement on behalf of the Partnership as the Partnership's sole general partner, and to carry out the terms and provisions of this Agreement, and all transactions contemplated hereby.
- 25.7 <u>Valid Execution by Corporation</u>. The execution and delivery of this Agreement by the Corporation on behalf of the Partnership as the Partnership's sole general partner has been duly and validly authorized by all necessary action. The Partnership shall provide to the District a written resolution of the Corporation authorizing the execution of this Agreement.
- 25.8 <u>Team Ownership</u>. The Partnership owns the Team as of the date hereof.
- 25.9 <u>Compliance With Laws</u>. The Partnership complies and shall comply, at all times, with all laws and regulations applicable to its use of the Leased Premises in accordance with the terms of this Agreement, and shall obtain licenses and permits (other than building permits and certificates of occupancy in connection with the construction thereof), necessary in connection therewith at its sole cost and expense.
- Partnership or its successors in interest is and will be the owner of the Franchise, through which the Team is authorized to play Major League baseball within the boundaries of the District, and the Partnership or its successor in interest shall maintain the Franchise in good standing with the National League and the Office of the Commissioner of Baseball.

ARTICLE XXVI

REPRESENTATIONS BY DISTRICT

The District represents and warrants as follows, as of the date hereof and at all times from and after the date hereof until the expiration or termination of this Agreement:

- 26.1 <u>Valid Existence</u>. Pursuant to §32-14-104 of the Act, the District is a body corporate and politic and a political subdivision of the State of Colorado.
- \$32-14-107(d), (g) and (h) of the Act, the District, through its Board of Directors, has the power and authority to execute and deliver this Agreement and to carry out and perform all of the terms and provisions of this Agreement, and all transactions contemplated hereby, to the extent required to be carried out or performed by the District. The District is not bound by any contract, agreement, indenture, trust agreement, note, obligation or other instrument which could prohibit, limit or otherwise affect the same.
- 26.3 <u>Valid Execution</u>. The execution and delivery of this Agreement by the District has been duly and validly authorized by all necessary action. This Agreement will be a legal, valid and binding obligation of the District, enforceable against the District in accordance with its terms. The District will provide to the Partnership a written resolution of the District authorizing the execution and delivery of this Agreement.
- 26.4 <u>Defaults</u>. The execution, delivery and performance of this Agreement (a) does not and will not violate or result in a violation of, contravene or conflict with, or constitute a default under: (i) Any agreement, document or instrument to which the District is a party or by which the District's assets may be bound or affected, or (ii) any law, statute, ordinance or regulation

- reaction or imposition of any lien or other encumbrance upon usets of the District.
 - 26.5 <u>Compliance With Laws</u>. The District shall comply at all with all laws and regulations applicable to its construction of the Leased Premises in accordance with the terms of this cent.

ARTICLE XXVII

MISCELLANEOUS

- Partnership shall have the right to sell the name of the dium in perpetuity. The District approves the name "Coors for the Stadium and has the right to approve any other name attributed to the Stadium, which approval shall not be unressonably withheld.
- 27.2 Architectural Drawings and Logo. The District owns architectural renderings of the Stadium, architectural drawings of the Stadium and the Stadium Land, and the Coors Field logo currently used by the District. The Partnership will contribute four percent (4%) of the gross sales amount of all retail sales derived from products utilizing one or more of such renderings, drawings or logo alone, and two percent (2%) of the gross sales amount of all retail sales derived from products utilizing one or more of such renderings, drawings or logo in conjunction with the Team's logo(s), to, at the option of the District, the Foundation or the Colorado Rockies Charity Fund, a fund of the Robert R. McCormick Tribune Foundation, in the name of the District. District hereby licenses the Partnership to use such renderings, drawings or logo in the sale of products during the Term and Grace Term.

If the District elects to create and market products utilizing its renderings, drawings or logos, it will negotiate with the Partnership and attempt to reach an agreement to license such products to the Partnership for production, marketing and sale. If no agreement can be reached, the District may license such products to any other entity with whom the District can reach an agreement regarding the production, marketing and sale of such products.

- 27.3 Entire Agreement. This Agreement contains the entire agreement between the parties hereto with respect to the subject matter hereof, and there are no promises, agreements, conditions, undertakings or warranties or representations, oral or written, express or implied, between them other than as herein set forth or as specifically referred to herein. This Agreement is intended to be an integration of all prior or contemporaneous promises or agreements, conditions or undertakings between the parties hereto. The District and the Partnership have jointly participated in the drafting of this Agreement and any ambiguity contained in this Agreement shall not be construed against the District or the Partnership solely by virtue of the fact that either the District or the Partnership should be considered the drafter of this Agreement.
- Default Interest, Attorneys' Fees. Except for advances, all amounts payable hereunder by either party shall, from and after the date on which such party is in default in the payment thereof, bear interest at the rate of four percent (4%) per annum in excess of the published prime rate charged from time to time by the Colorado National Bank or its successor bank(s). Should any party hereto be required to enforce any of the provisions of this Agreement as between itself and the other party hereto, through legal proceedings in a court of law or otherwise, the prevailing

party shall be entitled to receive all of its costs and expenses of enforcement including reasonable attorneys' fees.

27.5 Notices. All notices, consents, approvals, demands and submissions (hereinafter collectively referred to as a "Notice") shall be in writing and shall be served as provided in this Article XXVII (except as otherwise provided in this Agreement). Any notice to the District shall be deemed properly given via first class, registered or certified mail, postage prepaid, or if delivered personally (or by bonded courier) to the District at 1660 - 17th Street, Suite 100, Denver, Colorado, 80202 (or other address designated by notice so given), to the attention of the Executive Director of the District or to the District offices at the Stadium addressed to the attention of the individual regularly in charge thereof, with a copy to the District's general counsel, Fairfield and Woods, P.C., 1700 Lincoln Street, Suite 2400, Denver, Colorado 80203, Attention Jac Sperling. The District shall notify the Partnership of the name of such individual in charge of the District offices at the Stadium and Notices may thereafter be mailed or delivered to such individual until the Partnership has been notified in writing by the District that some other individual is in charge of the District offices and authorized to receive Notices. Any notice to the Partnership shall be deemed properly given via first class, certified or registered mail, postage prepaid, or if delivered personally (or by bonded courier) to the Partnership at 1700 Broadway, Suite 2100, Denver, Colorado, 80290, or its business offices in the Stadium (or other address designated by notice so given), addressed to the attention of the Chairman of the Board, President and Chief Executive Officer of the Corporation or to any other employee of the Partnership whom the Partnership has designated in a Notice to the District as an individual authorized to receive Notices hereunder. Any notice shall be

deemed to have been given three days after postmarked or when personal service is effected or delivery is refused.

- of God, earthquake, flood, epidemic, landslide, war, riot, civil commotion, general unavailability of certain materials, strike (except as set forth below), slowdown or labor dispute (any of the foregoing hereinafter referred to as "Force Majeure") prevent performance of this Agreement in accordance with its provisions, performance of this Agreement by either party shall be suspended or excused to the extent commensurate with such interfering occurrence, except that the Partnership shall still be obligated for payments to the District pursuant to Section 4.2(a) and payments to the Repairs Fund required by Section 8.11(a).
- 27.7 Reasons for Disapproval. Whenever in this Agreement a party is given the right to approve or disapprove any matter or action, such approval or disapproval shall not be unduly delayed and any disapproval shall be reasonable and shall state the reasons therefor, except where it is specifically provided that a party may withhold approval or otherwise act in its sole discretion. Notwithstanding, the District shall not be deemed to have unreasonably withheld its approval if the District disapproves because it determines, in its sole discretion, that such approval could result in a Material Adverse Effect upon the District.
- 27.8 <u>Successor Bound</u>. The covenants, terms, provisions and conditions of this Agreement shall be binding upon and inure to the benefit of the District and the Partnership and their respective successors and, to the extent permitted herein, assigns.
- 27.9 Governing Law. This Agreement is made, and shall be construed, under the laws of the State of Colorado.
- 27.10 <u>Captions and Headings</u>. The captions and headings throughout this Agreement are for convenience and reference only

and the words contained therein shall in no way be held or deemed to define, limit, describe, explain, modify amplify or add to the interpretation, construction or meaning of any provisions of this Agreement or the scope or intent thereof, nor in any way affect this Agreement.

- 27.11 <u>Plural/Singular</u>. Wherever appropriate herein, the singular includes the plural and the plural includes the singular.
- 27.12 Execution of Counterparts. This Agreement may be simultaneously executed in several counterparts, each of which shall be an original and all of which shall constitute but one and the same instrument.
- 27.13 Further Assurances. The Partnership and the District shall execute, acknowledge and deliver, after the date hereof, without additional consideration, such further assurances, instruments and documents and shall take such further actions, as the District or the Partnership shall reasonably request of the other in order to fulfill the intent of this Agreement and the transactions contemplated thereby.
- 27.14 <u>Anti-Discrimination Clause</u>. The District and the Partnership shall comply with all applicable state, local and federal laws, rules, regulations, executive orders and agreements pertaining to discrimination in employment, unlawful employment practices and affirmative action.
- 27.15 <u>Material Adverse Effect</u>. The Partnership will not take, permit or cause any action under this Agreement that would have a Material Adverse Effect on the District. "Material Adverse Effect" shall mean an event, action or inaction which results in any of the following:
 - the making of any payment prohibited by Section 19.1;

- 2. causing a default under any agreement to which the District or its property is bound including, but not limited to, this Agreement, and of which the Partnership has actual knowledge;
- 3. causing the District to be in violation of any law, regulation, or court or administrative ruling applicable to it or its property;
- 4. causing the District to breach its covenants to maintain the tax-exempt status of the Bonds, or otherwise breach any agreements relative to the Bonds.
- 27.16 Amendment: Waiver. No alteration, amendment modification hereof shall be valid unless executed by an instrument in writing by the parties hereto with the same formality as this The failure of the Partnership or the District to insist in any one or more instances upon the strict performance of any of the covenants, agreements, terms, provisions or conditions of this Agreement or to exercise any election herein contained shall not be construed as a waiver or relinquishment for the future of such covenant, agreement, term, provision, condition, election or option, but the same shall continue and remain in full force and effect. No waiver by the Partnership or the District of any covenant, agreement, term, provision or condition of this Agreement shall be deemed to have been made unless expressed in writing and signed by an appropriate official on behalf of the District or the Partnership. The payment by either party of sums due and payable hereunder, with knowledge of the breach of any covenant, agreement, term, provisions or condition herein contained, shall not be deemed a waiver of such breach.
- 27.17 <u>Severability</u>. If any article, section, subsection, term or provision of this Agreement or the application thereof to any party or circumstance shall, to any extent, be invalid or unenforceable, the remainder of the article, section, subsection,

term or provision of this Agreement or the application of same to parties or circumstances other than those to which it is held invalid or unenforceable shall not be affected thereby and each remaining article, section, subsection, term or provision of this Agreement shall be valid and enforceable to the fullest extent permitted by law.

- 27.18 No Joint Venture. Nothing in this Agreement shall be construed as creating a joint venture, partnership, or any other association between the District and the Partnership. To the extent that the Partnership performs Stadium Management activities hereunder, the Partnership shall be an independent contractor of the District.
- 27.19 Words of Limitation. Whenever the words "including but not limited to" or "by way of example but not limitation" or any other similar prefatory words are used throughout this Agreement, such words shall be deemed to preface an example or list of examples, which example(s) are set forth for informational purposes only and not for purposes of limitation.
- agree that the enforcement of the terms and conditions of this Agreement, and all rights of action relating to such enforcement, shall be strictly reserved to the District and the Partnership, or their successors or assigns, and nothing contained in this Agreement shall give or allow any such claim or right of action by any third person or entity whatsoever on such Agreement. It is the express intention of the District and the Partnership that any person or entity other than the District or the Partnership, or their successors or assigns, receiving services or benefits under this Agreement shall be deemed to be an incidental beneficiary only.

IN WITNESS WHEREOF, the Parties have executed this Amended and Restated Lease and Management Agreement as of the date first above written.

COLORADO ROCKIES BASEBALL CLUB, LTD., a Colorado Limited Partnership, formerly known as Colorado Baseball Partnership 1993, Ltd.

By: COLORADO BASEBALL 1993, INC.,

as General Partner

Jerry D. McMorris

Chairman, Fresident and Chief

Executive Officer

DENVER METROFOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT

R. G. Galler

Chairman of the Board of Directors

INDEX OF DEFINED TERMS

ACT	
Attendee	• •
Average Ticket Price	1
Average Ticket Price Board Bonds	1
Bonds	. :
Bonds Broadcast Fees Commandance Date	
Commencement Date	9:
Commencement Date Condemnation Corporation	1:
Corporation	8:
Corporation Damaged Property Defeasance Date Deterioration	95
Defeasance Date	4 (
Deterioration	86
Deterioration .	38
Development bocuments	. 3
District Default	. 3
Development Documents District District Default Equipment Lessors	74
Equipment Lessors	10
Equipment Lessors First Extension Term Force Majoure	12
Force Majeure	102
Foundation	91
	. 1
Grace Term	13
Guaranteed Minimum Payment	17
Improvement(s)	40
Leased Equipment	חד
Grace Term Guaranteed Minimum Payment Improvement(s) Leased Equipment Legislature Maintenance (Maintain) Major League Baseball Game	1
Maintenance (Maintain)	ं २ व
Major League Baseball Game	16
MOTOT DEGULE MASSUSTI KOSOON	
Material Adverse Affect Material part National League Non-Baseball Event	103
Material part	בטו
National League	7
Non-Baseball Event	, <u>.</u>
Non-Baseball Event Notice Off-Season	20
Off-Season	-0T
Partnership	20
Partnership Default	. T
Partnership Property	70
Repair Area	- 7
Repair(s)	39
Off-Season Partnership Partnership Default Partnership Property Repair Area Repair (s) Repairs Fund	38
Second Extension Term	62
Caulin Cino	
Stadium Land	22
Stadium Management	2

Suite	•	•	•		•										:						L		_	_		_	25
Team "									 _		_			_							1		Ī	Ĭ.	•	•	- 1
rerm	•	٠	-												_	_	_	_			:_	~ -	_	_			7.7
Third	E۶	ct e	ens	310	on	$T\epsilon$	3II	n																-	_	•	1.2
Third Extension Term Transportation Management									Da	200	77	3777			_	-	_	-	-	·	-	-	•	-	•	42	

EXHIBIT A

STADIUM LAND

್ಷಣೆ ಆಲ್ವಾ

EXHIBIT B

Map

==:

EXHIBIT D

DISTRICT'S CONSTRUCTION OBLIGATIONS

EXHIBIT E

<u>.</u> :

PARTNERSHIP PROPERTY

EXHIBIT G

TRANSPORTATION NANAGEMENT PEOGRAM

FIRST AMENDMENT TO THE MARCH 30, 1995

AMENDED AND RESTATED LEASE AND MANAGEMENT AGREEMENT BY AND BETWEEN

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT AND COLORADO ROCKIES BASEBALL CLUB, LTD.

THIS First Amendment to the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. ("Amendment") is dated this 31st day of December, 1995.

RECITALS:

- A. All defined terms set forth in the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball District and Colorado Rockies Baseball Club, Ltd. dated March 30, 1995 ("Agreement") shall have the same meanings ascribed to such terms in the Agreement for the purposes of this Amendment.
- B. The first Major League Baseball Season in the Stadium commenced on March 31, 1995 and ended on October 4, 1995. Subsequent to that Major League Baseball Season, the District and the Partnership have determined that certain payment schedules, notice periods and other various provisions of the Agreement need modification, and the District and the Partnership have agreed to amend such provisions.
- C. The District and the Partnership also agree that it is in the best interests of the District and the Partnership to increase the potential for a variety of uses for the Stadium during the Major League Baseball Seasons throughout the Term and Grace Term.

AGREEMENT

NOW, THEREFORE, in consideration of the premises, the consideration set forth in the Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the District and the Partnership hereby agree to amend and restate those portions of the Agreement set forth in Paragraph 1 of this Amendment and to further amend (without restatement) the Agreement as set forth in Paragraphs 2 through 4 of this Amendment. For clarity purposes, any portion of any Article, Section or Subsection of the Agreement which is being amended by Paragraph 1 of this Amendment will be restated in such Paragraph 1. Therefore, each

restated Article, Section or Subsection of the Agreement set forth in Paragraph 1 below shall operate to supersede in its entirety the current Article, Section or Subsection of the same number in the current Agreement. No portion of any Article, Section or Subsection of the Agreement shall be amended by Paragraph 1 hereof unless such portion thereof is set forth in such Paragraph 1.

1. The following portions of the Agreement shall be and are hereby amended and restated in their entirety:

TITLE: AMENDED AND RESTATED USER AGREEMENT, LEASE AND MANAGEMENT AGREEMENT BY AND BETWEEN DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT AND COLORADO ROCKIES BASEBALL CLUB, LTD.

ARTICLE I. LEASED PREMISES:

In consideration of the mutual agreements contained in this Agreement, incorporating herein the recitals set forth above, and subject to the limitations set forth in Article V concerning permitted purposes and periods of use, the District hereby leases to the Partnership, and the Partnership hereby leases from the District, the real property described in Exhibit A attached hereto and made a part hereof (the "Stadium Land"), together with all the improvements to be constructed thereon, including but not limited to the stadium constructed thereon, and all improvements, additions, alterations, fixtures, equipment, and installations owned or leased by the District and constructed, provided or added to the stadium at any time (the "Stadium"), and all other improvements, additions, alterations, fixtures, equipment and installations owned or leased by the District and constructed, provided or added to the Stadium Land at any time (collectively, the "Leased Premises"). A map of the Leased Premises delineating the Stadium and the Stadium Land is attached hereto as Exhibit B.

SECTION 3.1:

above written upon execution of this Agreement by the District and the Partnership. The Partnership's right to occupy the Leased Premises subject to the limitations set forth in Article V concerning permitted purposes and periods of use shall commence upon substantial completion of construction of the Stadium, permitting the Partnership to use the Stadium for its intended use (the "Commencement Date"). Subject to the provisions of Section 3.3, the Partnership's right to occupy the Leased Premises as set forth in Article V hereof shall end twenty-two (22) years after the Commencement Date. Pursuant to the May 15, 1995 "Supplement (Regarding Expiration Date)" to the Lease, the Commencement Date of this Agreement is March 31, 1995. The last day of such twenty-two (22) year period of occupancy shall be March 30, 2017.

INTRODUCTORY PARAGRAPH TO SECTION 4.2:

4.2 Payments to the District. The Partnership will pay to the District the following payments (which payments shall not be set-off against any sums due to the Partnership from the District hereunder) for the Term and Grace Term:

SUBSECTION 4.2(c):

(c) Subject to the provisions of Section 24.6, all parking net profits (net profits shall equal gross receipts less operating costs; operating costs are all direct incremental costs paid to or retained by third parties or Affiliates of the Partnership, provided costs attributable to any Affiliates of the Partnership are competitive with the costs third parties would charge for providing the same services, as a direct result of parking operations) shall be shared on a twenty percent (20%) (to

Baseball Games played during the Term and Grace Term and Non-Baseball Events for which the Partnership is entitled to use the Leased Premises pursuant to Section 5.1. Subject to the provisions of Section 24.6, all other parking net profits shall be shared on an eighty percent (80%) (to the District), twenty percent (20%) (to the Partnership) basis. Payment shall be made to the District, along with a written report itermizing gross receipts and operating costs, after the conclusion of the 1995 Major League Baseball Season, and thereafter on February 15 (for the period beginning with the day following the end of the immediately preceding Major League Baseball Season and ending December 31), August 15 (for the period beginning January 1 and ending June 30) and December 15 (for the period beginning July 1 and ending with the last day of the immediately preceding Major League Baseball Season) of each year during the Term and Grace Term.

The Partnership shall ensure that the operator of the parking on the Stadium Land, including, but not limited to, the parking structure, if any, provides an annual accounting of all revenues from the previous year, to be provided on or before January 31 of each year during the Term and Grace Term beginning the year after the year of the Commencement Date. If necessary, the amounts paid to the District pursuant to this Section and Section 24.6 shall be adjusted annually pursuant to the results of the accounting, with any such adjustment reflected in the payment due on the February 15 next following.

SUBSECTION 4.2(d):

(d) The Student Movers Building located between 21st and 22nd on Blake Street has been designed to become an integral part of the Stadium and is included in the term "Stadium". All or part of the first floor and basement of that building will be incorporated into the Stadium design as a bar and restaurant for use by fans during Major League Baseball Games and potentially year-round. As part

of the construction of the Stadium, the District will provide, at its expense, a shell space, as more particularly described in Exhibit E attached hereto, on the first floor and basement of the Student Movers building to be used for a bar and sestaurant. The District will receive from the Partnership three percent (3%) of monthly gross sales of the bar and restaurant if the Partnership contracts with the concessionaire under the concession agreement described in Section 20.3 for operations. If the Partnership contracts with any operator other than such concessionaire, the District shall receive from the Partnership five percent (5%) of the monthly gross sales of the bar and restaurant. Payment shall be made by the Partnership to the District, along with a written report of gross sales, after the conclusion of the 1995 Major League Baseball Season, and thereafter on February 15 (for the period beginning with the day following the end of the immediately preceding Major League Baseball Season and ending December 31), August 15 (for the period beginning January 1 and ending June 30) and December 15 (for the period beginning July 1 and ending with the last day of the immediately preceding Major League Baseball Season) of each year during the Term and Grace Term.

The Partnership shall ensure that the concessionaire operating the bar and restaurant provides an annual accounting of all revenues from the previous year, to be provided on or before January 31 of each year during the Term and Grace Term beginning the year after the year of the Commencement Date. If necessary, the amounts paid to the District pursuant to this subsection shall be adjusted annually pursuant to the results of the accounting, with any such adjustment reflected in the payment due on February 15 next following.

SUBSECTION 4.2(e):

(e) The Partnership will pay the District an amount equal to forty percent (40%) of the net profit (defined as gross receipts from all sources less operating costs - operating costs are all incremental out-of-pocket costs (including

costs of goods sold) paid to third parties or Affiliates of the Partnership, provided costs attributable to any Affiliates of the Partnership are competitive with the costs third parties would charge for providing the same services, as a direct result of the event) from any and all of the Non-Baseball Events allowed to be sponsored by the Partnership and from any and all other events held by the Partnership on the Leased Premises (except for Major League Baseball Games or any event held immediately preceding or following a Major League Baseball Game) from which the Partnership realizes revenue. These payments shall be submitted along with a written report itemizing gross receipts and operating costs within thirty (30) days after each such Non-Baseball Event held by the Partnership.

SUBSECTION 4.2(f):

(f) The Partnership will remit to the District 100% of the gross receipts from any and all Non-Baseball Events held by the District pursuant to Section 5.2(d), including, without limitation, gross receipts from novelty and concession sales, after deducting (1) operating costs (as defined in Section 4.2(e) above) paid or incurred by the Partnership as a direct result of the event, plus (2) the "managerial and operational costs" for which the Partnership is entitled to reimbursement pursuant to Section 5.2(d) hereof, plus (3) the "management fee," if any, payable to the Partnership pursuant to Section 5.2(d) hereof, plus (4) any amounts for which the Partnership is entitled to payment or reimbursement pursuant to Sections 6.13, 8.2(i), 8.3(b)(ii) or 8.10(d). Payments shall be submitted within ninety (90) days after the conclusion of each Non-Baseball Event held by the District along with a written report itemizing gross receipts, deductible items being currently deducted from gross receipts and a reasonable description and good faith estimate of the nature and amount of deductible items not determinable at the time the written report is submitted. The District shall reimburse the Partnership for any excess of operating costs over gross receipts from any such event and for any deductible costs not determinable until after the written report is submitted to the District.

TITLE TO ARTICLE V: USE OF STADIUM AND LIMITATIONS (USER AGREEMENT).

SECTION 5.1:

- 5.1 Partnership's Use. Commencing with the Commencement Date and continuing for the balance of the Term and Grace Term, the Partnership shall, subject to all limits and conditions imposed upon the Partnership in this Agreement (specifically including but not limited to the limitations on the time and duration of such use set forth below in this Section 5.1), be entitled to use the Leased Premises for the following purposes:
- (a) Playing of Major League Baseball Games and activities related to the playing of Major League Baseball Games and the maintenance of a Major League Baseball team, including, but not limited to: training, practices, maintenance and preparation of the Leased Premises to suit such purposes, advertising and marketing of games and ticket sales;
- (b) the operation of club/restaurant facilities to be known as the "Stadium Club", a separate "Club Seating Lounge", and the restaurant and bar facilities to be constructed in the Student Movers Building:
 - (c) the operation of the Partnership's general offices;
- (d) the sale of food and alcoholic and non-alcoholic beverages, souvenirs and other items normally considered "concessions";
- (e) the sale of space on or for advertising signs and billboards as provided in the Development Documents, subject to the provisions of Article XXI;

- (f) any and all other activities which, from time to time during the Term and Grace Term, are associated with, or are conducted in connection with, or are related to, the conduct of the business of a Major League Baseball team; and
- (g) up to five (5) Non-Baseball Events annually. The Partnership may, from time to time, request from the District on a specific event basis an increase in the number of Non-Baseball Events which the Partnership may sponsor annually, and the District will review such request and determine in its sole discretion whether to grant the request.

The Partnership's possession and use of the Leased Premises shall be subject to the following limitations:

- (1) The Partnership shall not occupy or use the Leased Premises or any portion thereof (or permit the use or occupancy of the Leased Premises) for any purpose or in any manner which is unlawful or which violates this Agreement or any other agreement binding upon the District, a copy of which has been provided to the Partnership or law under which the District or its property is bound.
- Event proposed to be held by the Partnership, which approval will not be unreasonably withheld. The Partnership shall provide written notice to the District, no later than sixty (60) days prior to the occurrence of any Non-Baseball Event, of its intent to hold such an event. The District shall approve or disapprove such Non-Baseball Event within thirty (30) days after receipt of such notice by the Executive Director of the District. The District shall not be deemed to have unreasonably withheld its consent to a Non-Baseball Event if the District shall, in good faith, determine that such use:
 - (i) is likely to cause material damage to the Leased Premises;
 - (ii) is incompatible with the character of the neighborhood in which the Leased Premises is located:

- (iii) is likely to tamish the image or reputation of either the Leased Premises or the District; or
- (iv) is in violation of any applicable law, ordinance, rule or regulation, of any governmental agency with jurisdiction over the Stadium or its use.

This District shall develop a policy and procedure regarding its internal approval process which shall not be inconsistent with the provisions of this Agreement. The District shall advise the Partnership, in writing, of the provisions of such policy and procedure, and any amendments thereto.

- Game to be played in any facility other than the Leased Premises without first obtaining the written approval of the District, which may be withheld in its sole and absolute discretion. This subsection shall not require the Team to play any Major League Baseball Game in the Stadium if any governmental agency with jurisdiction over the Stadium and its use prohibits such use on the basis of health or safety risks attendant to such use.
- (4) The Partnership shall be responsible at its own expense for advertising and promoting events it conducts on the Leased Premises.
- (5) The Partnership's permitted uses of the Leased Premises shall be limited to those times and those portions of the Leased Premises set forth in this Section 5.1(5). Aside from those times and those portions of the Leased Premises granted to the Partnership hereby, the District, as the owner of the Leased Premises, shall have the exclusive right to use the Leased Premises subject to any obligations and limitations placed upon the District's use thereof by this Agreement during times when the Partnership has no user rights thereto. The Partnership's use of the Leased Premises (in the Partnership's capacity as the "Tenant" or "User" hereunder) shall be limited to the following times and portions:
 - (i) The Partnership shall have year-round use of the following portions of the Leased Premises:

21

- That portion of the Leased Premises used as general offices by the Partnership, including but not limited to the ticketing area/windows located on Blake Street and any other space designated for the Partnership's general office use by this Agreement;
- The basement plus floors 1, 2, 3 and 4 of the Student Movers Building (subject to the District's right to use office and conference room space on the fourth floor thereof pursuant to this Agreement);
- The Stadium Club and the Club Seating Lounge (now known as the Mountain Ranch Club) and adjacent areas of the Club Level concourse above Sections 214-227 of the seating bowl of the Stadium;
- The home clubhouse and all associated facilities (e.g., offices for team managers and coaches, locker facilities, training room, indoor batting cage, indoor bull pen, weight room, whirlpool and shower/bathroom facilities);
- The scoreboard operations room;
- Concessionaire's offices located under the right field bleachers of the Stadium;
- The Diamond Dry Goods Store located at 2151 Blake Street; and
- Sufficient parking on the Stadium Land to permit normal business operations of the

Partnership's general offices and the restaurants and clubs operating on the Leased Premises as agreed upon by the District and the Partnership.

- (ii) The Partnership shall have the right to use the following portions of the Leased Premises ONLY from 6:00 a.m. local time on those dates when a Major League Baseball Game is scheduled at the Stadium and those dates when a Non-Baseball Event sponsored by the Partnership is scheduled to be held at the Stadium pursuant to Section 5.1(g) hereof to 2:00 a.m. local time of the following day:
 - The Playing Field (as defined in Section 8.3(a) of this Agreement);
 - The seating bowl of the Stadium (including all seats, stands, boxes, concourses and aisles);
 - The concession areas and all associated facilities; and
 - All other areas of the Stadium, Stadium Land and/or Leased Premises necessary to conduct the Major League Baseball Game or the Non-Baseball Event sponsored by the Partnership (including but not limited to parking areas).
- (iii) The Partnership shall have the right to use the following portions of the Leased Premises ONLY from 6:00 a.m. to 10:00 p.m. local time on those dates during any post-season playoff series when off-day workouts and/or practice sessions are conducted by either team involved in such playoff series (up to a maximum of five (5) days for all playoff series combined):
 - The Playing Field;

7

- The visitors clubhouse and all associated facilities;
- The seating bowl; and
- The press box.
- (iv) The District hereby grants to the Partnership a license to use the following portions of the Leased Premises during the specified time periods in order to complement the other user rights of the Partnership set forth herein:
 - Periodic use, as and when required, of the General Manager's Suite and the Business Operations Suite and the conference room adjacent thereto;
 - Casual access to the seating bowl of the Stadium as and when required to show seat locations and box locations/amenities to prospective purchasers or current ticket holders and their guests and to provide tours of the Stadium to guests of the Partnership;
 - Periodic use (but no more than 5 days per Major League Baseball Season) of the Playing Field for off-day workouts and/or practice sessions for the baseball team during the Major League Baseball Season; provided that such off-day workouts and/or practices shall not conflict with any District sponsored Non-Baseball Event; and
 - All necessary routes of ingress to and egress from the Leased Premises required by all uses

granted to the Partnership pursuant to this Agreement.

The license rights granted by this Section 5.1(5)(iv) may be revoked or modified only by a written amendment to this Agreement executed by both the District and the Partnership. The "periodic use" and "casual access" described immediately above are subject to the District's exclusive right to use the Leased Premises or portions thereof when such areas are not being used by the Partnership pursuant to Sections 5.1(5)(i) through (iii) above.

baseball players who are on the then current roster of the Team, casual access by such major league baseball players who are on the current roster of the Team (specifically excluding all other persons) to the Playing Field to conduct informal workouts only during the Off-Season and only when no other activities of any kind or nature are scheduled thereon by either the District or the Partnership and subject to the discretion of the Partnership (or the then current "Manager" of the Stadium, if not the Partnership) in conducting its management of the Stadium pursuant to Article VIII. The District and the Partnership hereby agree and acknowledge that the license granted in this Section 5.1(5)(v) is not intended to allow the Partnership any additional user rights to the Stadium other than those set forth in Sections 5.1(5)(i) through (iv) above.

Except for the foregoing limitations and subject to the provisions of Section 5.2 and all other provisions of this Agreement, the Partnership shall have the right during those time periods specified in Sections 5.1(5)(i) through (iii) above to determine how the Leased Premises (or the applicable potions thereof) shall be used and who may be present therein, and, further, to select, competitively, and employ, concessionaires, licensees, or subtenants with respect to any Stadium operations,

including, but not limited to parking lots, concession areas, Stadium Club, Suite catering or advertising space in the Stadium. Subject to the provisions of this Agreement (including but not limited to Section 4.2 and Article XXII), the Partnership shall have the right to proceeds from the conduct of any of the activities permitted to be conducted by the Partnership, including but not limited to radio and television fees and revenues, signage and advertising sales, concession and vending machine rents or payments, sales of souvenirs, sales of food and beverage (including alcoholic beverages), and club membership fees related to any or all of the foregoing upon terms and conditions deemed acceptable by the Partnership, provided that no contract entered into by or on behalf of the Partnership shall impair any right of the District hereunder, impose any contractual liability on the District or have a Material Adverse Effect on the District.

SUBSECTION 5.2(d):

Leased Premises, or any portion thereof, not occupied by the Partnership pursuant to Section 5.1, subject to the limitations of this Section 5.1(d). The District may use the Stadium (or any portion thereof) only for Non-Baseball Events sponsored by the District and for other de minimis uses that do not interfere with the uses permitted by the Partnership under Section 5.1. In the case of Non-Baseball Events to be held by the District, the District will be required to "ensure the integrity of the playing field" (as defined below) in the Stadium to the reasonable satisfaction of the Partnership. The District shall notify the Partnership at least forty-five (45) days in advance of any Non-Baseball Event, which notice will allow the Partnership to reasonably reach a decision as to whether such Non-Baseball Event will fail to ensure the integrity of the playing field, in which case such Non-Baseball Event will be disallowed, or, if the Partnership concludes that such Non-Baseball Event does ensure the integrity of the playing field, to reach a good faith estimate of managerial

and operational costs (specifically including any increased costs of Maintenance or Repair of the Leased Premises as described in Section 8.10(d) hereof) that may be incurred as a result of such Non-Baseball Event (or a combination of more than one Non-Baseball Event, if applicable), and to coordinate with the District in the scheduling of all such events so as not to interfere with the Partnership's rights and duties as a "Tenant," as a "User" and as a "Stadium Manager" under this Agreement. Within fifteen (15) days after receipt of such notice the Partnership shall advise the District if it reasonably believes that the proposed Non-Baseball Event will fail to ensure the integrity of the playing field or if the proposed Non-Baseball Event cannot be held due to a conflict with a Major League Baseball Game or a Partnership Non-Baseball Event previously approved by the District. (In determining whether a proposed Non-Baseball Event will conflict with a Major League Baseball Game, the District hereby acknowledges that the Partnership is entitled to assume for any Major League Baseball Season that the Team will participate in all applicable post-season playoff series for that season, as and when each game of each such series may be scheduled by Major League Baseball, unless and until the Team is mathematically excluded from participation in or is eliminated from such playoffs.) The failure by the Partnership to so notify the District shall be deemed an affirmative statement by the Partnership that it does not reasonably believe that the proposed Non-Baseball Event will harm the integrity of the playing field and that no such scheduling conflict exists. If the Partnership does not so notify the District, within thirty (30) days after receipt of the notice from the District provided for in this section 5.2(d) the Partnership shall provide the District with a good faith estimate of managerial and operational costs (specifically including any increased costs of Maintenance or Repair as described in Section 8.10(d) hereof) which will be incurred by the Partnership as a result of any Non-Baseball Event (or a combination of more than one such event, if applicable), which shall not include prorated salaries or incremental payments to employees of the Partnership or its Affiliates not specifically incurred as a result of such event. Within thirty (30) days after receipt of an itemized

statement of such actual costs from the Partnership, the District shall reimburse the Partnership for all reasonable operational and managerial costs incurred as a result of any Non-Baseball Event sponsored by the District (except for any such costs that have been deducted by or reimbursed to the Partnership pursuant to Section 4.2(f) hereof). In addition, the District shall pay the Partnership a management fee for each Non-Baseball Event sponsored by the District equal to 10% of the sum of the costs that the Partnership is entitled to deduct under Section 4.2(f)(1) and (2), provided that the management fee shall not in any event exceed 50% of the net profit (if any) from the Non-Baseball Event (calculated in the same manner as is described in Section 4.2(e) for Partnership-sponsored Non-Baseball Events). A District-sponsored Non-Baseball Event "ensures the integrity of the playing field" if and only if (1) the Partnership concludes in the exercise of reasonable discretion that, after completion of the Non-Baseball Event and prior to the commencement of the next Major League Baseball Game, the playing field can and will be restored to levels of appearance, physical condition and playability consistent with first-class major league standards, and (2) the District agrees to pay the costs and expenses required to complete such restoration as provided in Section 8.3(b)(ii). The Partnership's judgment as to whether a proposed Non-Baseball Event ensures the integrity of the playing field shall be accorded substantial deference. The District hereby acknowledges that any Non-Baseball Event requested by the District during the "Off-Season" (which Off-Season shall commence upon the expiration of any Major League Baseball Season and end fifteen (15) days prior to commencement of the following Major League Baseball Season) is less likely to be rejected by the Partnership as failing to ensure the integrity of the playing field than those Non-Baseball Events requested during the Major League Baseball Season, all other factors being equal between the proposed Non-Baseball Events, due to the tight scheduling of Major League Baseball Games during the Major League Baseball Season. However, the Partnership acknowledges the District's ownership rights of the Leased Premises and agrees to carefully evaluate any Non-Baseball Event requested by the District during any Major League

Baseball Season on its own merits based upon the potential harm to the integrity of the playing field and the date of the next use thereof by the Partnership.

SECTION 5.4:

Management. The Partnership shall be exclusively responsible for the management and operation of the Leased Premises pursuant to the terms and conditions of Article VIII and the District hereby grants to the Partnership all required access rights to any portion or all of the Leased Premises at all times necessary for the Partnership to fulfill its obligations in its capacity as the "Manager" thereof pursuant to such Article VIII, specifically including but not limited to the right to grant access to the concession areas of the Stadium to those concessionaires who desire access thereto for the purposes of conducting maintenance, repair, cleaning, stocking, restocking or other activities relating to the operation thereof by such concessionaires.

SUBSECTION 8.1(1):

(f) <u>Damaged Property</u>. "Damaged Property" means any equipment, facility, structure, furniture, fixture or any other component of a Repair Area which is damaged or destroyed, including any damage or destruction resulting from the acts or omissions of third parties (including licensees or invitees of the Partnership or the Team and including licensees or invitees of the District with respect to any District sponsored Non-Baseball Events).

SUBSECTION 8.2(i):

(i) <u>Obligations Regarding Damaged Property</u>. The Partnership shall provide the insurance required by the provisions of Article VI to cover the repair, restoration or replacement of any Damaged Property. The Partnership shall

be responsible for the payment of all deductibles attendant to such insurance, specifically excluding those instances where the Damaged Property is the result of any District sponsored Non-Baseball Event, in which case the District shall be solely responsible for payment of all the deductibles applicable to such Damaged Property loss (unless such loss is caused by the Partnership, any invitees or licensees of the Partnership or members of the Team in attendance at such District sponsored Non-Baseball Event). To the extent that such insurance is insufficient to cover the costs of any repair, restoration or replacement of any Damaged Property, the Partnership shall be solely responsible for such costs (including the costs of all labor, supplies, materials and equipment), specifically excluding those instances where the Damaged Property is the result of any District sponsored Non-Baseball Event, in which case the District shall be solely responsible for such costs (including the costs of all labor, supplies, materials and equipment) unless the Damaged Property is the result of action by the Partnership, its licensees or invitees, or members of the Team in attendance at such District sponsored Non-Baseball Event. No moneys in the Repairs Fund shall be used for the repair, restoration or replacement of any Damaged Property if the damage is the result of the gross negligence or intentional or willful misconduct of the Partnership.

SUBSECTION 8.3(b)(ii):

(ii) Non-Baseball Events. The Partnership shall be responsible for performing and completing any day-to-day landscaping and maintenance of any portion of the Leased Premises required in connection with Non-Baseball Events which are permitted to be held thereon pursuant to the terms and conditions of this Agreement, including any necessary preparation and conditioning of the Playing Field before or during such Events, and the landscaping and maintenance (including sodding and seeding) as may be required in the Partnership's reasonable judgment after such Events in order to restore the Playing Field to first-

class major league condition. Such restoration shall be completed by: (1) the beginning of the next Major League Baseball Season after such Non-Baseball Events occurring during the Off-Season; or (2) by the date of the next Major League Baseball Game after such Non-Baseball Events occurring during the Major League Baseball Season. For Non-Baseball Events sponsored by the District, the District shall promptly pay and/or reimburse the Partnership (or cause a promoter or other third party to reimburse the Partnership) for the Partnership's personnel and material costs in providing such landscaping and maintenance and for reasonable costs associated with the repair or restoration of any portion of the Leased Premises which becomes Damaged Property as a result of a Non-Baseball Event held by the District (except for any such costs that have been deducted by or reimbursed to the Partnership pursuant to Section 4.2(f) hereof). The Partnership shall supply the District and any promoters or third parties designated by the District with itemized invoices for such personnel and material costs, including supporting invoices for costs incurred in acquiring materials. If the District believes that the Partnership is failing to perform its obligations under this Section 8.3(b)(ii), the District shall promptly notify the Partnership, and the Partnership shall then be obligated to cause its groundskeepers to perform such duties.

SUBSECTION 8.10(c):

(c) Advances or Reimbursements from the Repairs Fund. Subject to the provisions of Section 8.10 (a), 8.10 (b), and 8.11, the Partnership shall be entitled to use funds from the Repairs Fund for the payment of Repair work on the Leased Premises and Partnership Property. The Partnership may, in its discretion, elect to perform any Repair and seek reimbursement from the Repairs Fund after completion or seek advances from the Repairs Fund to pay for the Repair construction as the work progresses. The Partnership recognizes that the purpose of the Repairs Fund is to ensure the availability of adequate funds to perform Repair Work

throughout the Term and Grace Term. The Partnership shall exercise prudence in its use of the Repair Fund. The District may withhold its approval of advances or reimbursements from the Repairs Fund if the Partnership does not comply with the provisions of Section 8.10(b). Subject to the terms of Section 8.10(d) hereof, nothing in this Section 8.10(c) or elsewhere in this Agreement shall be interpreted as a limitation upon the Partnership's obligation to perform all Repair work, at its sole expense, throughout the Term and Grace Term. The Partnership shall not submit a request for advance from the Repairs Fund or part thereof prior to completion of any Repair or portion thereof for which advance or reimbursement is requested. By submission of the request, the Partnership warrants that to the best of its knowledge, the Leased Premises or Partnership Property, is free and clear of liens, claims, security interests or encumbrances (except for such liens, claims, security interests or encumbrances previously approved by the District or permitted pursuant to the provisions of Section 2.4) in favor of the Partnership, subcontractors, material suppliers or other entities making a claim by reason of having provided labor, materials or equipment relating to the Repair and that the requirements of any governmental authority with jurisdiction over the Leased Premises, Partnership Property or both have been met. The District may refuse to consent to payments from the Repairs Fund to such extent as may be necessary to protect the District from loss because of:

- (i) Defective performance in effectuating the Repair or any part thereof;
- (ii) Third-party claims made or filed or reasonable evidence indicating the probability of such filing; or
- (iii) Failure of the Partnership to make payments properly to subcontractors for labor, materials or equipment. If the District in its sole discretion determines that any work or materials do not conform to any approved plans and specifications or sound building practices, or otherwise depart from any of the requirements of this Agreement, the District may require the work to be stopped

and withhold disbursements until the matter is corrected. In such event, the Partnership will promptly correct the work to the District's reasonable satisfaction.

Upon completion of any Repair, the Partnership shall provide or make available to the District a complete set of record documents regarding such Repair.

SECTION 24.5:

- 24.5 <u>Public Parking</u>. The District, as owner of the Stadium Land, hereby directs the Partnership, in its capacity as "Manager" thereof, to allow a portion of the parking facilities located on the Stadium Land to be used for paid public parking on days on which no Major League Baseball Games or Non-Baseball Events are being held on the Leased Premises and Partnership Property. The parking facilities to be used for such public parking shall be as agreed to by the District and the Partnership or, if no agreement can be reached, as reasonably designated by the District. In no event will the Stadium Land be used for public parking if it would result in a monetary loss to the Partnership or the District.
- 2. Revision of Definition. Any and all references in the Agreement to an "Off-Season Non-Baseball Event" shall hereinafter be amended to read "Non-Baseball Event."
- 3. New Section 6.13 Added. Article VI of the Agreement shall be and is hereby amended by the addition of new Section 6.13 which shall read as follows:
 - 6.13 Increased Cost of Partnership Insurance Due to District Sponsored Non-Baseball Events. If, due solely to the use of the Stadium and the Stadium Land by the District (or its licensees or invitees) for greater than ten (10) District sponsored Non-Baseball Events during any calendar year, there is an increase in the cost of any insurance premiums with respect to any insurance required to be maintained by the Partnership by this Article VI (or any portion thereof), then the District shall be solely liable for all such increases in premiums and shall promptly reimburse the

Partnership for such increased insurance costs upon notification thereof by the Partnership.

- 4. New Subsection 8.10(d) Added. Article VIII of the Agreement shall be and is hereby amended by the addition of a new Subsection 8.10(d) which shall read as follows:
 - Baseball Events. Notwithstanding any contrary term or provision contained within this Agreement, if during any calendar year of the Term or Grace Term the District sponsors or causes to be held greater than ten (10) Non-Baseball Events, then the Partnership shall be entitled to charge to and collect from the District any increased costs of Maintenance or Repair of the Leased Premises which are caused by each such District Non-Baseball Event (after the tenth) over those Maintenance and Repair costs which would have been incurred by the Partnership without the occurrence of such additional Non-Baseball Events.
- 5. Further Indemnification of the District. If the adoption of this Amendment (applied without regard to this Paragraph 5) would cause the District to bear any property tax liability with respect to the Leased Premises that would, under the terms of the Lease without regard to this Amendment, have been borne by the Partnership, the Partnership shall indemnify the District against any such property tax liability. The indemnity provided by this Paragraph 5 shall not apply to property taxes imposed as a result of the District's ownership of the Leased Premises or as a result of District-sponsored Non-Baseball Events. Any indemnification required by this Paragraph 5 shall be made by the Partnership to the District in sufficient time to enable the District to pay timely any property tax liability for which the District is entitled to indemnity (but not less than thirty days after the Partnership's receipt of notice from the District of the amount of property tax for which the District claims entitlement to indemnity and copies of all documents then in the possession of the District and not previously provided to the Partnership pertaining to the property tax liability for

which indemnity is claimed). The Partnership's obligation to indemnify the District under this Paragraph 5 shall be conditioned on the District's compliance with the following:

- (a) Promptly after receipt of any "Notice of Valuation" or other communication indicating an assessment of property tax for which the District would be entitled to indemnity under this Paragraph 5 is contemplated or intended (and in any event in sufficient time to enable the Partnership to respond to such communication within applicable time limits), the District shall provide a copy of such notice or other communication to the Partnership. Thereafter the District shall promptly furnish to the Partnership copies of any additional written communications and summaries of any oral communications received by the District pertaining to the assessment.
- (b) The Partnership shall have the right, at the Partnership's option and the Partnership's sole cost, risk and expense, to contest in any appropriate forum and by any appropriate remedy [including without limitation both pre-assessment remedies (such as valuation protests) and post-assessment remedies (such as petitions for abatement or refund)] the amount, validity or applicability of any assessment of property tax for which the District would be entitled to indemnity under this Paragraph 5. The District shall cooperate with the Partnership in any such contest (provided that the Partnership reimburses the District for any costs incurred by the District as a result of its cooperation) and shall allow the Partnership (and hereby grants to the Partnership the right) to prosecute any such contest in the name of the District. In the event that any such contest results in a reduction or elimination of a property tax liability for which the Partnership has already indemnified the District by payment to the District or to the appropriate taxing authority, the District shall promptly pay over to the Partnership the amount of any refund received by the District on account of such reduction, including any interest received by the District in connection with such refund.

The Partnership's payment of any indemnity claimed by the District pursuant to this Paragraph 5 shall not operate as a waiver of the Partnership's right to dispute the applicability of this Paragraph 5 to the property tax liability for which indemnity is claimed or the amount of the property tax liability to which this Paragraph 5 applies, provided that the Partnership gives notice of its dispute to the District no later than one year after its payment of the indemnity. If the Partnership and the

District are unable to resolve any such dispute, the Partnership may bring an action in the District Court in and for the City and County of Denver for a determination of the dispute.

- 6. Ratification. Both the Partnership and the District specifically intend and acknowledge that this Amendment shall become effective immediately upon the execution thereof by both the District and the Partnership. However, the District will be required to present this Amendment to its Board of Directors for ratification. If the District's Board of Directors fails to ratify this Amendment on or before July 1, 1996, the terms and conditions hereof shall terminate effective as of July 1, 1996 or, if earlier, the date upon which the District's Board of Directors fails to ratify this Amendment by final action of such Board of Directors.
- 7. Reaffirmation. All provisions of the Agreement, except as specifically amended in this Amendment, shall remain in full force and effect and are hereby reaffirmed by the District and the Partnership.
- 8. <u>Counterparts</u>. This Amendment may be executed in two counterparts which, when taken together, shall constitute the original document. After execution by both parties hereto, the District and the Partnership shall, within a reasonable time, provide a fully executed original hereof to the other party. Facsimile copies of any originally executed signature page of any counterpart original hereof shall be sufficient to bind the party executing such counterpart to the terms and conditions set forth herein.

[EXECUTIONS APPEAR ON THE FOLLOWING PAGE]

IN WITNESS WHEREOF, the Distri	ct and the Partnership have executed this Amendment
and the late and the without	
	COLORADO ROCKIES BASEBALL CLUB, LTD., a Colorado limited partnership By: Colorado Baseball 1993, Inc., general partner
	Ву:
	Jerry D. McMorris, Chairman, President and Chief Executive Officer
	DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT
	By:
[#17609, v.16]	R.T. Baker, Chairman of the Board of Directors
•	

ğ

A CONTRACTOR OF THE PARTY OF TH

þ

March Co.

IN WITNESS WHEREOF, the Distras of the date first above written.	ict and the Partnership have executed this Amendment
	COLORADO ROCKIES BASEBALL CLUB, LTD., a Colorado limited partnership By: Colorado Baseball 1993, Inc., general partner By: Jerry D. McMorris, Chairman, President and Chief Executive Officer
	DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT
[#17609, v.16]	By:

.

ì

.

transference and the second

6,000

12.29.1995 13:39

P. 62

IN WITNESS WHEREOF, the District and the Partnership have executed this Amendment as of the date first above written.

COLORADO ROCKIES BASEBALL CLUB, LTD., a Colorado limited partnership

By: Colorado Baseball 1993, Inc., general partner

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT

By:

T. Haker, Chairman of the Board of Directors

[#17609]

FIRST AMENDMENT TO THE DECEMBER 31, 1995 FIRST AMENDMENT TO THE MARCH 30, 1995 AMENDED AND RESTATED LEASE AND MANAGEMENT AGREEMENT BY AND BETWEEN

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT AND COLORADO ROCKIES BASEBALL CLUB, LTD.

This First Amendment to the First Amendment to the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. (this "Amendment") is dated this 24th day of July, 1996.

In consideration of the premises and the consideration set forth in the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. (the "Agreement"), the receipt and sufficiency of which are hereby acknowledged, the Denver Metropolitan Major League Baseball Stadium District (the "District") and the Colorado Rockies Baseball Club, Ltd. (the "Partnership") hereby agree to amend numbered section 5 of the First Amendment to the March 30, 1995 Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. (the "First Amendment") as provided herein.

1. The first sentence of numbered section 5 of the First Amendment, entitled "Further Indemnification of the District", is hereby amended to read:

If the Adoption of this Amendment causes the District to bear any property tax liability with respect to the Leased Premises, the Partnership shall indemnify the District against any such property tax liability.

2. All provisions of the Agreement and the First Amendment, except as specifically amended in this Amendment, shall remain in full force and effect and are hereby reaffirmed by the District and the Partnership.

[EXECUTIONS APPEAR ON THE FOLLOWING PAGE]

1

IN WITNESS WHEREOF, the District and the Partnership have executed this Amendment as of December 31, 1995.

COLORADO ROCKIES BASEBALL CLUB, LTD., a Colorado limited parmerahip

By: Colorado Baseball 1993, Inc., general partner

By:

12 Mcmon 8-9-96

Verry D. McMorris, Chairman, President, and Chief Executive Officer

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT

By:

R. J. Baker, Chairman of the Board of Directors

SECOND AMENDMENT TO THE MARCH 30, 1995

AMENDED AND RESTATED LEASE AND MANAGEMENT AGREEMENT BY AND BETWEEN

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT AND COLORADO ROCKIES BASEBALL CLUB, LTD.

This Second Amendment to the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. (this "Amendment") is dated this Lott day of Licenses, 1997.

RECITALS

- A. All defined terms set forth in the March 30, 1995 Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. (the "Agreement") shall have the same meanings ascribed to such terms in the Agreement for the purposes of this Amendment.
- B. Section 8.11(a) of the Agreement establishes a Repairs Fund and provides that the Partnership will make payments to the Repairs Fund on January 15 of each year throughout the Term and Grace Term, and that these payments shall increase, annually, at a rate equal to the lesser of 4% or the percentage increase in the F.W. Dodge Construction Costs Index for Denver for the preceding year.
- C. The District and the Partnership have learned that the F.W. Dodge Construction Costs Index is no longer published. Section 8.11(a) of the Agreement provides that in the event of such a discontinuation in publication the District and the Partnership are to agree upon a substitute publication to be utilized in determining the annual percentage increase in the Partnership's payments to the Repairs Fund.
- D. The District and the Partnership have agreed that the Marshall Valuation Service Index of Comparative Cost Multipliers for Denver published by Marshall and Swift, L.P. shall be substituted for the no longer published F.W. Dodge Construction Costs Index.

AGREEMENT

NOW, THEREFORE, in consideration of the premises and the consideration set forth in the Agreement, the receipt and sufficiency of which are hereby acknowledged, the District and the Partnership hereby agree to amend section 8.11(a) of the Agreement by making the following changes thereto:

1. The second sentence in the first paragraph of section 8.11(a) shall be replaced with the following sentence:

THIRD AMENDMENT TO AMENDED AND RESTATED LEASE AND MANAGEMENT AGREEMENT DATED MARCH 30, 1995 BY AND BETWEEN

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT AND COLORADO ROCKIES BASEBALL CLUB, LTD.

This Third Amendment to the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. (this "Amendment") is dated as of April 17, 2009.

In consideration of the premises and the consideration set forth in the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd., as amended by a First Amendment dated December 31, 1995 and a Second Amendment dated December 10, 1997 (as amended, the "Agreement"), the receipt and sufficiency of which are hereby acknowledged, the Denver Metropolitan Major League Baseball Stadium District (the "District") and the Colorado Rockies Baseball Club, Ltd. (the "Partnership") hereby agree to amend Exhibit A to the Agreement to include the property acquired by the District on April 16, 2009.

1. Exhibit A is hereby amended to include the following described property:

Lots 8 through 13, Block 9, Gaston's Addition to the City of Denver, except that portion of Lot 13 conveyed to the Regional Transportation District in the Warranty Deed recorded April 15, 1993 under Reception No. 930047852, City and County of Denver, State of Colorado;

and together with the Southeast ½ of vacated Delgany Street adjacent thereto, as contained in Vacation Ordinance #340 Series of 2001 recorded April 20, 2001 under Reception No. 2001059787, City and County of Denver, State of Colorado;

and together with the Northwest ½ of Vacated Alley adjacent to said lots, vacated by Ordinance No. 619, Series of 2006, recorded September 27, 2006 under Reception No. 2006155052.

and

Lot 14, Block 9, Gaston's Addition to the City of Denver, except that portion of Lot 14, conveyed to the Regional Transportation District in the Warranty Deed recorded April 15, 1993 under Reception No. 930047852, City and County of Denver, State of Colorado;

and together with the Northwest ½ of vacated alley adjacent to said lot, vacated by Ordinance No. 619, Series of 2006, recorded September 27, 2006 under Reception No. 2006155052.

FOURTH AMENDMENT TO AMENDED AND RESTATED LEASE AND MANAGEMENT AGREEMENT BY AND BETWEEN DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT AND COLORADO ROCKIES BASEBALL CLUB, LTD.

This Fourth Amendment to the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd. (this "Amendment") is dated as of March 18, 2015.

In consideration of the premises and the consideration set forth in the Amended and Restated Lease and Management Agreement by and between Denver Metropolitan Major League Baseball Stadium District and Colorado Rockies Baseball Club, Ltd., as amended by a First Amendment dated December 31, 1995, the First Amendment to the First Amendment dated as of December 31, 1995, the Second Amendment dated December 10, 1997 and as further amended by the Third Amendment dated as of April 17, 2009 (as amended, the "Agreement"), the receipt and sufficiency of which are hereby acknowledged, the Denver Metropolitan Major League Baseball Stadium District (the "District") and the Colorado Rockies Baseball Club, Ltd. (the "Partnership") hereby agree to amend the Agreement as follows:

- 1. A new section, Section 8.17 is added to the Lease to read as follows:
 - 8.17 Expedited ADR Regarding Use of Repair Fund. Disputes regarding the approval of the use of funds from the Repair Fund pursuant to Article VIII hereof shall be determined using an expedited alternative dispute resolution process ("Expedited ADR"). The District and the Partnership shall mutually agree on a qualified person to resolve Repair Fund Disputes, and will designate the person as the "Neutral" to whom Repair Fund Disputes are to be submitted for resolution under this Section 8.17.

If the person designated as the Neutral refuses, or for any other reason is unable to serve, as the Neutral with respect to a given Repair Fund Dispute, the District and the Partnership shall promptly designate another individual to serve as the Neutral with respect to such Expedited ADR Dispute. Such other individual shall be independent of the District and the Partnership (and their respective Affiliates) and shall hold no financial interest in, or have any material financial or personal relationship with, either the District or the Partnership (or their respective Affiliates). Such individual shall be employed by a nationally-known architecture or design firm and shall have experience in the design and/or construction of major league baseball facilities.

The Expedited ADR shall be conducted by the Neutral at a time and location in the City and County of Denver. Colorado selected by the Neutral. The Neutral shall give the District and the Partnership reasonable notice of the Expedited ADR, and shall make reasonable efforts to accommodate the schedules of the District and the Partnership in a manner that does not delay the prompt resolution of the issue(s) to be decided by the Neutral. The Neutral shall conduct the Expedited

ADR in such manner as the Neutral deems appropriate, consistent with the provisions of this Section 8.17. The District and the Partnership intend that the Neutral have the sole and exclusive authority and power to resolve Expedited ADR Disputes by making the decisions described herein. The Neutral shall not have the power or authority to award any damages or require any payments other than those described in the last paragraph of this Section 8.17.

There shall be no discovery permitted with respect to any Expedited ADR other than that required by the Neutral, and each of the District and the Partnership shall present its position with respect to the issue(s) to be determined by such Expedited ADR by an oral presentation to the Neutral. Each of the District and the Partnership shall be given the opportunity to hear and orally respond to the other's presentation to the Neutral, and to present documents to the Neutral in support of such party's position. The Neutral shall have the right to limit the documents presented to the Neutral to assure a prompt resolution of the issue(s) to be determined by the Neutral. [Each of the District and the Partnership may have its counsel present at such Expedited ADR, but there shall be no examination or cross-examination of witnesses other than required or permitted by the Neutral.]

The District and the Partnership shall cooperate in good faith to permit a conclusion of the Expedited ADR within seven (7) days following the submission of the Expedited ADR Dispute to the Neutral.

The District and the Partnership shall use Expedited ADR exclusively, rather than litigation, as a means of resolving all Repair Fund Disputes. The written award by the Neutral shall be the binding, final determination on the merits of the Repair Fund Dispute, and shall preclude any subsequent litigation on such merits. The District and the Partnership agree that any disputes which arise out of such a written award shall be resolved exclusively by Expedited ADR pursuant to this Section 8.17, provided that the District and the Partnership may institute legal proceedings in a court of competent jurisdiction to enforce judgment upon an Expedited ADR award in accordance with Applicable Law. The fees and costs of the Neutral shall be borne as directed by the Neutral; provided, however that the substantially prevailing party in Expedited ADR shall be entitled to reimbursement for any costs of such proceedings, reasonable attorneys' fees, reasonable costs of investigation and any other expenses incurred in connection with such Expedited ADR in the manner directed by the Neutral.

2. A new article XXVIII is added to the Lease to read as follows:

28.1 Arbitration. Except for (i) Expedited ADR Disputes to be resolved pursuant to Section 8.17, and (ii) disputes with respect to termination of the Partnership's stadium management responsibilities pursuant to Section 8.8(e), any event of default or any other dispute between the District and the Partnership under, in connection with or relating to this Agreement (each, an "Arbitration Dispute") shall be submitted to binding arbitration ("Arbitration") under this Section 28.1.

The Arbitration shall be conducted before the Judicial Arbiter Group ("JAG") at a location in the City and County of Denver, Colorado selected by the Arbiter. Either party may commence the arbitration by written demand to the other party with a copy to JAG. JAG shall provide to the parties, within fifteen (15) days after the demand, a list containing an odd number of the names of at least five (5) potential arbiters. The parties, beginning with the Partnership, shall each strike one name in turn until only one (1) name remains. If it is one party's turn and that party does not strike a name within three (3) business days, then that turn is forfeited, and the other party shall strike the next name. When only one (1) name remains, the remaining person shall be selected as the single arbiter. The Arbitration shall be conducted under the Colorado Arbitration Act, subject to this Agreement and any other documents executed by the parties hereto. In the event of a conflict between this Section 28.1 and the Colorado Arbitration Act, this Section 28.1 shall govern. The District and Partnership shall make reasonable efforts to agree on discovery rules and to the extent and scope of discovery with respect to any Arbitration Dispute and as outlined in the Colorado Arbitration Act. In the event the parties are not able to agree on such rules and the extent and scope of such discovery, all issues relating to such discovery shall be resolved by the Arbiter in his/her sole discretion. Unless waived by each of the parties participating in the Arbitration, the Arbiter shall conduct an Arbitration hearing at which the parties and their respective counsel may be present and have the opportunity to present evidence and examine cross-examine witnesses. Witnesses shall, unless waived by the parties, present testimony under oath.

If the Arbiter determines that the matters or issues involved in any Arbitration Dispute are outside the scope of the Arbiter's expertise, the Arbiter shall have the right to retain and rely on experts with respect to such matters and issues. The cost of any expert retained by the Arbiter shall be a cost of the Arbitration to be paid as set forth herein. Any information obtained by the Arbiter from an expert engaged by the Arbiter shall be disclosed by the Arbiter to both the District and the Partnership, and each such party shall have the right to present evidence and/or testimony from such party's own expert with respect to such matter or issue.

The parties agree to expedite the arbitration to the fullest extent possible. Unless the parties agree in writing to a different schedule, the arbitration hearing shall be commenced with thirty (30) days from the date of the selection of the arbiter. Within fifteen (15) days following the closing of the hearing, a written award and/or decision shall be made by the arbiter and accompanied by findings of fact and conclusions of law, which shall be contemporaneously delivered to the parties.

The parties hereto shall use Arbitration exclusively, rather than litigation, as a means of resolving all Arbitration Disputes. Notwithstanding any other provision of this Section 28 to the contrary, in the event any party hereto desires to seek interim equitable relief with respect to an Arbitration Dispute in the form of a

temporary restraining order, preliminary injunction or other interim equitable relief concerning an Arbitration Dispute, including specific performance, provisional remedies, stay of proceedings in connection with special action relief or any similar relief of an interim nature, at any time before an arbiter has been selected pursuant to the procedure in this Section 28.1., such party may initiate litigation in in the City and County of Denver to obtain such interim equitable relief ("Equitable Litigation"). Nothing herein shall be construed to suspend or terminate the obligation of any party hereto to promptly proceed with Arbitration concerning the Arbitration Dispute that is the subject of such Equitable Litigation while such Equitable Litigation (and any appeal therefrom) is pending. Regardless of whether such interim relief is granted or denied, or whether such Equitable Litigation is pending or any appeal is taken from the grant or denial of such relief, the parties shall at all times diligently proceed to complete the Arbitration. Any interim relief granted in such Equitable Litigation, or any appeal therefrom, shall remain in effect until, and only until, the Arbitration concerning the Arbitration Dispute that is the subject of such Equitable Litigation results in the issuance of an award or decision following Arbitration or until the parties enter into a private written settlement agreement (hereinafter a "Settlement Agreement").

The award and/or decision by the Arbiter shall be the binding, final determination on the merits of the Arbitration Dispute (including any equitable relief but excluding any award of attorneys' fees or costs awarded or granted in the Equitable Litigation), shall supersede and nullify any decision in the Equitable Litigation on the merits of the Arbitration Dispute, and shall preclude any subsequent litigation on such merits, notwithstanding any determination to the contrary in connection with any Equitable Litigation granting or denying interim relief or any appeal therefrom. Judgment upon the award rendered by the Arbiter may be entered in any court of competent jurisdiction. The parties agree that any disputes which arise out of a Settlement Agreement resolving an Arbitration Dispute or arising from an arbiter decision shall be resolved exclusively by Arbitration pursuant to this Section 28, provided that any party may institute legal proceedings in a court of competent jurisdiction to enforce judgment upon an Arbitration award. Each of the District and the Partnership shall bear its own costs and fees and an equal share of the fees and expenses of the Arbitration and the administrative fees of the Arbitration, including costs and expenses of experts retained by the Arbiter. All aspects of the arbitration shall be treated as confidential. Neither the arbitrating parties, JAG, nor the Arbiter may disclose the existence, content, or results of the arbitration, except as necessary to enforce the award or to comply with legal or regulatory requirements, including but not limited to, the Colorado Open Records Act.

3. Subsection 4.2(e) is amended by the addition of the following after the first sentence:

"In calculating net profit, the Partnership shall be entitled to deduct fifteen percent (15%) of gross receipts as operating costs to cover the Partnership's

- administrative costs and overhead. All amounts received by the District shall be deposited into the Repairs Fund established pursuant to Section 8.11."
- 4. Section 16.5 is amended by the addition of the following at the end of the section:
 - "The District, alternatively, upon written consent signed by the Partnership and the District, may deposit any such proceeds into the Repairs Fund established pursuant to Section 8.11 and use such funds in accordance with Article VIII.
- 5. All provisions of the Agreement, except as specifically amended in this amendment, shall remain in full force and effect and are hereby reaffirmed by the District and Partnership.

[EXECUTIONS APPEAR ON THE FOLLOWING PAGE]

DENVER METROPOLITAN MAJOR LEAGUE BASEBALL STADIUM DISTRICT

Raymond T. Baker, Chairman

COLORADO ROCKIES BASEBALL CLUB, LTD. a Colorado limited partnership

By: COLORADO BASEBALL 1993, INC., its

general partner

By: